Tax as a force for good

Aligning tax systems with the SDGs and the inclusive circular economy

CASE STUDY BANGLADESH

CAAA

The Ex'tax Project In cooperation with Cambridge Econometrics Supported by C&A Foundation

Ara

Colophon

Lead author Femke Groothuis (The Ex'tax Project)

Main partners Cambridge Econometrics, C&A Foundation

Research and editing Valérie van de Luitgaarden, Peter Gersen (The Ex'tax Project)

Design Frank Paats (www.frankpaats.nl)

Acknowledgements

We thank the following people for providing valuable information and insight: Douwe Jan Joustra, Megan McGill, Sandra Polaski, Jill Tucker (C&A Foundation), Hector Pollitt, Alistair Smith (Cambridge Econometrics), Md. Ahsanul Hoque Bashar, Md. Shafaat Ali Choyon, Noor-E-Fayzun Nahar, Rashed Gomadder Mohammad, Prawma Tapashi Khan (ACCA Bangladesh), Kanon Kumar Roy (National Board of Revenue, Bangladesh), M. Abul Kalam Mazumdar (ICMAB), Md. Abdur Rahman Khan (Ministry of Finance, Government of Bangladesh), Maruf Hasan (Fernhill Consulting House), Arif Khan (IDLC Finance Limited), Md. Kausar Alam (ICMAB), Dr. Shahriar Hossain (journalist, ecologist and social justice advocate), Saurav Roy (Ashoka), Mohammad Azad Rahman (BRAC), Suvojit Chattopadhyay (Adam Smith International), Haseeb Irfanullah (environmental specialist), Rezaul Siddique (journalist and professor at Savar College), Eelco van der Enden (PwC), Tony Henshaw (Sustainable Apparel Coalition), Sebastian Groh (Solshare).

This study builds on the paper 'Tax as a force for good; Rebalancing our tax systems to support a global economy fit for the future',' written by the author and published by the ACCA in 2018.

The views expressed in this publication do not necessarily reflect the views of the C&A Foundation or its Executive Board. September 2019

Parts of this publication may be copied with reference to:

The Ex'tax Project (2019), Tax as a force for good: Aligning tax systems with the SDGs and the inclusive circular economy. Case study Bangladesh.

This publication is made possible by the support of the C&A Foundation.



THE EX'TAX PROJECT IS A THINK TANK FOCUSED ON FISCAL INNOVATION FOR INCLUSIVE, CIRCULAR ECONOMIES.

www.ex-tax.com



CAMBRIDGE ECONOMETRICS' MISSION IS TO PROVIDE CLEAR INSIGHTS, BASED ON RIGOROUS AND INDEPENDENT ECONOMIC ANALYSIS, TO SUPPORT POLICY-MAKERS AND STRATEGIC PLANNERS. WWW.camecon.com



C&A FOUNDATION IS A CORPORATE FOUNDATION AIMING TO FUNDAMENTALLY TRANSFORM THE FASHION INDUSTRY.

www.candafoundation.org

Table of Content

Abst	ract		5
Exec	cutive \$	Summary	6
Intro	ductio	n	. 10
1.		the SDG era: countries in transition he SDGs and the importance of tax reform	. 12 13
		he 'polluter pays' principles are not applied	16
		abour taxes on the rise	19
		pdating tax systems to support the inclusive circular economy	21
		requently asked questions	26
	1.5.1	How does tax reform work?	26
	1.5.2 1.5.3	What are the impacts of tax reform for the economy?	28 29
	1.5.4	Where has tax reform been put in practice? Does tax reform increase inequality?	29 30
-			
2.		adesh introduction	. 31
		eography, population and climate	32
		ocio-economic development	33
		nvironmental issues	34
		ax structure rogress on achieving the SDGs by 2030	36 38
•			
3.		oping tax reform scenarios	. 39
		he Ex'tax Methodology and Toolkit	40
		imiting factors	40
	3.3 T 3.3.1	wo preliminary tax reform scenarios Raising revenues through a Carbon Tax (for industries)	41 42
	3.3.2	Raising revenues through a Carbon Tax (for industries) Raising revenues through removal of fossil fuel subsidies (for industries)	42 45
	3.3.3	Use of revenues: investing in clean technologies	47
	3.3.4	Use of revenues: investing in infrastructure	48
	3.3.5	Use of revenues: increasing social spending	48
4.	Model	ling results	. 50
		ntroducing the macro-econometric model	51
		ey results	52
		npacts by sector	55
5.	Explo	ring the implications of tax reform for the textiles sector	. 58
	5.1 S	etting the scene: global megatrends	59
	5.2 S	taying competitive in a 1.5-degree warmer world?	60
6.	Concl	usion and recommendations	. 62
Арре	endix A	A: FRAMES: Bangladesh	. 66
Арре	endix E	3: Sources	. 74

Figures

Figure 1: Tax policy targets in the SDGs	15
Figure 2: The linear 'take-make-waste' economy	16
Figure 3: Environmental tax as a % of GDP (2000-2016)	. 17
Figure 4: Labour taxes as a % of GDP (2000-2016)	. 19
Figure 5: Value Extracted Tax (Ex'tax) principles	. 21
Figure 6: Shifting from a linear to a circular economy	. 22
Figure 7: The Ex'tax Toolkit – Raising Revenues	. 27
Figure 8: The Ex'tax Toolkit – Use of Revenues	. 27
Figure 9: The Ex'tax Methodology	40
Figure 10: The Infrastructure Scenario and the Social Scenario (in 2025, Bangladesh)	. 42
Figure 11: Overall result: decoupling (2020-2025, % difference from baseline, Bangladesh)	. 52
Figure 12: Key modelling results (2025, % difference from baseline, Bangladesh)	53
Figure 13: Social Scenario - real incomes per quintile (2025, % diff. from baseline, Bangladesh)	55
Figure 14: Output by sector (2025, % diff. from baseline, Bangladesh)	55
Figure 15: Employment per sector (2025, diff. from baseline, # of employed persons, Bangladesh)56
Figure 16: CO ₂ emissions per sector (2025, % difference from baseline, Bangladesh)	. 57

Tables

Table 1: Examples of business model innovation by sector	. 24
Table 2: Bangladesh Budgetary Central Government Revenues (2016)	. 36
Table 3: Bangladesh Budgetary Central Government Expense by Economic Type (2016)	. 37
Table 4: Status of fiscal targets mentioned in the 7th Five Year Plan of Bangladesh	. 37
Table 5: Selection of Bangladesh' national strategies to address SDGs	. 38
Table 6: Purpose, areas of concern and potential solution of the revenue-raising measures	. 46
Table 7: Cumulative results (2020-2025, difference from baseline, Bangladesh)	. 54
Table 8: Economic, social & environmental impacts (2025, difference from baseline, Bangladesh)) 54

Boxes

Box 1: Social and environmental megatrends (a selection)	13
Box 2: The Sustainable Development Goals	14
Box 3: External costs studies	16
Box 4: Support for shifting from labour to green taxes	22
Box 5: Carbon tax studies Bangladesh	43

Abstract

This study analyses opportunities to align tax systems with the goals of the SDGs and the inclusive circular economy, by putting a price on natural resource use and pollution and using the revenues to lower the tax burden on labour and increase (social) spending.

The case study focusses on Bangladesh. Bangladesh has one of the largest gaps between tax revenue and GDP. The country is highly prone to climate disruption. Pollution levels are high and renewable energy sources provide only a fraction of energy needs. Bangladesh has a large, growing and resilient population. The country has a proven track record of effectively addressing poverty and other challenges. Sustainable inclusive growth is a major national priority.

Cambridge Econometrics has modelled some of the impacts of two preliminary scenarios, which include introducing a price on carbon emissions and phasing out certain fossil fuel subsidies, while using the revenues to invest in clean technologies, infrastructure and income support. The modelling suggests that by 2025, such tax reforms could lead to higher GDP and employment levels, while reducing carbon emissions and energy imports. The transition can be highly progressive when revenues are mainly used to increase social spending.

The findings suggest that tax reform can be a viable strategy to reach the dual goal of socio-economic development and environmental protection.

Executive Summary

Bridging the SDG finance gap

This is the era of the Sustainable Development Goals (SDGs); an era of unprecedented global social and environmental challenges. The most daunting task will be to adapt the metabolism of our economies to match the carrying capacity of the earth and stay well below 2°C global warming. We face equally important social challenges in our societies, including enabling a growing population to develop to their full potential and find decent work. According to the United Nations, achieving the SDGs is going to cost US\$5-6 trillion a year. Developing countries alone face an annual financing gap of \$2.5 trillion. Official development assistance (ODA) is just \$150 billion per year, and on a downward trend. Private flows of investment are also considerably below the level of SDG investment needs. This means there is a massive financing gap.

Tax and the circular economy

Aligning economic growth with the goals of the SDGs will be key. Therefore, a shift is needed from the linear 'take-make-waste' economy, towards an inclusive circular economy, which is regenerative, carbon neutral and distributive. Such circular economies require labour- and knowledge-intensive activities (e.g. repair and maintenance services, recycling, refurbishment and R&D), which could significantly contribute to job creation. Currently, however, governments tend to increase the tax burden on human labour, which incentivises businesses to reduce labour input. At the same time, governments put low or no taxes on natural resource use (such as carbon emissions, fossil fuels and water), thereby leaving resource use unrestrained, causing overconsumption, pollution and waste.

Leapfrogging tax systems to the SDG era

Given the high unemployment rates in low- and middle-income countries, increasing the tax burden on labour may not necessarily be the best option to raise revenues sustainably. Could taxing pollution rather than people, as advocated by UN Secretary-General António Guterres, be a way of 'leapfrogging' tax systems to the SDG era? Studies indicate that such tax reform (also known as Environmental Tax Reform) can indeed have positive economic, social and environmental impacts. A World Bank study, for example, demonstrated that a domestic carbon tax of \$30 per tonne of CO₂ would provide the resources to more than double current levels of social assistance in 60 countries. Over the years, many researchers and international organisations have called for tax reform; putting a price on pollution and resources and using the revenues to lower the tax burden on labour and increase (social) spending.

Opportunities and risks of tax reform

The goal of this study is to analyse opportunities and risks of aligning tax policy with the goals of the inclusive circular economy in low- and middle-income countries, focussing on Bangladesh as the first case study. Bangladesh has one of the largest gaps between tax revenue and GDP. Green taxes (including restructuring of fossil fuel subsidies), could help raise much-needed tax revenues. The country is highly prone to climate disruption, pollution levels are high and renewable energy sources provide only a fraction of energy needs. It has a large, growing and resilient population of 165 million, with 2 million youths entering the job market every year. The country has a proven track record of effectively addressing poverty and other challenges. Sustainable inclusive growth is a major national priority.

Cambridge Econometrics has modelled some of the impacts of two preliminary scenarios, which include putting a price on carbon emissions and abolishing fossil fuel subsidies, while using the revenues to invest in clean technologies, infrastructure and social spending. The modelling suggests that by 2025, such tax reforms could lead to higher GDP and employment levels, while reducing carbon emissions and energy imports. The transition can be highly progressive when revenues are mainly used to increase social spending.

Two preliminary scenarios

For the purpose of this study, Cambridge Econometrics developed the 'FRAMES: Bangladesh' model, to estimate potential green tax revenues and macro-economic impacts of tax reform scenarios. The Ex'tax Toolkit and Methodology were used to develop two preliminary scenarios (see Figure A). Leading factors in this process were, amongst others, the national priorities of Bangladesh, data availability and the scope of the FRAMES model. The measures are assumed to be introduced gradually from 2020, to reach to reach the full measures by 2024. In the year 2025, both scenarios are expected to raise \$4.3 billion in revenues by 1) introducing a carbon tax of \$30 per tonne of CO₂ emitted by industries and the power sector, and 2) phasing out oil and natural gas subsidies for industries and power generation. In the modelling, every year, the revenues are fully recycled. In the *Infrastructure Scenario*, all revenues are recycled through investments in clean technology and infrastructure. In the *Social Spending Scenario* (or *Social Scenario*), all revenues are recycled through investments in clean technology and social spending targeted towards the lowest two income quintiles. In both scenarios, the cleantech investments are targeted towards the textiles sector (Bangladesh's most important export sector).



Figure A: The Infrastructure Scenario and the Social Scenario (in 2025, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Figure B provides some key results over the 2020-2025 period, demonstrating decoupling effects in the scenario, as GDP is higher, and emissions are lower. An increase in employment is observed in each scenario.

Cumulative results

Over the 2020-2025 period, the scenarios add \$6.9 billion (in the *Infrastructure Scenario*) and \$7.8 billion (in the *Social Scenario*) to GDP, compared to business as usual. Additional results over the six-year period are:

- **Resource mobilisation.** Phasing out fossil fuel subsidies could potentially raise \$4.7 billion in domestic resources, while a carbon tax could add another \$10.6 billion in domestic resources.
- Job creation. Both scenarios show increases in employment (540,000 and 670,000 years of employment respectively).
- **Carbon emission reductions.** Both scenarios demonstrate a significant reduction in carbon emissions (saving 19.9 and 18.5 megatonnes of carbon respectively).

- Savings on energy imports. In both scenarios, Bangladesh saves significant amounts on energy imports (\$429 million and \$405 million respectively).
- Public investments. In the *Infrastructure Scenario* \$12.8 billion is invested in infrastructure. In the *Social Scenario*, \$12.9 billion is invested in social protection. Cleantech investments in both scenarios are to \$2.6 and \$2.5 billion respectively over the 2020-2025 period.



Figure B: Overall result: decoupling (2020–2025, % difference from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Distribution of benefits and costs

As with any reform, the benefits and costs will not be spread evenly. The modelling results suggest that a progressive impact with higher benefits (in relative terms) for lower income households is possible, particularly in the *Social Scenario*. For businesses, in the *Infrastructure Scenario*, the largest increases in output, both in relative and absolute terms, are in the construction, manufacturing (excluding textiles), mining and quarrying sectors. These sectors benefit from the investments in infrastructure. Some of the manufacturing companies in the supply chain also benefit. Output falls marginally in sectors which are supplying consumer final demands, such as retail. In the *Social Scenario*, most sectors demonstrate an increase in output as they benefit from higher local consumer spending. Manufacturing and construction also have relatively high increases, because of the investment in cleantech for textiles.

Results for the textiles sector

In both scenarios, the textiles sector shows a slight negative result in terms of gross output (0.24% and 0.15% respectively) by 2025, but overall, the Bangladesh economy would be stronger and more competitive in terms of carbon intensity and energy import dependency. Also, it's important to note that the competitiveness impacts of the cleantech investments (totalling more than \$2.5 billion) are not yet captured in the model. Bangladesh is the world's second-largest exporter of clothing, and the sector represents 80 percent of foreign earnings. As one of the most polluting industries, operation in a fast-changing global market, the global textiles industry is at a crossroads; continuing the linear model (while imposing external costs to society and future generations), or shifting to circular models, and adapting to changing circumstances. In light of global trends, tax reform could be a way to reduce risks and future-proof the sector.

The key message from the results is that it is possible to design policy measures that reduce harmful emissions and final energy consumption, while at the same time stimulating the economy of Bangladesh, creating jobs and (in the *Social Scenario*) increasing income for the lowest income groups. These results demonstrate that Bangladesh doesn't need to choose between development and environment.

Balancing the interests of stakeholders

The scenarios presented in this study do not claim to be a blueprint or short-term solution, but are a medium- to long-term pathway, to assist in the process of balancing the interests of different stakeholders. Any implementation pathway should be researched and then monitored by the designated national institutions with full access to national statistics. This study is meant to set an example for similar analyses in other countries—particularly those with low- and middle-incomes, facing similar challenges.

Prosperity based on human capital

The foundations of modern tax systems were laid down in the era of the industrial revolution; before globalisation and mass consumption, before the emergence of climate disruption and water supply risks, and before digitisation and automation. In the SDG era it is vital for governments to serve the interests of the people and business at the same time, and to fairly distribute risks and opportunities. Smart tax policies could help countries to 'leapfrog' into the SDG era. Stakeholders, businesses, governments and NGOs should work together to turn tax into a 'force for good' and help build modern tax systems that enable prosperity based less on natural resource use and more on the abundance of human capacities and talents. For this is growth that can be sustained by generations to come.

Introduction

Bridging the SDG finance gap

This is the era of the Sustainable Development Goals (SDGs); an era of unprecedented global social and environmental challenges. The most daunting task will be to adapt the metabolism of our economies to match the carrying capacity of the earth and stay well below 2°C global warming. We face equally important social challenges in our societies, including enabling a growing population to develop to their full potential and find decent work. According to the United Nations, achieving the SDGs is going to cost US\$5-6 trillion a year. Developing countries alone face an annual financing gap of \$2.5 trillion while Official Development Assistance (ODA) is just \$150 billion per year, and on a downward trend. Private flows of investment are also considerably below the level of SDG investment needs. This means there is a massive financing gap.

Tax and the circular economy

Aligning economic growth with the goals of the SDGs will be key. Therefore, a shift is needed from the linear 'take-make-waste' economy, towards an inclusive circular economy, which is regenerative, carbon neutral and distributive. Such circular economies require labour- and knowledge-intensive activities (e.g. repair and maintenance services, recycling, refurbishment and R&D), which could significantly contribute to job creation. Currently, however, governments tend to increase the tax burden on human labour, which incentivises businesses to reduce labour input. At the same time, governments put low or no taxes on natural resource use (such as carbon emissions, fossil fuels and water), thereby leaving resource use unrestrained, causing overconsumption, pollution and waste.

Leapfrogging tax systems to the SDG era

Given the high unemployment rates in low- and middle-income countries, increasing the tax burden on labour may not necessarily be the best option to raise revenues sustainably. Could taxing pollution rather than people, as advocated by UN Secretary-General António Guterres, be a way of 'leapfrogging' tax systems to the SDG era? Studies indicate that such tax reform (also known as Environmental Tax Reform (ETR)) can indeed have a positive economic, social and environmental impact.

Opportunities and risks of tax reform

The goal of this study is to analyse opportunities and risks of aligning tax policy with the goals of the inclusive circular economy in low- and middle-income countries, focussing on Bangladesh as the first case study. Bangladesh has one of the largest gaps between tax revenue and GDP. The country is highly prone to climate disruption, pollution levels are high and renewable energy sources provide only a fraction of energy needs. It has a large, growing and resilient population and the country has a proven track record of effectively addressing poverty and other challenges. Sustainable inclusive growth is a major national priority. This study is meant to be a steppingstone for similar analyses in other countries, facing similar challenges.

Prosperity based on human capital

Turning tax into a 'force for good' should mean building modern tax systems that enable prosperity based less on natural resource use and more on the abundance of human capacities and talents. This is not a simple task, especially in countries with low- and middle-incomes, which face the multiple challenge of developing the economy and social systems while at the same time preserving natural resources. This study confirms that countries may not need to choose between those goals.

The structure of this report

Chapter 1 looks into the challenges in the 'SDGs era', the role of tax systems in achieving the SDGs and the shift from the current linear economic systems towards circular and inclusive economies. Chapter 2 provides an overview of the geographic, socio-economic, environmental and fiscal context in Bangladesh. Chapter 3 introduces the Ex'tax approach, and how two preliminary scenarios were

developed for the case study of Bangladesh. Chapter 4 provides the key macro-economic modelling results. Chapter 5 explores the implications of tax reform for the Bangladeshi textiles industry in the context of global megatrends. The final chapter 6 includes recommendations for next steps.

It should be emphasised that the scenarios in this study are not a blueprint for implementation. They are meant to provide directions on how tax systems can be rationalised in light of the SDGs and the circular economy and what the potential impacts are. Any implementation pathway should be researched and then monitored by the designated national institutions with full access to national statistics.

1. Tax in the SDG era: countries in transition

E E E E E



This chapter explores how tax systems in general need to be updated in order to support the SDGs. Value Extracted Tax (in short: Ex'tax) is the proposal to increase taxes on natural resource use and pollution and using the revenues to lower the tax burden on labour and increase (social) spending. Such tax reform aligns financial incentives with economic growth based on the abundance of talents and capabilities of people, instead of natural resources.

1.1 The SDGs and the importance of tax reform

Interconnected social and environmental challenges

Humanity is facing unprecedented challenges (Box 1 provides a selection). The most daunting task will be to adapt the metabolism of our economies to match the carrying capacity of the earth and stay below 1.5°C of global warming. According to the latest Intergovernmental Panel on Climate Change (IPCC) report, global carbon emissions must start to reduce well within 12 years if we are to prevent large-scale natural and human risks from becoming irreversible reality.¹ We face equally important social challenges in our societies, including enabling a growing global population² to fulfil their basic needs, develop their full potential and find decent work. Governments need coherent strategies to deal with these megatrends.³

Box 1: Social and environmental megatrends (a selection)

Pollution kills 9 million people each year, with most of those deaths in low- and middle-income countries.⁴

Climate change hits the poorest people the hardest. As the effects of climate change worsen, escaping poverty becomes more difficult.⁵

Water scarcity affects nearly half the world's population.⁶ Climate change is projected to further reduce water availability in many water scarce regions.⁷

Biodiversity is severely under threat; one million species of animal and plants face extinction.⁸

Food: One-third of all food produced for human consumption is lost or wasted globally – while at the same time, almost one billion people go hungry.⁹

Plastics: A truckload of plastic waste is dumped in the oceans every minute, which means that in a few decades there will be more plastics than fish in the oceans.¹⁰

Unemployment is a reality for 170 million people in the world today.¹¹ In addition, 140 million people are classified as 'underutilized labour'. Almost 700 million workers in low- and middle-income countries are living in extreme or moderate poverty (i.e. having to live on less than \$3.20 per day in PPP terms).¹²

The Sustainable Development Goals

In 2015, the 193 countries of the United Nations (UN) General Assembly adopted the 2030 Development Agenda titled *Transforming our world: the 2030 Agenda for Sustainable Development*.¹³ The agenda sets out 17 ambitious Sustainable Development Goals (the SDGs or Global Goals) that integrate social issues (such as fighting poverty and improving global health) and specific environmental issues (such as combating climate change) (see Box 2). As humanity's 'to do list', the SDGs will dominate the global agenda for the upcoming decades.

Financing the SDGs

According to the UN, achieving the SDGs is going to cost \$5 to 6 trillion a year. Developing countries alone face an annual financing gap of \$2.5 trillion.¹⁴ Since Official Development Assistance (ODA) is \$153 billion per year, the gap is huge.¹⁵ Overall, ODA levels have fallen since 2016 for the second year in a row,¹⁶ and only five countries met or exceeded the UN benchmark for ODA contributions of at least 0.7 percent of gross national income.¹⁷ Private financing flows (including foreign direct investment, institutional investing, remittance (the money sent home by immigrants), foundations, microfinance and private equity) are higher than public flows, but they are mostly received by developed economies and they are considerably below the level of incremental SDG investment needs.¹⁸

Challenges in collecting taxes

Between 50 percent and 80 percent of what is required for the SDGs in low- and middle-income countries would need to come from domestic resources.¹⁹ Taxation will be a key instrument in financing the SDGs, but the regions that are most in need of resources face significant challenges in collecting taxes:

"The rate of taxation (ratio of tax revenue to GDP) in the least developed countries declined from a peak of 11.1 per cent in 2012 to 8.8 per cent in 2016. Countries in sub-Saharan Africa faced a similar trend: a decline from 14.9 per cent in 2006 to 10.7 per cent in 2016." (United Nations 2018)²⁰

Research indicates that tax revenues above 15 percent of a country's GDP are critical for economic growth and, ultimately, poverty reduction.²¹ In 2016, within the group of emerging countries with tax-to-GDP ratios below 15%, Nigeria had the largest tax-revenue gap, followed by Indonesia and Bangladesh. The total tax-revenue gap in the top-20 countries was \$180 billion (compared to ODA of \$46 billion).²²

Box 2: The Sustainable Development Goals

Goal 1. En	nd poverty in all its forms everywhere.
Goal 2. En	nd hunger, achieve food security and improved nutrition and promote sustainable agriculture.
Goal 3. En	nsure healthy lives and promote well-being for all at all ages.
Goal 4. En:	nsure inclusive and equitable quality education and promote lifelong learning opportunities for all.
Goal 5. Ac	chieve gender equality and empower all women and girls.
Goal 6. En:	nsure availability and sustainable management of water and sanitation for all.
Goal 7. En:	nsure access to affordable, reliable, sustainable and modern energy for all.
Goal 8. Pro	omote sustained, inclusive and sustainable economic growth, full and productive employment and
de	ecent work for all.
Goal 9. Bu	uild resilient infrastructure, promote inclusive and sustainable industrialization and foster innovation
Goal 10. Re	educe inequality within and among countries.
Goal 11. Ma	ake cities and human settlements inclusive, safe, resilient and sustainable.
Goal 12. En	nsure sustainable consumption and production patterns.
Goal 13. Tal	ake urgent action to combat climate change and its impacts.
Goal 14. Co	onserve and sustainably use the oceans, seas and marine resources for sustainable development.
Goal 15. Pro	otect, restore and promote sustainable use of terrestrial ecosystems, sustainably manage forests, combat
de	esertification, and halt and reverse land degradation and halt biodiversity loss.
Goal 16. Pro	omote peaceful and inclusive societies for sustainable development, provide access to justice for all and build
eff	fective, accountable and inclusive institutions at all levels.
Goal 17. Str	rengthen the means of implementation and revitalize the global partnership for sustainable development. ²³

Tax policy reform mentioned in four SDG targets

Taxes are explicitly mentioned in the targets of four SDGs (see Figure 1):

- The first target of **SDG 1 (No Poverty)** is to "Ensure significant mobilization of resources from a variety of sources, including through enhanced development cooperation, in order to provide adequate and predictable means for developing countries, in particular least developed countries, to implement programmes and policies to end poverty in all its dimensions."
- **SDG 10 (reducing Inequality)** includes target 10.4: "Adopt policies, especially fiscal, wage and social protection policies, and progressively achieve greater equality."
- SDG 12 (sustainable consumption and production) includes target 12.C: "Rationalize inefficient fossil-fuel subsidies that encourage wasteful consumption by removing market distortions, in accordance with national circumstances, including by restructuring taxation and phasing out those harmful subsidies, where they exist, to reflect their environmental impacts, taking fully into account the specific needs and conditions of developing countries and

minimizing the possible adverse impacts on their development in a manner that protects the poor and the affected communities."

• **SDG 17 (Partnership for the Goals)**, target 17.1, again mentions the need to "Strengthen domestic resource mobilization, including through international support to developing countries, to improve domestic capacity for tax and other revenue collection."

Taxation strongly connected to the SDGs

According to the *Platform for Collaboration on Tax* (a joint initiative of the International Monetary Fund (IMF), the Organisation for Economic Co-operation and Development (OECD), the UN and the World Bank), taxation is a significant factor in 10 of the 17 SDGs. Taxes and the SDGs are connected in multiple ways:

"(1) taxes generate the funds that finance government activities in support of the SDGs; (2) taxation affects equity and economic growth; (3) taxes influence people's behavior and choices, with implications for health outcomes, gender equity, and the environment; and (4) fair and equitable taxation promotes taxpayer trust in government and strengthens social contracts that underpin development."²⁴

Figure 1: Tax policy targets in the SDGs



Source: United Nations (2015), Transforming our world: the 2030 Agenda for Sustainable Development. https://sustainabledevelopment.un.org/content/documents/21252030 Agenda for Sustainable Development web.pdf

Not simply a matter of taxing more

Mobilizing domestic resources is not simply a matter of taxing more:

*"it's also about taxing better by expanding the tax base, ensuring an appropriate distribution of the tax burden among taxpayers, simplifying and improving the efficiency of tax administration, bringing tax laws up to date, and making sure that tax administrators know how to audit local and multinational companies." (World Bank 2018)*²⁵

Total ODA for capacity-building and national planning amounted to \$20.4 billion in 2016 (18% of total aid).²⁶

A \$12 trillion business opportunity

It may be clear that the massive investment gap will not be met by public finances alone and higher private sector investment will be indispensable.²⁷ Governments, corporations and organizations are now exploring how to make an impact and how to measure progress towards achieving the SDGs. According to a report by the Business and Sustainable Development Commission (BSDC), achieving the SDGs will open up \$12 trillion of business opportunities in food and agriculture, cities, energy and materials, and health and well-being.²⁸ But as UNEP FI concludes:

"(...) private finance is constrained by risk and return requirements, while public finance is in scarce supply. If the resulting financing gap remains unresolved, investment needs will grow over

time because of a cumulative effect. Should we conclude that the SDGs are beyond reach? Or could business models be rethought in ways that would increase SDG serving financial flows, but also make them less risky?"²⁹

The next section will explore the need to align tax policy with the SDGs and shift financial incentives towards sustainable and inclusive business models. In recent years, corporate income taxes have been in the public eye. This study focuses on two other types of taxes that are less publicised but directly related to today's socio-economic challenges: labour taxes (including social contributions) and environmental (or 'green') taxes.

1.2 The 'polluter pays' principles are not applied

The linear economy

Today's linear 'take-make-waste' industrial model is based on the extraction of natural resources, which are used to create products that become waste after a short lifespan (see Figure 2). This growth model has brought wealth and prosperity, but still leaves many in poverty. Also, the negative side-effects of the model are becoming clear: climate disruption, waste, pollution, loss of biodiversity and scarcity of resources. These impacts are harmful for people, nature and economies worldwide, as waste and pollution are piling up. Every minute, for example, a truck load of textiles is landfilled or incinerated and every second, a truck load of plastics is dumped in the oceans.³⁰ Between 400,000 and one million people die each year in developing countries because of diseases related to mismanaged waste. At the upper end that is one person every 30 seconds.³¹

Figure 2: The linear 'take-make-waste' economy



External costs

The costs of environmental challenges, such as climate disruption and pollution, are immense. The Lancet Commission, for example, estimates global welfare losses from pollution at \$4.6 trillion a year, or 6.2 percent of global economic output.³² The long-term negative impacts on the global economy caused by carbon emissions in 2017 alone were \$16 trillion.³³ In practice, such costs are 'externalised', meaning that they are passed on to society, individuals and future generations, rather than absorbed by the polluter. In recent years, many studies have quantified external costs (or 'negative externalities'). A few examples are provided in Box 3.

Box 3: External costs studies

Plastics. The economic costs (e.g. revenue losses to fisheries, aquaculture, and marine tourism industries) associated with ocean-based consumer plastic pollution amounts to \$13 billion per year. (UNEP 2014).³⁴

Air pollution. Taking into account only premature deaths, air pollution costs the world economy \$5 trillion in welfare losses. The losses are greatest in East Asia and the Pacific, where they amount to 7.5% of GDP, and South Asia, at 7.4% of GDP. (World Bank and Institute for Health Metrics and Evaluation, 2016)³⁵

Climate disruption. Southern Asia and Western Africa will be most affected by rising temperatures, with productivity losses equivalent to 4.8% and 4.6%, corresponding to around 40 million and 9 million full-time jobs, respectively. (ILO 2018) ³⁶

Impacts on businesses. Government inaction on climate change will exacerbate costs and risks, to the tune of \$1.2 trillion for 30,000 of the largest listed companies. (UNEP FI 2019)³⁷

Considering issues like climate disruption, water scarcity and geopolitical tensions over fuels and materials, it would be rational to put a price on pollution and resource use.³⁸ Pollution and resource use, however, tend to be relatively tax-free, or even subsidised, as will be addressed next.

Environmental/green taxes

The OECD identifies environmental taxes as taxes on energy, transport, pollution and resources.³⁹ Throughout this paper, 'green taxes' will refer to all tax measures that put a price on the use of a natural resource. Green taxes are considered growth-friendly, as they are less distortive to the economy than taxes on labour and income.⁴⁰ The administrative costs and transaction costs of green taxes are lower than other taxes (notably income taxes).⁴¹ Environmental taxes tend to be much easier to collect than many other taxes, especially direct income taxes. Also, environmental taxes can be levied on a small number of taxpayers, especially upstream taxes on fossil fuel extraction or import. Reducing the number of taxpayers can reduce the overall costs of compliance of the tax system.⁴²

Taxation vs regulation

Unlike emission trading systems or other regulation, green taxes provide more opportunity for recycling the revenues through reducing other taxes to compensate for any welfare loss that may occur. If effectively implemented, cap-and-trade systems can provide greater certainty on environmental outcomes, but green taxes avoid fluctuations in price and provide consumers and businesses with greater certainty on investment decisions.⁴³ Green taxes are also more economically efficient than direct regulation. MIT's *Global Change* programme found that higher gas taxes are at least 6 to 14 times more cost-effective than stricter fuel standards at reducing gasoline consumption.⁴⁴

Green tax use limited and on declining trend

Despite their benefits, the use of green taxes is limited and on a declining trend. Among OECD countries, environmental tax revenues grew in revenues between 2000 and 2016 (from \$471 billion to \$742 billion) but declined as a share of total tax revenues (from 5.8% to 5.3%) and as a share of GDP (from 1.8% to 1.6%). Environmental tax as a share of GDP declined in 62% of the countries in the OECD database over the 2000-2016 period (see Figure 3).⁴⁵



Figure 3: Environmental tax as a % of GDP (2000-2016)

*2015, **2014. Source: OECD (Accessed May 2019), Global Revenue Statistics Database.

Low green tax revenues around the world

Green tax revenues are low around the world. In Canada, for example, environmental taxes raise 3.5 percent of total tax revenues, in the United States 2.6 percent, and in Brazil 2 percent. Data available from Asian economies also show modest green-tax revenues: 3.6% of total tax revenues in China; 12.7% in India; 4.5% in Japan; 10.6% in Korea; 1.5% in Malaysia; 11.3% in Kazakhstan; and 1.3% in the Philippines for example. In Africa, South Africa raises 5.4% of budget through green taxes, Rwanda 8.0%, and Cameroon 4.1%.⁴⁶

Fossil fuel subsidies \$373 billion

Besides levying relatively low tax levels on pollution, almost all nations apply direct and indirect subsidies for environmentally damaging activities.⁴⁷ Tax credits – defined as a subsidy by the WTO⁴⁸– are a key route of support for the fossil fuel industry.⁴⁹ Support measures are typically provided for fossil fuel consumption and production through lower rates, exemptions, or rebates of VAT and excise taxes.⁵⁰ The OECD has identified more than 1,000 individual government policies that support fossil fuel production and consumption.⁵¹ In 2009, G20 and Asia-Pacific Economic Cooperation leaders agreed to phase out inefficient fossil fuel subsidies in the medium term.⁵² In 2012 global fossil fuel support measures were worth \$617 billion. By 2015 they had gone down, but still amounted to \$373 billion. The majority of these measures are tax expenditures.⁵³

Fossil fuel subsidies \$5.2 trillion when taking into account external costs

IMF researchers recently updated their comprehensive report on global fossil-fuel subsidies. They found that, by the common definition of the term, governments subsidized fossil fuels by \$296 billion in 2017. The report adds that there is another kind of subsidy, though, which it calls a *post-tax subsidy*. This subsidy reflects the difference between actual fuel consumer prices and the full societal and environmental costs of a fuel. Burning of fossil fuels, the researchers argue, "releases deadly air pollution, hastens the destruction of the climate, and (sometimes) increases traffic fatalities. And since all of those things kill people, they also depress a country's tax base." When taking into account such negative effects, fossil fuel post-tax subsidies are projected as high as \$5.2 trillion (6.5 percent of GDP) in 2017.⁵⁴

Fossil fuel subsidies are regressive and distortive

There can be good reasons for governments to make energy more affordable, particularly for the poorest and most vulnerable groups. In practice, however, according to the Asian Development Bank (ADB), fossil fuel subsidies are highly *regressive* as they typically benefit medium- to high-income households, which are bigger energy consumers.⁵⁵ The OECD states:

"Not only do fossil-fuel subsidies undermine global efforts to mitigate climate change, but they also aggravate local pollution problems, causing further damage to human health and the environment. They represent a considerable strain on public budgets as well, draining scarce fiscal resources that could be put to better use, such as strategic investment in the education, skills, and physical infrastructure that people value most in the 21st century. Last, fossil-fuel subsidies distort the costs and prices that inform the decisions of many producers, investors, and consumers, thereby perpetuating older technologies and energy-intensive modes of production."⁵⁶

Countries are not sufficiently taxing environmental externalities, as they are "systematically pricing fuels too low."⁵⁷ UN Secretary-General António Guterres says it most clearly:

"Fossil fuel subsidies mean using taxpayers' money to boost hurricanes, spread droughts and melt glaciers."⁵⁸

Carbon pricing on the rise but prices are low

As of 2019, 57 national and sub-national jurisdictions around the world have implemented or scheduled carbon pricing schemes; 29 have opted for a carbon tax. Governments raised approximately \$44 billion in carbon pricing revenues in 2018, with more than half generated by carbon taxes. This is an increase of nearly \$11 billion compared to the previous year. Still, the vast majority (80 percent) of all global greenhouse gas emissions are free of charge. About half of the emissions covered by carbon pricing

mechanisms are priced at less than \$10 per tonne. Only 5 percent of emissions covered under a pricing mechanism are priced at a level consistent with meeting the goals of the Paris Agreement.⁵⁹

The benefits of action

Many studies underline the multiple benefits of effective environmental action. A few examples:

- Restoring the oceans could result in being able to feed a billion people a healthy seafood meal each day.⁶⁰
- Elimination of fossil fuel subsidies would have created global net economic welfare gains of more than \$1.3 trillion in 2015. At the same time, air pollution deaths would have been 46 percent lower.⁶¹
- Bold climate action, including carbon pricing, could yield a direct economic gain of \$26 trillion through to 2030.⁶²

Despite the clear benefits of healthy oceans, clean air, fresh water and, natural resources are structurally overutilized. In our economies, at the same time, human potential is underutilized. Addressing the need for decent jobs will be key in the SDG era. Unfortunately, our tax systems do not support job creation effectively yet, as will be explored in the next sections.

1.3 Labour taxes on the rise

Tax burden on labour high

Consistent with the classification of the European Commission, this study identifies labour taxes including personal income tax, social contributions and payroll taxes.⁶³ In 2016, in the 36 member countries of the OECD, labour taxes accounted for 51.1 percent of total tax revenues. Such labour taxes are on a rising trend. Labour taxes as a share of GDP increased in 73% of the countries in the OECD database over the 2000-2016 period (see Figure 4).⁶⁴



Figure 4: Labour taxes as a % of GDP (2000-2016)

Source: OECD (Accessed May 2019), Global Revenue Statistics Database. Labour taxes are assumed to consist of the sum of the following line items: (1100) taxes on income of individuals, (2000) social security contributions and (3000) payroll taxes. Costa Rica's earliest available data for PIT: 2003.

Labour tax wedge increased

The *labour tax wedge* is a measure of the tax burden on employment incomes. It is the difference between the employer's cost of hiring a worker and the worker's net disposable income. Between 2009 and 2016, the average labour tax wedge across the OECD increased by 0.7 percentage points, to 36 percent.⁶⁵ This means that, on average, of every dollar an employer pays in labour costs, only \$0.64 ends up in the pocket of the employee.

Labour versus green tax revenues across continents

In general, while OECD countries rely more on labour taxes, African, Asian, Latin American and Caribbean countries may rely more on taxes on goods and services. Still, labour tax revenues provide significant shares of revenues in these regions, and substantially more than green taxes. In Brazil, labour taxes provide 36 percent of total tax revenues, whereas green taxes raise 2 percent of revenues. In Costa Rica this ratio is 45:10, in Mexico 36:2. Such differences in proportions can also be found in Africa, with South Africa raising 36% on labour and 5% on green taxes. In Rwanda the ratio was 29:8 and in Cameroon 15:4. For Asia, the OECD database includes full data sets for Japan (59:1), Korea (41:11), the Philippines (28:1), Malaysia (15:1), Singapore (17:2) and Kazakhstan (21:11).⁶⁶

Links between labour taxes and employment

Labour taxes have unintended side-effects. Of all taxes, they have the clearest and most direct impact on employment.⁶⁷ High taxes on labour income can hamper both job creation and work incentives.⁶⁸ Research has demonstrated that a lower tax burden on labour creates employment opportunities.⁶⁹ The influence of taxation on employment and unemployment was, for example, significant in a sample of 21 OECD countries. Between 1983 and 2003:

"a 10 percentage points reduction of the tax wedge in an average OECD country would reduce equilibrium unemployment by 2.8 percentage points and increase the employment rate by a larger 3.7 percentage points (due to the positive impact on participation)". (Quoted in Dolenc and Laporšek 2010)⁷⁰

According to the OECD, low-income workers, single parents, second earners and older workers are especially responsive to changes in labour income taxation. The retirement decision of older workers is also highly responsive to tax incentives.⁷¹ In general, both the decision to enter the labour force and the hours worked are affected by labour taxes.⁷² Over the years, many researchers and international institutions have called for lower labour taxes to reduce unemployment - see for example (staff) contributions published by the OECD (2018; 2011),⁷³ World Bank (2015),⁷⁴ IMF (2015, 2014),⁷⁵ the European Commission (2017, 2013)⁷⁶ and the Eurogroup (2014).⁷⁷

Incentives to minimise labour input

For employers, labour taxes increase payroll costs - in addition to net salaries. Such costs tend to encourage employers (both in business and public services such as education and healthcare) to gain efficiency by minimising the number of employees. In general, employers may resort to a number of options to reduce labour input, including:

- Reducing service levels provided to customers (e.g. in hotels,⁷⁸ or selling products without additional repair and maintenance services)
- Replacing manual service with machines (e.g. self-checkout lanes in supermarkets⁷⁹ and automated assembly lines or 'sewbots').⁸⁰
- Shifting to mass production rather than bespoke, custom-made products (e.g. massproduced versus hand-made shoes).⁸¹
- Hiring short-term, informal workers rather than workers on a permanent contract (hiring interns⁸² or self-employed persons)⁸³ or operating in the gig- economy.⁸⁴
- Understaffing (putting pressure on workers to produce more in less time).85
- Shifting production to ever-lower income countries.⁸⁶

Automation risks particularly high in South Asia

Cost considerations can also drive the replacement of human roles with automation and artificial intelligence.⁸⁷ ILO research found that 137 million workers in South Asia are at a high risk of losing their jobs to automation in the next two decades. Especially garment workers face a high automation risk.⁸⁸ This trend makes it even more important to foster inclusive economies in which labour demand is sufficient to enable people whose tasks or jobs are taken over by machines to find new roles.

Human capital in abundance

With roughly 83 million people being added to the world's population every year, the upward trend in population size is expected to continue.⁸⁹ This means that there is a growing abundance of human capital (talents and capabilities).

One of the tasks in the SDG era is to accommodate for the full capacity of human potential and provide meaningful employment to a growing world population. Tax could play a significant role. Avoiding a high tax burden on labour while boosting social protection seems to be key. An option for financing such strategy is to increase the tax burden on pollution and resource use. This would enable the 'inclusive circular economy' to grow, as will be discussed next.

1.4 Updating tax systems to support the inclusive circular economy

Tax reform principles

As mentioned before, many tax systems increasingly penalize human input while incentivizing the depletion of resources. Environmental Tax Reform (ETR) (also known as Green Tax Reform (GTR), Value Extracted Tax, or 'Ex'tax') is the proposal to shift financial incentives, by putting a price on natural resource use and pollution and using the revenues to lower the tax burden on labour and increase (social) spending (see Figure 5). Since both ETR and GTR put emphasis on environmental rather than social impact, the broader term 'tax reform' will be used throughout this document.⁹⁰

Figure 5: Value Extracted Tax (Ex'tax) principles





Support for tax reform

Over the years, many researchers and international organisations have called for tax reform - see for example (staff) discussion papers published by the IMF (2016; 2015; 2012),⁹¹ ILO (2018, 2012, 2009),⁹² and World Bank (2015)⁹³ as well as IPCC (2018),⁹⁴ OECD (2018; 2015; 2013),⁹⁵ the European Parliament (2015; 2013; 2012)⁹⁶ and the European Commission (2018; 2017; 2016; 2015).⁹⁷ A selection of quotes is provided in Box 4. Business groups such as WBCSD,⁹⁸ the Ellen MacArthur Foundation⁹⁹ and the BSDC ¹⁰⁰ have also supported tax reform. According to the European Commission, ¹⁰¹ a shift from labour to green taxes is 'a winning strategy' and United Nations Secretary-General António Guterres has stated most clearly:

*"First, let's shift taxes from salaries to carbon. We should tax pollution, not people. Second, stop subsidizing fossil fuels."*¹⁰²

Box 4: Support for shifting from labour to green taxes

"offsetting increased carbon prices with lower labour taxes can potentially decrease labour costs (without affecting salaries), enhance employment and reduce the attractiveness of informal economic activity)." - IPCC 2018¹⁰³

"Increased or more effective use of environmentally related taxes can drive growth-oriented reform by shifting the tax burden away from more distortive taxes, e.g. on corporate or personal income, and contribute to fiscal consolidation—" - **OECD 2015**¹⁰⁴

"taxing 'bads' (pollutants) rather than 'goods' (labor, capital) can allow for a less costly tax system (...). Revenues can also be used to reduce the social charges imposed on labor costs. This may reduce unemployment rates and help increase real wages." - High-Level Commission on Carbon Prices, led by Nobel Laureate Joseph Stiglitz and Lord Nicholas Stern 2017¹⁰⁵

More statements in support of tax reform are available at www.ex-tax.com.

From a linear to a circular economy

In recent years, the concept of the 'circular economy' has gained traction and governments and institutions are adopting circular economy goals, which will likely boost the need for tax reform. A circular economy entails a shift from today's linear industrial model to a carbon-neutral and regenerative model in which products are 'made to be made again'. The Ellen MacArthur Foundation defines a circular economy as follows:

"an industrial system that is restorative or regenerative by intention and design. It replaces the "end-of-life" concept with restoration, shifts towards the use of renewable energy, eliminates the use of toxic chemicals, which impair reuse, and aims for the elimination of waste through the superior design of materials, products, systems, and, within this, business models".

In this way, finite resources and materials are not wasted, and businesses can add value over and over again by applying business models such as repair and maintenance services, recycling, refurbishment and remanufacturing (as illustrated by Figure 6). The circular economy is now an official goal of the 28 countries of the European Union¹⁰⁶ as well as the governments of China,¹⁰⁷ Germany,¹⁰⁸ France,¹⁰⁹ Scotland,¹¹⁰ Slovenia,¹¹¹ Portugal, ¹¹² Italy, ¹¹³ Finland¹¹⁴ and Greece.¹¹⁵ The Netherlands aims for a 50 percent reduction in the use of primary raw materials (minerals, fossil and metals) by 2030 and a fully circular economy by 2050.¹¹⁶ Several of these national and regional plans specifically refer to tax reform as a means to achieve a circular economy.¹¹⁷

Figure 6: Shifting from a linear to a circular economy



From a linear system ...

A circular economy is inclusive by design

There is a strong link between labour taxes and the circular economy, as the development of sustainable products and services requires more time, effort, craftsmanship and R&D than linear, mass-produced goods. A World Bank study, for example, showed that clean energy (energy efficiency and renewable energy) creates three times as many jobs as oil and natural gas per million dollars of spending.¹¹⁸ ILO researchers project that under a circular economy scenario, worldwide employment would grow by 0.1 per cent by 2030 in comparison with a business-as-usual scenario.¹¹⁹ A circular economy is therefore as much about people and their livelihoods as it is about resource use, emissions and waste.

Climate risks to business \$1 trillion

Business leaders have become increasingly aware of the risks of the environmental megatrends. It is now clear that companies that use fewer resources and cause less pollution will in future be less vulnerable to resource supply risks. A recent report by CDP revealed that 215 of the biggest global companies have valued the climate risks to their businesses at almost \$1 trillion - with many likely to hit within the next 5 years.¹²⁰ The Financial Stability Board's TCFD has stated:

"Companies that invest in activities that are susceptible to climate-related risks may be less resilient to the transition to a lower-carbon economy"¹²¹

The role of business has changed

The role of business has changed significantly in the debate around carbon and the climate crisis, with businesses actively supporting the Paris Climate Agreement. In 2015, companies and investors with over \$11 trillion in assets signed the *Paris Pledge for Action*. Another example is the *We Are Still In Coalition*, whose members represent over \$9.4 trillion and comprises more than 3,500 cities, states, businesses and institutions that are working towards the U.S. targets under the Paris Climate Agreement.¹²² And by September 2019, a record 515 institutional investors managing \$35 trillion in assets had signed a call to governments to 'put a meaningful price on carbon' and 'end government subsidies for fossil fuels'.¹²³

Internal carbon pricing

In IEA's *World Energy Outlook*, carbon prices will reach \$75–\$100 per tonne of CO₂ by 2030, in a scenario consistent with meeting Paris Climate Agreement goals.¹²⁴ Many business leaders are acting in anticipation of carbon policies, which they believe are inevitable. In 2017, almost 1,400 companies factored an internal 'shadow price' on carbon into their business plans, representing an eight-fold leap over four years.¹²⁵ Unilever, for example, has been internally pricing carbon emissions since 2016, creating a fund of €50 million a year to reinvest in clean technology.¹²⁶ Applying such pricing has proven to shift investment decisions toward low-carbon options as they become more competitive than polluting options.¹²⁷ Most attention goes out globally towards carbon pricing, but more than 50 companies, including Colgate Palmolive Company and Diageo Plc, have established internal water pricing.¹²⁸

Evolution of business models

Driven by the risks of megatrends and driven by the opportunities of new technologies, new business models are emerging in every sector. Table 1 provides a selection of examples. Companies shift activities from classic linear business models towards urban mining (e.g. Teck and GemChina), towards reuse (e.g. Maersk, Adidas, BMW), towards resale models (e.g. H&M, Ikea, Ycloset) and towards clean energy (e.g. Vattenfall, Shell, Total, BP, Apple). In the apparel industry, for example, H&M has committed to use 100 percent recycled or sustainably sourced materials by 2030¹²⁹ and Kering aims to reduce its impact across its supply chain by 40 percent by 2025.¹³⁰ IKEA has announced its goal of being a fully circular business by 2030.¹³¹ This means IKEA products will be moving in a 'loop' rather than in a line that ends in landfill or an incinerator.

Table 1: Examples of business model innovation by sector

Food sector

Nescafé has planted 2 million native shade-providing trees within and around coffee farms to protect soil against erosion and improve soil fertility and water retention while offering opportunities for income diversification to the farmers, through the sale of timber and fruit.¹³² **Heineken** has built circular brewery in Mexico where every piece of waste is repurposed or recycled.¹³³

Tyson Foods Inc., the largest U.S. meat producer, is moving into the vegetarian protein business.¹³⁴

Energy

Total invests \$300 million to install solar capacity at 5,000 gas stations around the world.¹³⁵ **BP and Dupont** have developed a second-generation biofuel.¹³⁶

Vattenfall is building the first offshore wind farm without subsidies. 137

Tata Power Solar commissions the world's largest rooftop solar array. 138

Metals & mining

Teck recovers materials from e-waste¹³⁹

Tata Steel has developed technology that reduces CO_2 emissions in the steelmaking process by 50%.¹⁴⁰

SSAB, LKAB and Vattenfall are building a testing facility for steel using hydrogen.¹⁴¹

GemChina collects and recycles materials from discarded batteries.¹⁴²

ArcelorMittal steel plant pilot scheme converts harmful by-products into renewable energy.¹⁴³ **Goldcorp and Sandvik** convert a mine into an all-electric operation, which improves air quality, reduces noise pollution, and eliminates negative health effects that diesel drills have on miners.¹⁴⁴ **Glencore** has announced to cap its coal production at current levels after coming under pressure from investors.¹⁴⁵

Apparel

Levi's makes clothes from hemp and recycled jeans.¹⁴⁶

Adidas sold 1 million shoes made out of ocean plastic in 2017¹⁴⁷ and develops a running shoe that will be collected and repurposed from consumers.¹⁴⁸

North Face is cutting waste by selling refurbished coats. 149

Reebok trainers are comprised of a bio-based sole made from a corn-based plastic substitute, the insole is made from castor bean oil, the fabrics are undyed, and the packaging is made from recycled materials.¹⁵⁰

Ralph Lauren produces shirts made of plastic bottles and dyed through a process that uses zero water.¹⁵¹

H&M sells second-hand clothes.¹⁵²

YCloset offers an online clothing subscription service in China.¹⁵³

Other consumer goods

IKEA is renting furniture, is replacing the use of polystyrene with biodegradable packaging made of mushrooms,¹⁵⁴ and offers solar panel purchase and installation services.¹⁵⁵

Interface turns discarded fishing nets into carpet tiles.¹⁵⁶

Carlsberg glues beer cans together to abandon plastic rings.¹⁵⁷

Unilever launched an education campaign to help consumers save water in Brazil during the country's water shortage. The brand grew at nearly double the market rate.¹⁵⁸

11 brands (Amcor, Ecover, Evian, L'Oréal, Mars, M&S, PepsiCo, The Coca-Cola Company,

Unilever, Walmart and Werner & Mertz) are working towards using 100% reusable, recyclable or compostable packaging by 2025. Together this represents more than 6 million tonnes of plastic packaging per year.¹⁵⁹

Apple global facilities – retail stores, offices, data centres and co-located facilities in 43 countries, including the U.S., China and India – powered with 100% clean energy.¹⁶⁰

Transport & logistics

Deutsche Post/DHL is deploying electric vehicles.¹⁶¹

Central Japan Railway Company (JR Central) high speed train has a market share of 86% versus 14% for the airplane between Tokyo and Osaka (515 kilometres).¹⁶²

Maersk Line has developed a Cradle to Cradle Passport for ships, creating a detailed inventory that can be used for identifying and recycling the components.¹⁶³

United Airlines will be using biofuel -produced from household trash- to power some of their flights.¹⁶⁴

SkySails produces large kite-like sails that attach to bulk carrier ships, which can save 30 tonnes of CO_2 per day.¹⁶⁵

Qantas has trialled the world's first flight ever to produce no landfill waste as the airline embarks on a mission to cut out the use of 100 million plastic items from its planes.¹⁶⁶

Vehicles

Daimler aims to have 50 all-electric test trucks on roads by end of 2019.¹⁶⁷
Ford is to invest \$4.5 billion in electric cars.¹⁶⁸
BMW/Mini offers car-sharing services reusing batteries for flexible storage of renewable energy.¹⁶⁹
Volvo has built electric refuse truck model which can carry a gross weight of 27 metric tons.¹⁷⁰
Porsche has introduced its first fully electric sports car.¹⁷¹
Various

Italcementi has developed a special mix for porous and pervious pavements, roads, walkways and parking lots, specialised for rain and storm water management.¹⁷² Lafarge has developed cement with a 25–30% smaller carbon footprint.¹⁷³ AkzoNobel Eco Premium Solutions – products that have a significant, measurable sustainability benefit over the competition – account for 19% of sales.¹⁷⁴ ABN AMRO bank has committed to financing €1 billion in circular assets by 2020.¹⁷⁵ Credit Suisse Group, ING Bank, FMO and UNDP-UN Social Fund jointly provide funding for the circular economy, sustainable energy and social impact in Asia.¹⁷⁶ Philips aims to deliver 15% of total revenues from circular solutions and take back and repurpose all the large medical systems by 2020.¹⁷⁷ Johnson & Johnson, Procter & Gamble, Goldman Sachs, Nike, Starbucks, Salesforce, Steelcase, Voya Financial and Walmart commit to sourcing 100% renewable electricity.¹⁷⁸

Levelling the playing field

The business models mentioned above are generally more resource-efficient than 'business as usual' models, but at the same time, they require more R&D and human input, to organise take-back systems for products, new supply-chain management and improved service models. Such sustainable business models currently need to compete with activities that require tax-free pollution in order to be competitive. Therefore, in principle, applying the 'polluter pays' principles will level the playing field; what may be a disadvantage to a polluter may mean an advantage to cleaner and innovative solutions.

Tax reform is a potential tool to 'future-proof' economies. The risks and opportunities of tax reform are not evenly distributed, though, and depend on the details of implementation. The next section will look more closely at options in the implementation process and the impacts of tax reform.

1.5 Frequently asked questions

This section will briefly discuss some of the most frequently asked questions on tax reform. How does tax reform work? What are the impacts for the economy? Has it ever been put in practice? And does tax reform increase inequality?

1.5.1 How does tax reform work?

A step-by step approach

Simply put, tax reform requires the following steps:

Step 1: Assessing the pathway necessary to achieve national, regional and global ambitions.

Step 2: Putting a price on pollution and natural resource use.

Step 3: Using the revenues to lower the tax burden on labour, improve social protection (in particular addressing the needs of lower-income households) and increase public investments.

Step 4: Monitoring and adjusting policy measures over time.

During the process, engaging with businesses and the public ahead of any change and communicating the impacts in a transparent manner will be key.

Wide range of policy options

In practice, each country faces a unique set of circumstances and the 'low-hanging fruits' in terms of tax reform differ per country. The Policy Toolkit below was developed by The Ex'tax Project to assess potential options to raise revenues (Figure 7) and use revenues (Figure 8).

- Figure 7 provides more than a hundred tax base options available to governments to apply the 'polluter pays' principles and raise revenues based on natural resource use and consumption (e.g. air pollution, energy, fossil fuels, metals and minerals).¹⁷⁹ Each category includes several sub-categories. For example, pricing schemes could be focused on different types of air pollution, such as carbon, particulate matter and/or nitrogen oxides.
- Figure 8 shows policy options to use tax revenues for the benefit of society (e.g. investments, social assistance and environmental protection). Each category includes several sub-categories. Investments, for example could, amongst others, be directed towards infrastructure, R&D, renewable energy, housing or coastal protection.

Beyond carbon pricing

In international research, most attention goes out to carbon pricing. It is important to keep in mind, though, that carbon emissions, water and materials usage are strongly interconnected. According to the OECD, half of carbon emissions are related to resource management.¹⁸⁰ The Toolkit therefore keeps a broad perspective. Also, a common argument against carbon taxes is that the tax drives carbon emissions down, and therefore, erodes its own tax base, so that there is less and less to tax. This challenge can be solved by increasing rates progressively according to a pre-announced schedule and expanding the tax bases. The Toolkit illustrates that governments have plenty of options to raise revenues, and to make up for any potential eroding tax base.

Climate damages payments

It is important to note, that in addition to putting a price on pollution and resource use in national systems, there is a global dimension that needs to be addressed, since many poor countries are being significantly harmed by the impacts arising from wealthy countries' carbon emissions.¹⁸¹ Some form of climate damages payments will be needed to raise funding for adaptation measures in low-income countries.¹⁸²

Fossil fuels		Energy		Food production		Air pollution	
Diesel	Coal	Electricity	Biofuels & waste	Fertilizer	Pesticides	Carbon (CO2)	Particulate Matter (PM)
LPG (Propane)	Crude oil	Hydropower	Renewables	Fish	Meat	Laughing gas (N2O)	Methane (CH4)
Natural gas (Methane)	Gasoline	Solar energy	Wind energy	Manure	Palm oil	Sulfur dioxide (SO2)	Nitrogen oxides (NOx)
Jet fuel	Tar sands	Biomass	Geothermal energy	Growth hormone	Antibiotics	Fluorine compounds	Ammonia (NH2)
Shale gas	Peat	Waste incineration	Nuclear energy	Deforestation	Soy	Heavy metals	Volatile Organic Compounds
Building	materials	Vari	ous	V	VAT		minerals
Cement	Crushed rock	Fossil Fuels (non-energy use)	Land use	Consumer goods	Energy products	Non-metallic minerals extraction	Salt
Steel	PVC	Fossil Fuel subsidies	Helium	Food products	Other products	Phosphate	Extraction of metalli minerals
Clay	Sand	Plastics	Noise	Specific services	Polluting products	Selenum	Uranium
Gravel	Wood	Odours	Biodiversity loss	Products of scarce materials	Water consumption	Platinum Group Metals	Zinc
Glass	Asbestos	Logging	Toxic chemicals			Rare Earth Metals	Other
Wa	aste	Water		Traffic & transport		Ecosystem services	
Landfill	Food waste	Groundwater	Surface water	Plane tickets	Air Freight	Carbon sequestration	Biodiversity
Electronic waste	Nuclear waste	Tap water	Water footprint	Road traffic	Road transport	Hydrological services	Pollination
Construction waste	Packaging	Bottled water	Water pollution	Congestion	Registration fees	Fertile soil	Prevention of erosion
Incineration	Sewage	Waste water		Maritime transport		Flood control	Recreational value
Mining waste	Toxic waste					Medical value	Water & air purification

Figure 7: The Ex'tax Toolkit – Raising Revenues

Source: Adaptation from The Ex'tax Project, Deloitte, EY, KPMG, Meijburg and PwC (2016), New Era. New plan. Europe. A Fiscal Strategy for an Inclusive, Circular Economy.

Figure 8: The Ex'tax Toolkit – Use of Revenues

Investments		Health/health care		Labour market		Other	
Infrastructure	Coastal protection	Immunization	Care at child birth	Employment scheme	ІСТ	Nutrition/ food security	Electricity production
Research & development	Renewable energy	Fertility	Safe drinking water	Jobs in manufacturing	Support to informal waste collectors	Sanitation/latrines	Telecom
Electricity coverage	Public housing	Primary care	Secondary cares			Gender issues	Support for SMEs
Energy efficiency	Public transport	Tertiary care	Quaternary care			Fossil-free cooking	Waste collection
Educatior	n/training	Environmental protection		Social as	ssistance	Social security	
Elementary school	Secondary school	Industrial effluents	(Urban) wetlands	Social protection (unspecified)	Income support	Unemployment insurance	Health & disability insurance
Post-secondary education	Vocational training	Forest maintenance	Biodiversity protection	Disability benefits	Child/family benefits	Maternity	Pensions (old-age &
		Carbon sequestration	Sustainable agriculture	Cash transfers	In-kind transfers	Employment injury insuarance	Insurance against bankruptcy
		Water management	Ocean plastics	Means-tested benefits	Non-contribu- tory benefits	Loss of spouse or parent insurance	
Budget res	tructuring	Personal income tax		Corporate income tax		VAT/Sales tax	
Deficit reduction	Debt reduction	Exemptions	Allowances	Rates	Deductions	Services	Product groups
		Rates	Deductions	Exemptions	Allowances	Products	
				SMEs			

Source: Adaptation from The Ex'tax Project, Deloitte, EY, KPMG, Meijburg and PwC (2016), New Era. New plan. Europe. A Fiscal Strategy for an Inclusive, Circular Economy.

1.5.2 What are the impacts of tax reform for the economy?

Theoretical debate

The foundations of modern tax systems were laid down in the era of the industrial revolution; before globalisation and mass consumption, before the emergence of climate disruption and water supply risks, and before digitisation and automation. Economies and societies have now become highly complex and interconnected. Because of this complexity, modelling the impacts of multiple policy measures is not easy. Since the 1990s there has been much discussion among economists around the different theories and methods used to assess the economy-wide effects of tax reform. The discussion revolves around the question under which circumstances tax reform could contribute to environmental, economic and employment benefits. Studies tend to focus on a variety of countries and use different models, scenarios and assumptions, which makes it difficult to compare the outcomes.¹⁸³ Over the years, a large body of research has emerged, though, showing that the costs of inaction are higher (see sections 1.1 and 1.2), and that carefully designed tax reform can actually have multiple benefits.¹⁸⁴

The many 'dividends' of tax reform

Pigato (2019), published by the World Bank, provides an extensive overview of the considerable benefits of tax reform found in the literature, and summarizes that tax reform can help countries reap a 'triple dividend' of cutting pollution, raising economic activity, and generating and funding public goods and development co-benefits (such as direct improvements in human health). Green taxes, for example, tend to be much easier to collect than many other taxes, especially direct income taxes; therefore, shifting from income taxes to green taxes can reduce the costs of tax evasion. Also, green taxes can be levied on a small number of taxpayers, especially upstream taxes on fossil fuel extraction or import. Reducing the number of taxpayers can reduce the overall costs of compliance of the tax system. Given such considerable benefits, tax reform should be an integral part of all modern fiscal systems, the report states.¹⁸⁵

Tax reform particularly desirable in developing countries

Tax reform is considered particularly desirable in developing countries, as the benefits reflect several features that are common in developing countries, including:

"(i) large informal sectors, which create opportunities to increase employment and output by using ETR revenues to reduce formal-sector taxes; (ii) inefficient tax systems, which create opportunities for ETR to reduce tax distortions, broaden the tax base, and tax rents rather than profits; and (iii) low levels of domestic taxation, which create opportunities for ETR to mobilize domestic resources to fund growth-enhancing public investment. These factors make it likely that ETR will raise measures of economic activity (for example, output and employment) more in developing countries than in developed countries."

Development co-benefits tend to be particularly large in developing countries:

"By discouraging pollution-intensive activities, environmental taxation can promote improvements in air quality and public health, alleviate costly traffic congestion, and reduce the frequency of road accidents. (...) Meanwhile, the revenue from environmental taxes can help the economy to adapt to climate change and fund increased spending on education, infrastructure, social services, and other public goods."

As a result, co-benefits alone may justify the use of tax reform in developing countries, even before the benefit of climate mitigation is considered.¹⁸⁶

Impacts on competitiveness

Contrary to popular belief, existing literature (mainly focused on high-income countries) has found little evidence of adverse competitiveness effects from tax reform. Any adverse effects tend to be small and concentrated in a few energy-intensive and trade-exposed sectors. If anything, tax reform incentivizes firms to adopt or invent more efficient technologies. Increases in fuel prices faced by firms across 11 upper-middle-income countries in Latin America and Eastern and Central Europe, as well as Indonesia and Mexico are associated with improvements in labor productivity and profitability. Also,

competitiveness losses for some types of firms or sectors may be more than offset by gains in others. In addition, there is little empirical evidence of significant carbon leakage (when firms relocate to less regulated markets).¹⁸⁷ The European Emission Trading System (EU ETS) induced about 10 percent carbon emission reductions between 2005 and 2012 but had no negative impact on the economic performance of regulated firms. In fact, the carbon pricing mechanism led to an increase in regulated firms' revenues and fixed assets.¹⁸⁸

1.5.3 Where has tax reform been put in practice?

The most well-known and documented tax reforms have taken place in Europe and in British Colombia:

The case of Europe

In the 1990s and early 2000s, seven European countries took steps to shift the tax burden from labour to energy and transportation. In total, these reforms increased green tax revenues by more than \in 25 billion annually, for a corresponding decrease in labour taxes. The associated reductions of carbon emissions have been documented in several studies. The burden for specific energy-intensive industries remained modest (1%-2% increase in energy costs) and the tax shifts generally had a positive effect on economic activity, depending on how the revenues from the green taxes were recycled. Also, the reforms caused employment in some of the countries to increase by as much as 0.5 percent.¹⁸⁹

The case of British Colombia

In 2008, the Canadian province of British Columbia started taxing fossil fuel users, in exchange for business tax cuts and tax credits, personal income tax cuts (targeted at lower income categories), low-income tax credits and reductions in property taxes.¹⁹⁰ The tax credit for low-income households has made the carbon tax progressive.¹⁹¹ Since then:

"the economy has grown by an average of nearly 2 percent a year, despite a big national recession through 2009, outpacing the rest of Canada. The use of gasoline, coal and other carbon-based fuels has dropped 16 percent during the same period, reducing greenhouse gas pollution." (Scientific American 2015)¹⁹²

Employment in the most carbon-intensive industries fell, but this was more than offset by increases in other industries, and net employment rose 0.74 percent between 2007 and 2013.¹⁹³ As of 2019, carbon pricing applies throughout Canada.¹⁹⁴

Key learnings for implementation

Numerous studies have looked at the prerequisites for successful implementation of -especially- carbon pricing. Recommendations include building public trust and low corruption levels, for example.¹⁹⁵ Developing countries are advised to consider taxing upstream rather than downstream. Upstream fuel taxes tend to entail taxing fewer entities (importing or extracting firms) compared to downstream emissions taxes (such as taxes on the pollution released from individual entities burning the fuel), which minimizes the administrative burden, both for firms and for government bureaucracy.¹⁹⁶ In some countries, avoiding the term 'tax' and instead using the term 'CO2 levy', or 'fee and dividend' could make measure more acceptable to citizens. Overall, there seems to be a consensus that reforms should be introduced gradually, and public acceptance should be built with an information campaign.¹⁹⁷ Drawing on international research, especially in Canada, the *Guide to Communicating Carbon pricing* recommends that "communicators foreground the way that revenues are used and the benefits they generate. For this reason, decisions around revenue use will be critical for how the policy is received."¹⁹⁸ It may be clear that countries will adopt the principles of tax reform only when there is long-term vision, good governance and accountability.

1.5.4 Does tax reform increase inequality?

Progressive tax reform

An often-heard worry is that environmental taxes could increase income inequality: they hit low-income households more, as they pay higher shares of their incomes towards energy-intensive goods. It is possible, however, to prevent taxes from increasing income-inequality if the proceeds are used to benefit the poorest population.¹⁹⁹ Also, against popular belief, fuel and coal taxes can have progressive effects. A meta-analysis of 21 country studies found that taxes on motor fuels were progressive in 10 of the 12 developing countries studied. Overall, concerns about distributional equity and poverty are not strong justifications for maintaining low green taxes. Even in instances where the poorest lose out in relative terms, only a small portion (up to a third) of green revenues is needed to compensate lower-income groups.²⁰⁰

Options to alleviate low-income groups

In practice, plenty of policy options are available to alleviate the impacts on households in need. Benefits can take the form of (means-tested) tax credits, exemptions, allowances or deductions. In some countries, cash transfers might ease the transition for the unemployed and those who live in poverty: the right solution will differ from one country to another. If desirable, green taxes can also be made more progressive by applying block tariffs (higher rates for higher use) or a tax-free threshold (e.g. leaving a certain amount of water or energy untaxed). Careful design and implementation can alleviate many, if not all of the concerns about discriminatory effects. Overall, it is important to consider all taxes (and transfer payments) together as a system:

"not all taxes need be progressive as long as the overall system is." (Mirrlees 2011)²⁰¹

Failing to tax externalities is regressive

It is important to note that failing to tax environmental externalities is regressive. Fossil fuel subsidies are highly regressive as they typically benefit medium- to high-income households, which are bigger energy consumers (see section 1.4). Furthermore, a large share of the welfare costs of environmental externalities falls on the poor, including increased mortality and morbidity, depletion of resources,²⁰² crop failures and loss of income.²⁰³

Policies to leapfrog into the SDG era

Leapfrogging is the concept of bypassing intermediate stages of technology through which countries have historically passed during the development process.²⁰⁴ A well-known example is how some developing countries have skipped the development of landline infrastructure by moving directly to mobile telecommunications.²⁰⁵ In Bangladesh in 2017, for example, the share of people with a mobile money account jumped from 3% to 21%.²⁰⁶ In the field of sustainability, similar leaps are made. Between 2012 and 2017, the Chinese city of Shenzhen electrified 16,000 diesel buses.²⁰⁷

Leapfrogging is generally associated with technological innovation but could also apply to policy development.²⁰⁸ Low-income countries often have a low tax take (if not rates) on labour taxes. As experience in higher-income countries shows, and given the high unemployment rates, increasing taxes on labour may not necessarily be the best option to increase revenues sustainably. Taxing the use of natural resources might be a way of leapfrogging tax systems to the SDG era and the development of social protection systems. For example, in 60 out of the 87 countries reviewed by the World Bank, a domestic carbon tax of \$30 per tonne of CO₂ would provide the resources to more than double current levels of social assistance in the country.²⁰⁹

How would tax reform work in a country like Bangladesh; a country in transition from low-income to middle-income status? A country with a large and growing population and most vulnerable to climate change? The next chapter will explore the extraordinary achievements and challenges of Bangladesh in terms of economic, social and environmental development. This information will serve as the backdrop to the policy scenarios developed in Chapter 3.

2. Bangladesh introduction

Bangladesh has demonstrated remarkable resilience in addressing severe predicaments. The country is still facing multiple challenges, though, including a need for economic development, jobs and social protection, while at the same time effectively addressing the effects of climate disruption, pollution and resource depletion. This section explores why an alignment of taxes with the SDGs is highly topical for Bangladesh.

The following section explores a selection of geographic, socio-economic, environmental and fiscal characteristics of Bangladesh. This is by no means a complete overview, but the selected information will help assess the needs and potential for tax reform.

2.1 Geography, population and climate

A densely populated delta

Bangladesh is a densely populated country in South Asia. It is bordered by India and has a small border with the Republic of the Union of Myanmar (also known as Burma). The major part of Bangladesh lies in the delta of the Ganges, Brahmaputra and Meghna rivers, which spring from the Himalaya mountain range. The rivers deposit their freshwater and sediment in the Ganges-Brahmaputra-Meghna delta, which historically supported high population densities through the provision of ecosystem services including highly productive farming and fishing systems.²¹⁰ The aggregate population of Bangladesh increased from 38 million in 1950²¹¹ to almost 165 million in 2017,²¹² and is projected to exceed 200 million by 2050.²¹³ 73 percent of the population live in rural areas.²¹⁴

A tropical monsoon climate

About 80 percent of the country is made up of fertile alluvial lowland. The country is flat with some hills in the northeast and southeast. About seven percent of the total area of Bangladesh is covered with rivers and inland water bodies and the surrounding areas are routinely flooded during the monsoon. Bangladesh has a tropical monsoon climate with four main seasons: 1) the pre-monsoon during March-May, which has the highest temperatures and experiences the maximum intensity of cyclonic storms; 2) the monsoon during June-September, when the bulk of rainfall occurs; 3) the post-monsoon during October-November which, like the pre-monsoon season, is marked by tropical cyclones on the coast; 4) the cool and sunny dry season during December-February.²¹⁵

A disaster-prone county

Bangladesh is one of the most disaster-prone countries in the world. On average, the country experiences severe tropical cyclone every three years, and about 25 percent of the land mass is inundated with flood waters every year. Severe flooding occurs every 4-5 years and covers 60 percent of the land mass.²¹⁶ Over the last decade, nearly 700,000 Bangladeshis were displaced on average each year by natural disasters, such as tropical cyclones, storm surges and floods.²¹⁷ Two-thirds of the country's land area is less than five meters above sea level.²¹⁸

Most vulnerable to climate disruption

Following devastating cyclones in 1970 and 1991, Bangladesh has made significant efforts to reduce its disaster vulnerability and is now considered a global leader in climate resilience.²¹⁹ Despite these efforts, the vulnerability of the population is on the rise due to climate change,²²⁰ which is causing rising (sea) temperatures, heavier rainfall, droughts,²²¹ salinization of soil and water,²²² intensified cyclones and sea level rises. Bangladesh is widely recognized as one of the countries that is most vulnerable to climate change.²²³ According to the Ministry of Environment:

"(...) extreme temperatures, erratic rainfall, floods, drought, tropical cyclones, rising sea levels, tidal surges, salinity intrusion and ocean acidification are causing serious negative impacts on the lives and livelihoods of millions of people in Bangladesh, and are gradually offsetting the remarkable socio-economic development gained over the past 30 years"²²⁴

Sheikh Hasina, Prime Minister of Bangladesh has stated:

"To our utter dismay, without contributing to environmental destruction, we are confronted with a situation where Bangladesh is one of the most climate-vulnerable countries in the world."²²⁵

"The risks are so grave that, for Bangladesh, climate change may displace more than 30 million people by 2050."²²⁶

2.2 Socio-economic development

A lower middle-income country

Despite the political turmoil that has characterized its history since independence from Pakistan in 1971, according to the OECD, Bangladesh is a 'competitive economy with ambitions to achieve middle-income country status'.²²⁷ Bangladesh has been a lower middle-income country under World Bank classification since 2015 and has officially started the UN graduation process out of the least developed category towards becoming a developing country in 2024.²²⁸

High growth rates and development

During the last decade, Bangladesh achieved on average more than 6% economic growth.²²⁹ In 2017, GDP was \$250 billion (or \$1,517 per capita).²³⁰ More than half of GDP is generated through the services sector.²³¹ Bangladesh is the world's second-largest exporter of clothing after China²³² and the ready-made garment (RMG) sector represents 80 percent of foreign earnings.²³³ Bangladesh has made remarkable progress on key development indicators with poverty falling from 49% in 2000 to 25% in 2015²³⁴ and life expectancy growing from 46 years in 1972 to 72 years in 2011.²³⁵

Poverty still widespread

Despite the aforementioned progress, one-fourth of the population lives below a poverty line of approximately \$29 per month.²³⁶ Income distribution in Bangladesh is very uneven. 53% of national income is owned by the richest 20% of the population. The 40% lowest income groups (representing 40% of the population), have only 13% of total income. More than one-third of children under five are stunted, limiting their ability to grow and learn.²³⁷

A growing need for decent jobs

The working age population (15-64 years) consists of more than 100 million people.²³⁸ The vast majority of the workforce (87%) is in the informal economy.²³⁹ Almost 40% of Bangladeshis are employed in the agriculture sector, with rice as the single-most-important product.²⁴⁰ The RMG (ready-made garment) sector is the country's largest employer, employing some four million people.²⁴¹ A few striking facts on the labour market (in 2017):

- Mean nominal earnings of employees are \$165 per month.²⁴²
- 56% of employment is 'vulnerable' (according to the ILO definition of own-account workers and contributing family workers who have a lower likelihood of having formal work arrangements).²⁴³
- 40% of the population is underemployed (working only a few hours a week and at low wages).²⁴⁴
- 11% of the *employed* live on less than \$1.90 a day.²⁴⁵
- There is extensive migration of labour to Saudi Arabia, Kuwait, UAE, Oman, Qatar and Malaysia.²⁴⁶
- Overall, 50 per cent of manufacturing workers in Bangladesh work excessively long hours (over 48 hours per week).²⁴⁷
- 1.7 million children (5-17 years) are engaged in child labor.²⁴⁸

The World Bank has identified job creation as the country's top development priority:

"Bangladesh needs to create more and better jobs for the 2 million youths entering the job market every year. To do so, Bangladesh will need to remove the barriers to higher growth posed by low access to reliable and affordable power, poor transportation infrastructure, limited availability of serviced land, uncertain and complex business regulation, rapid urbanization and vulnerability to climate change and natural disasters, among others."²⁴⁹

Electrification success

Bangladesh is considered one of the world's great success stories in access to energy. In 2017, electricity reached some 80 percent of the population, up from 20 percent in 2000.²⁵⁰ The government target is universal access to electricity by 2021.²⁵¹ The estimated cost of rural electrification 2015-2030 are \$3 billion.²⁵² Since 2013, Bangladesh has installed more than 4 million Solar Homes, ensuring the supply of solar electricity to 18 million people who previously used kerosene lamps for lighting purposes.²⁵³

Business climate

The World Bank's *Ease of Doing Business* survey put Bangladesh at 176th of 190 countries. The ranking is based on 10 areas in the life cycle of a business, including starting a business, getting electricity and credit, paying taxes and trading across borders.²⁵⁴ The World Economic Forum *Executive Opinion Survey* identified the five highest risks of doing business in Bangladesh being energy price shock, failure of national governance, cyber-attacks, failure of regional and global governance, and unemployment or underemployment.²⁵⁵ The OECD concludes:

*"Bangladesh has made remarkable progress on key development indicators (...). Nonetheless, inequality remains endemic and climate change, population growth and natural disasters continue to constrain progress. Corruption, the weak rule of law and limited transparency contribute to an insecure environment."*²⁵⁶

Cheap labour for the apparel industry

Bangladesh offers cheap labour for the apparel industry. The legal minimum wage for garment workers in the country is only 8,000 taka (\$95) a month. The amount was increased by 51 percent (from \$63) in December 2018, but campaigners say workers need at least 16,000 taka (\$190).²⁵⁷ The last time garment workers' salaries were raised was in 2013, after the Rana Plaza disaster, when an industrial building housing several garment factories collapsed, killing more than 1,130 workers.²⁵⁸ The legal minimum wage in Bangladesh is significantly lower than in other sourcing countries such as China, Vietnam and Cambodia, (where minimum wage are \$326, \$180 and \$182 respectively).²⁵⁹

2.3 Environmental issues

Carbon emissions relatively low but rising

Carbon emissions in 2017 were 0.51 tonne of CO_2 per capita in Bangladesh. By comparison, Americans emit 30 times more per person. When ranking counties by their share of global cumulative emissions, Bangladesh ranks 75th out of 226 countries. The main emitters of carbon in Bangladesh are the power sector (42% of total), other industrial combustion (21%), buildings (14%), transport (12%) and 'other' (12%).²⁶⁰ In the period 2004-2016 the annual average growth rate of carbon emissions was 9.2%.²⁶¹

The costs of climate disruption

Bangladesh is responsible for less than 0.1 percent of total cumulative CO₂ emissions in the world,²⁶² but the country is bearing the brunt of the devastating effects of climate disruption. The World Bank estimates that the adaptation costs of tropical cyclones and storm surges plus the cost of flooding will be around \$6.6 billion by 2030.²⁶³ The Ministry of Environment of Bangladesh estimates the country will need to invest \$40 billion from 2015 to 2030 to address the adverse impacts of climate change.²⁶⁴ The country has two climate change trust funds.²⁶⁵ The Bangladesh Climate Change Trust Fund (BCCTF), operational since 2010, is the first ever national climate fund established by a least developed country.²⁶⁶

Limited use of clean energy

Main energy sources are natural gas (56%), biofuels and waste (25%), whereas renewables such as hydro, geothermal and solar provide less than 0.1 percent.²⁶⁷ There is one hydropower station, which was installed in 1988.²⁶⁸ Some 80 percent of households (more than 140 million people) rely primarily on polluting fuels such as biomass for cooking. For lighting, 40 percent of households rely on polluting fuels such as kerosene.²⁶⁹

One of the highest incidences of air pollution

In 2017, annual exposures to PM2.5 (particulate matter fine enough to penetrate the lungs and bloodstream) were highest in South Asia. Nepal, India and Bangladesh had the highest exposures.²⁷⁰ In Bangladesh, the full 100 percent of the population is exposed to levels of air pollution exceeding guideline values set by the World Health Organization (WHO).²⁷¹ Bangladeshi would have an expected gain in life expectancy of nearly 1.3 years if they had met the WHO guideline.²⁷²

Access to water up, but water resources are severely polluted

Bangladesh has made remarkable progress with regard to access to water and sanitation. Between 1990 and 2015, the population experienced a 20-percentage point gain in water access and 29-percentage point gain in sanitation access. ²⁷³ The quality of water, however, is poor, as Bangladesh's water resources are heavily contaminated.²⁷⁴ Access to piped water supply was a mere 12% in 2016.²⁷⁵ Just 3% of the population is connected to a sewer system²⁷⁶ and only about half of manufacturing enterprises in Bangladesh have toilets.²⁷⁷

Climate crisis puts more pressure on water systems

Climate change is increasing the intensity and frequency of natural disasters that disrupt water and sanitation services. During times of disaster, about a third of households in the country's high-risk areas switch to contaminated, unimproved water sources.²⁷⁸ In 2002 Bangladesh became the first country in the world to implement a ban on plastic bags, after it was found they played a key role in clogging drainage systems during disastrous flooding.²⁷⁹

Industrial effluents

Around 250 industries are discharging chemical pollutants into the Buriganga and Sitalakhya rivers. Every day 4,000 tons of solid waste and 22,000 tons of tannery waste mixes with water in Buriganga river.²⁸⁰ In 2016, textile industries in Bangladesh generated around 217 million cubic meters of wastewater.²⁸¹ By 2021, the garments export target has been set at \$50 billion, but the success comes at a huge environmental cost:

"The dyeing and finishing plants are the major pollutants of water. Turag that flows by Tongi is almost dead with pollution. Its water looks ink black and gives out such a foul smell (Inam and Refayet, 2017). Wetlands around Dhaka city are being destroyed through land development and dumping of toxic effluents and untreated sewage. Industrial effluents have totally destroyed the ecology of rivers near these large urban areas (Shishir, 2018)."²⁸²

Rohingya refugees from Myanmar

In 2017, more than 723,000 Rohingya (a Muslim minority in Myanmar) fled to Bangladesh. The Kutupalong refugee settlement in the Cox Bazar district in Bangladesh has grown to become the largest of its kind in the world, with more than 600,000 people living in an area of just 13 square kilometres. The Bangladesh government has responded generously throughout the latest crisis, straining their already limited resources.²⁸³ In addition to the humanitarian challenges, the influx of refugees has resulted in environmental degradation both within the refugee camp and in the surrounding areas.²⁸⁴

2.4 Tax structure

Revenues

Table 2 provides an overview of the central government revenues in Bangladesh in 2016 (the latest year for which such data are available). As explained in section 1.1, tax revenues above 15 percent of a country's GDP are a key ingredient for economic growth and, ultimately, poverty reduction. Bangladesh has one of the lowest tax-to-GDP ratios in the world. In 2016, only ten countries in the World Bank database had a lower tax-to-GDP ratio.²⁸⁵ In that year, Bangladesh had a tax-revenue gap of \$14.9 billion.²⁸⁶ In 2017-18, tax revenue receipts amounted to \$25.5 billion, which was 17.3% higher than the previous year.²⁸⁷ The tax-to-GDP ratio of Bangladesh has been on the rise, albeit slowly; rising from 8.6% in 2010 to 10.9% in 2017.²⁸⁸ IMF staff advised Bangladesh:

*"To boost growth, higher public investment in critical infrastructure and human development is needed and for that purpose it is necessary to improve revenue generation."*²⁸⁹

	Revenue (\$ bln)	% of total revenue
Taxes	18.0	85%
Taxes on income, profits, and capital gains	5.3	25%
Payable by individuals	1.8	8%
Payable by corporations and other enterprises	3.6	17%
Taxes on payroll and workforce		-
Taxes on property		-
Taxes on goods and services	6.8	32%
General taxes on goods and services	6.5	30%
Excises	0.2	1%
Taxes on international trade and transactions	5.2	25%
Other taxes	0.6	3%
Social contributions	-	-
Grants from foreign governments	0.3	1%
Other revenue	3.0	14%
Total revenues	21.2	100%

Table 2: Bangladesh Budgetary Central Government Revenues (2016)

Source: IMF (2018), Government Finance Statistics Yearbook 2017.

Personal income tax

Total personal income taxes are only 1.4 percent of GDP,²⁹⁰ with less than one percent of the population filing tax returns.²⁹¹ Every year, a week-long Income Tax Fair is held across the country, with the aim of increasing the number of income tax payers²⁹² and November 30 marks Income Tax Day.²⁹³ The National Board of Revenue (NBR) then honours the highest paying citizens (in categories such as: highest paying architect, athlete, lawyer, the highest paying female, the highest paying individual aged under 40, etcetera) and corporations. The highest tax cardholders are entitled to a number of privileges, including priority while reserving seats in public transport and access to VIP lounges at the airports.²⁹⁴

Other observations

A few remarkable features of the Bangladeshi budgetary revenues are:

- There is **no concept of social security** in Bangladesh. However, companies of a certain size need to pay 5% of their profits into a Workers Profit Participation Fund. No contribution from employees is required in this case.²⁹⁵
- At 1.1% of total tax revenues, **excise duties are relatively low** compared to Pakistan (5%), Bhutan (13%) and India (16%).²⁹⁶
- Bangladesh has a complicated VAT system. The standard VAT rate is 15% and there are reduced rates of 10%, 7.5%, 6%, 5%, 4.5%, 4%, 3%, 2.25%, 1.5%, and 0%. A VAT rate of 15%, for example, applies to airconditioned restaurant, whereas 7% applies to unairconditioned restaurant.²⁹⁷
• Bangladesh is among the countries receiving a relatively large share of global development aid:

"In 2015-16, half of gross bilateral ODA expenditures (...) were directed at 8 countries out of the 48 LDCs [least developed countries]: Afghanistan (USD 3.4 billion), Ethiopia (USD 2.0 billion), the United Republic of Tanzania (USD 1.5 billion), Bangladesh (USD 1.4 billion), South Sudan (USD 1.3 billion), the Democratic Republic of Congo (USD 1.2 billion), Mozambique (USD 1.1 billion) and Uganda (USD 1.0 billion)." (OECD 2018)²⁹⁸

Budgetary expenses

Table 3 (below) provides an overview of the use of revenues by the central government. A few striking features from an SDG perspective:

- **Public social expenditure** is around 2% of GDP in Bangladesh, which is the second lowest in the Asia/Pacific region (after Myanmar).
- In Bangladesh, **public spending on education** amounts to less than 2% of GDP (which is half of the average spending across the Asia/Pacific region and the OECD).²⁹⁹
- Just 0.1% of the budget is used for environmental protection.³⁰⁰ Note that in general, in South Asian countries, environmental protection expenditures are low (e.g. 0.02% in Pakistan; 0.3% in Bhutan; and 1% in Nepal).³⁰¹

Table 3: Bangladesh Budgetary Central Government Expense by Economic Type (2016)

	Expense (\$ bln)	% of total
Compensation of employees	4.9	25%
Wages and salaries	4.9	25%
Employers' social contributions	-	-
Use of goods and services	2.7	14%
Consumption of fixed capital	-	-
Interest	4.0	20%
Subsidies	1.4	7%
Grants to other general government units	3.2	17%
Social benefits	2	10%
Social security benefits	-	-
Social assistance benefits	0.7	4%
Employment-related social benefits	1.2	6%
Other expense	1.3	7%
Total expense	19.5	100%

Source: IMF (2018), Government Finance Statistics Yearbook 2017.

Fiscal ambitions

Bangladesh has high fiscal ambitions, aiming to raise the number of registered taxpayers from 900,000 to 10 million between 2011 and 2023.³⁰² A number of targets have been set for fiscal year 2020, aiming to increase total revenues, government spending and Foreign Direct Investment (FDI), while maintaining the fiscal deficit.³⁰³ The status of these indicators is included in Table 4.

Table 4: Status of fiscal targets mentioned in the 7th Five Year Plan of Bangladesh

Indicator (% of GDP)	Status FY17-18	Target FY20
Total revenue (% of GDP)	12.2	16.1
Fiscal deficit (% of GDP)	5	5
Government spending (% of GDP)	17.5	21.1
FDI (\$ million)	1.583	9,600

Sources: Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission; Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018.

2.5 Progress on achieving the SDGs by 2030

Strategies to address the challenges

The *SDG Index and Dashboards* report presents an assessment of countries' distance to achieving the SDGs. In 2018, Bangladesh ranked 111th out of 156 countries.³⁰⁴ Table 5 provides an overview of some of the formal strategies developed by Bangladesh to address the SDGs.

Table 5: Selection of Bangladesh' national strategies to address SDGs

7th Five Year Plan FY2016- FY2020 ³⁰⁵ - update under development	Includes specific targets consistent with the scope of the SDGs, with regard to Income and poverty, Sector Development, Macroeconomic Development, Urban Development, Human Resource Development, Water & Sanitation, Energy & Infrastructure, Gender equality, income inequality & social protection, Environmental Sustainability & ICT Development.
Vision 2021 ³⁰⁶	Eight goals that reflect citizens' aspirations with regard to the future of Bangladesh by 2021, when Bangladesh will be celebrating 50 years of independence, including: 1) To become a participatory democracy, 2) To have an efficient, accountable, transparent and decentralised system of governance, 3) To become a poverty-free middle-income country, 4) To have a nation of healthy citizens, 5) To have a skilled and creative human resource, 6) To become a globally integrated regional economic and commercial hub, 7) To be environmentally sustainable and 8) To be a more inclusive and equitable society.
Bangladesh Delta Plan 2100 (BDP 2100) ³⁰⁷	In view of the long-term challenges presented by climate change and natural hazards, the BDP 2100 seeks to integrate the medium to long term aspirations of Bangladesh. The plan was formulated with the Dutch government and seeks to map out the land-use planning, and disaster management changes needed to meet growing pressures on the delta.
The Bangladesh Second Country Investment Plan (2016-2020) (CIP2) ³⁰⁸	The overarching goal of the CIP2 is to achieve improved food security and nutrition for all at all times by making food systems nutrition-sensitive and sustainable. It is a tool to mobilise funds and align sectoral and cross-sectoral food and nutrition security related programmes.
Bangladesh Climate Change Strategy and Action Plan (BCCSAP) ³⁰⁹	Sets out 44 programmes within six strategic areas – food security, social protection and health; comprehensive disaster management; infrastructure; research and knowledge management; mitigation and low carbon development; and capacity building and institutional strengthening.
Intended Nationally Determined Contributions (INDC) ³¹⁰	The INDC aims to put forth mitigation actions that Bangladesh can take to tackle its growing emissions and to play its role in global efforts to limit temperature rise to 2°C or preferably 1.5°C above pre-industrial levels. Targets include an unconditional GHG emissions reduction of 5 percent compared to business as usual levels by 2030 in the power, transport and industry sectors.
National Social Security Strategy (NSSS) of Bangladesh (2015) ³¹¹	Priority challenges for the medium-term to ensure a more comprehensive social protection system for its population. The NSSS envisions a Bangladesh where poverty and inequality are effectively tackled, growth and employment are efficiently accelerated, and the weak and vulnerable are adequately protected. ³¹²

The case for tax reform in Bangladesh

Based on the information above, Bangladesh has achieved much but is still facing significant challenges. The country has a large and growing population, which is in need of decent jobs and inclusive economic growth. Air and water pollution have grown to be detrimental for the economy, health and wellbeing. Although climate change is not the fault of the Bangladeshi people, the nation is already paying the price of the global climate crisis and climate disruption is high on the political agenda. As indicated by Ahmed and Chattopadhyay (2018), the case for tax reform in Bangladesh is clear. First, tax reform can help raise public revenues, which is attractive for Bangladesh as it has a very low tax-to-GDP ratio. Second, green taxes can be helpful in improving the environment. Third, fossil fuel subsidy reform can ensure resource efficiency and less pollution. Finally, the revenues mobilised can be a major contributor to public spending in infrastructure, health, education, water supply, environmental protection and social protection to support poverty reduction.³¹³

Taking into account the national context, how could a country like Bangladesh align its tax system with the SDGs? What would be the impacts of such alignment on the economy, labour market, inequality and business sectors? The next chapter will develop some ideas on a potential tax reform scenario and its impacts.

3. Developing tax reform scenarios

How could Bangladesh create a tax system that supports the SDGs? Applying the Methodology and Toolkit of The Ex'tax Project, two preliminary scenarios were developed, based on national priorities in Bangladesh, data availability and opportunities for macro-economic modelling.

3.1 The Ex'tax Methodology and Toolkit

The Ex'tax Methodology

In cooperation with tax experts of Deloitte, EY, KPMG and PwC, The Ex'tax Project has developed a Methodology and Toolkit to help assess country and regional tax reforms.³¹⁴ In this study, the step-bystep approach (see Figure 9) was applied to the Bangladeshi context. The first step in the Methodology is to identify priority challenges of the geographic area under review. Secondly, data are collected to explore the economic, social, environmental and fiscal landscape. Thirdly, the Ex'tax Policy Toolkit (see below) is used to create one or more policy scenarios, by:

- 1. Mapping the range of options available for a specific region to apply the Ex'tax principles;
- 2. Identifying a focus group of tax bases and
- 3. Identifying a focus group of policy options.

The final step in the Methodology entails an exploration of the potential impacts of the scenario(s) on the economy, environment, employment and business sectors.

Figure 9: The Ex'tax Methodology



The Ex'tax Toolkit

As mentioned in section 1.5, the Ex'tax Toolkit presents an overview of potential options to raise and use new revenues. Figure 7: The Ex'tax Toolkit – Raising Revenues (on page 27) provides policy options to apply the 'polluter pays' principle and raise revenues based on natural resource use and consumption (e.g. on air pollution, energy, fossil fuels, metals and minerals). Each category includes several sub-categories. For example, pricing schemes could be focused on different types of air pollution, such as carbon, particulate matter and/or nitrogen oxides. Figure 8: The Ex'tax Toolkit – Use of Revenues (see page 27) shows policy options to use tax revenues for the benefit of society (e.g. for investments, social assistance and environmental protection). Each category includes several sub-categories. Investments, for example could, amongst others, be directed towards infrastructure, R&D, renewable energy, housing or coastal protection.

3.2 Limiting factors

In the selection process of policy options for this analysis, several factors had to be taken into account, including national priorities, data limitations, modelling limitations and the stakeholder and political context. Also, a long-term perspective was taken, which means that the proposals are not aimed to be implemented immediately, but gradually over time.

National priorities

The national priorities of Bangladesh (Step 1 of the Methodology), were mapped based on the 7th Five Year Plan of Bangladesh (FYP) and other national plans (see Table 5 on page 38). The FYP, for example, specifically mentions investments in infrastructure, social protection and electricity coverage.

Using revenues to reduce the deficit or to lower personal income tax levels, for example, is not among the priorities.

Data limitations

In Step 2 of the Methodology, a large number of sources were consulted, including but not limited to databases of the Government of Bangladesh, the OECD, IEA, the IMF and the World Bank. A selection of the identified data is provided in Chapter 2. For modelling purposes, the available datasets provided a number of limitations. Some of the datasets provide recent and detailed information; others do not. For example, detailed data on fossil fuel use, energy consumption, carbon emissions and fertilizer usage are available. Data on other pollutants are limited and data on water usage are available yet outdated.

Modelling limitations

For the purpose of this study, a model (FRAMES: Bangladesh) was developed by Cambridge Econometrics, based on the theoretical framework of the E3ME macro-econometric model (see Chapter 4). Since this newly developed model is not yet as extensive as the original global E3ME model, some potentially desirable policy options could not be included in the scenarios. The 'business as usual' (or 'baseline') scenario was drawn from established sources such as HSBC and the World Bank. It is important to note that such official prognoses do not include the expected negative impacts of climate disruption on the GDP of Bangladesh (as mentioned in section 2.3). In future, the FRAMES model and its baseline could be adjusted to include such parameters.

Stakeholder and political context

Every country has its own dynamics in politics, culture and sensitivities. Through conversations with several stakeholders, some of the sensitivities could be taken into account. Due to the limited duration of this project, however, only a limited number of consultations could take place.

Medium- to long-term perspective

Putting a price on carbon emissions and abolishing fossil fuels subsidies are the two most documented and most obvious avenues to raise revenues in the SDG era. In countries around the world, such policies are generally met with resistance from corporations that are being subsidized to sustain their polluting activities. In the process of weighing such interests, in essence, the health and wellbeing of the people should be the top priority for governments. The scenarios presented in this study do not claim to be an immediate solution but should be viewed as a medium- to long-term pathway, to assist in the process of balancing the interests of different stakeholders.

Administrative capacities

Every government faces administrative challenges that inhibit the introduction of new policies. The authors are aware of such challenges in Bangladesh. Restraints in capacity have been taken into account as much as possible, but since the scenarios depict a medium- to long-term perspective, a certain development in capacity building can be assumed.

3.3 Two preliminary tax reform scenarios

Based on the above analyses and limitations, a few policy measures were chosen to be included in the scenarios under review. Unfortunately, due to data and modelling limitations, more detailed measures, such as pricing of water, fish, timber extraction and industrial water pollution are not yet included.³¹⁵ In future research, the number of options in the modelling could be expanded. The two scenarios under review include the following measures:

- Both scenarios raise revenues by introducing a carbon tax for industries and abolishing fossil fuel subsidies to industries.
- The *Infrastructure Scenario* invests the bulk of revenues in infrastructural development, while parts of the revenues are assumed to be invested in clean technology in the textiles sector.

• The *Social Scenario* assumes the bulk of revenues to boost social spending, while parts of the revenues are invested in clean technology in the textiles sector.

Both scenarios are revenue-neutral tax reforms, which means that the revenues from new tax rates introduced are used to offset something else.³¹⁶ Figure 10 visualizes the way revenues are raised and used in this study. Please note that of course, in practice, the government of Bangladesh would be able to spend on any combination of cleantech, infrastructure and social spending - or any other use of revenues.



Figure 10: The Infrastructure Scenario and the Social Scenario (in 2025, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

In the next sections, the measures under review are discussed in terms of their scope and rationale, their connection with the SDGs and national targets, as well as some international examples.

3.3.1 Raising revenues through a Carbon Tax (for industries)

The measure

Gradual introduction of a carbon tax of 30 per tonne of CO₂ emitted by industries and the power sector. The tax does not apply to transport and household/residential emissions. Such a measure would raise 3.0 billion in revenues in 2025 in each of the scenarios.

Rationale

It's important to note that first and foremost, high-income countries, who are responsible for historic carbon emissions, should lower their carbon emissions and compensate low-income countries for carrying the burden of the climate crisis.³¹⁷ However, in the SDG era, each and every economy should prepare for and benefit from carbon neutrality:

"Given the magnitude of the warming-induced growth penalties that poor countries have already suffered, expansion of low-carbon energy sources can be expected to provide a substantial secondary development benefit (by curbing future warming-induced growth penalties), in addition to the primary benefits of increased energy access." (Diffenbaugh 2019)³¹⁸

The Bangladeshi Ministry of Finance has stated that a carbon tax can be levied on the production, import, distribution, or use of fossil fuels. Such a tax increases the costs of carbon-emitting fossil fuels, compared to such non-carbon emitting energy sources as renewable energy and nuclear power.³¹⁹ Carbon taxes can be relatively simple to implement:

"[Carbon taxes] can be collected using the same administrative systems as existing taxes, or with only small modifications. In addition, when levied "upstream," such as on fuel wholesalers (importers and producers), carbon taxes need only be collected from relatively few firms. Wholesalers then pass the tax on to retailers, who pass it on to consumers. This "keep-it-simple" approach is strongly recommended by the experience of other countries." (Davies et al. 2018)³²⁰

According to these World Bank experts, introducing a carbon tax in Bangladesh could help rebalance the tax system away from workers and toward the payment of taxes on things that cause harm.³²¹

World Bank supports carbon tax in Bangladesh

The World Bank has estimated that Bangladesh could raise up to one percent of GDP in tax revenues, at \$30 per tonne of CO₂. The potential benefits of a carbon tax for Bangladesh are numerous, according to the World Bank, including:

- Carbon taxes are easier to collect than many other types of taxes.
- Carbon taxes can **boost Bangladesh's low revenue** while allowing it to lower other taxes to boost competitiveness.
- Carbon taxes will add to resources available to adapt to the impacts of climate change.
- Carbon taxes can help Bangladesh to **meet its commitments under the Paris Agreement** of reducing carbon emissions to 5 percent below the 'business as usual' scenario.
- Carbon taxes can help Bangladeshi firms to **prepare for changing global market** ("pre-empt changes in sentiment in increasingly environmentally aware export markets").
- Carbon taxes are less susceptible to economic cycles than most other taxes.³²²

Transport fuels versus other fuels

The main carbon emitters in Bangladesh are the power sector (42% of total), other industrial combustion (21%), buildings (14%), transport (12%) and other (12%).³²³ In this study, for a number of reasons, transportation and household mobility are as of yet excluded from carbon taxation. Firstly, alternatives in the form of public transport are not yet readily available for consumers in Bangladesh. Also, in the past, transport tax proposals have been met with resistance. In fiscal year 2014-2015, the National Board of Revenue contemplated the introduction of a carbon tax on private vehicles,³²⁴ which was not followed through. In 2017, a carbon tax proposal was abandoned due to backlash from (amongst others) the transport sector.³²⁵ Another reason to exclude transport and mobility is that recent studies modelled the impact of a carbon tax on petrol and diesel, which is not duplicated here (see Box 5).

Box 5: Carbon tax studies Bangladesh

World Bank: 'Options for a carbon tax in Bangladesh' (2018)

Carbon taxes could be a significant revenue source for the government of Bangladesh, raising up to one percent of GDP, at \$30 per tonne of CO₂ equivalent. Simulations show that most revenue would initially come from gas and petroleum products. Households of all wealth levels would pay a similar amount as a share of their income, making the first-round effects of a carbon tax distribution neutral. A carbon tax increases the cost of fuel but only by a small amount. At \$5/tCO₂, the motor gasoline price could be expected to increase from TK89/litre to TK90/litre. At \$30/tCO₂, it would increase to an estimated TK94-95/litre. Other fuels would see similarly small increases. However, some sectors would be more affected than others, particularly since some currently receive special treatment, benefiting from cheap fuel. The study provides an excellent overview of challenges, solutions and lessons from other countries' experiences. It concludes: "Given the potential benefits, and its ability to mitigate the risks, Bangladesh should consider implementing a carbon tax".³²⁶

Ahmed and Khondker: 'Towards a Carbon Tax in Bangladesh' (2018)

This study modelled the impacts of a carbon tax on petrol, octane and diesel, furnace oil and kerosene in Bangladesh, imposing a 10 percent tax on the current prices starting in FY2019 and gradually increasing to 25 percent by FY2041. The main conclusion is that such a tax lowers CO₂ substantially and raises considerable revenues. There are initial small negative output and employment effects that can be offset with fast transition to a clean energy environment. Additionally, the adverse output effects can be compensated with additional public investments in clean energy and infrastructure and social protection spending facilitated by higher revenues from the carbon tax. The report concludes that "a proper combination of fossil fuel pricing, carbon tax and investments can make carbon tax a win-win policy package".³²⁷

The political context for fossil fuel pricing reform is not easy, and all potential fuels taxes have been politically opposed.³²⁸ The World Bank, however, advices that:

"Implementing comprehensive energy sector reform that targets inefficiencies at different stages of power supply and distribution could boost supply while also limiting reliance on imported fossil fuel. This would not only minimize the direct cost of electricity generation, but also help avoid harmful emissions from fossil fuel-based generation."³²⁹

It's important to keep in mind that since 2009, the cost of wind turbines has dropped by nearly 33 percent, and that of solar photovoltaic modules by 80 percent, making both technologies increasingly competitive with fossil fuel power generation. Solar energy is now the cheapest generation technology in many parts of the world.³³⁰ Hence, alternatives for energy generation are getting more readily available.

The measure supports multiple priorities

The measure would tie in with the SDGs that mention domestic resource mobilization (1,10,12 and 17) as well as SDG 3 (healthy lives), 7 (energy), 8 (sustainable economic growth), 12 (sustainable consumption and production patterns) and 13 (climate change). Finally, the measure ties in with national priorities to 'Ensure energy mix for energy security' and 'Environmental, Climate Change and disaster risk reduction considerations are integrated into project design, budgetary allocations and implementation process'.³³¹ It also supports the INDC of Bangladesh (see page 38).

Pricing level

A wide variety of estimates exist of the so-called social cost of carbon (i.e. the damage that results from emitting a tonne of CO₂). The price of \$30 per tonne is roughly based on the (€30) minimum pricing level proposed by the OECD, as being "truly a minimum" estimate of climate damages from carbon emissions. In 2018 the OECD added a benchmark of €60 per tonne to their analyses to reflect the estimated cost of carbon in the future.³³² Other estimates do not take into account the external costs to society, but the price range needed to achieve certain reduction targets. The High-level Commission on Carbon Pricing concluded that a \$40–\$80 range in 2020, rising to \$50–\$100 by 2030, would be consistent with the core objective of the Paris Agreement of keeping the average temperature rise below 2°C.³³³

International examples

As of 2019, 57 national and sub-national jurisdictions around the world have implemented or scheduled carbon pricing schemes, 29 of which have opted for a carbon tax. 96 Parties - representing 55% of global GHG emissions - have stated in their Nationally Determined Contributions that they are planning or considering the use of carbon pricing to meet their commitments.³³⁴ In Asia, **India** introduced a carbon tax in 2010 as a tax on domestic coal, which was later broadened to include petrol and diesel.³³⁵ **Japan** introduced a carbon tax in 2012.³³⁶ **Singapore** is scheduled to introduce a carbon tax in 2019. The **China** national Emission Trading Scheme (ETS) was launched in 2017 and work is underway to prepare for its implementation. The **Kazakhstan** ETS was restarted in 2018 following a two-year suspension. The **Republic of Korea's** ETS started in 2015.³³⁷

Carbon taxes in Mexico and South Africa

Mexico has been moving from fossil fuel subsidies to carbon pricing. In 2015, fuel tax receipts accounted for 8% of total tax revenues in 2015 (the third largest tax in terms of revenues in Mexico). In addition, a modest carbon tax raised about 0.3% of total tax revenues.³³⁸ Revenues in Mexico are not earmarked, but flow into the general budget.³³⁹ **South Africa** has recently signed a carbon tax law. Set at \$8 per tonne, the tax will be largely offset by allowances to lower it to an effective rate of \$0.40-\$3 per tonne in the first three years. The tax is set to rise at two percent above inflation until 2022 and in line with inflation afterwards.³⁴⁰

Putting a price on the extraction of fossil fuels

Putting a price on the domestic extraction of a fuel (rather than its carbon content) can also be an option to serve development goals, and to a lesser extent, environmental goals. Since the 1960s, the government of the Netherlands has raised almost €417 billion in revenues for the national coffers from

the exploitation of oil and gas fields, which has enabled investments in social security and infrastructure.³⁴¹ At the same time, the extractive industry benefited from $\in 1$ trillion in turnover.³⁴² Considering the fact that natural gas is a finite resource, and causes pollution, it is important to plan for substitution in time. In the Netherlands, exploitation of the gas fields will come to an end soon, as supplies are running out. Also, the exploitation is causing earthquakes.³⁴³ Norway has invested its fossil fuel extraction revenues in a sovereign wealth fund, which could be looked at as an example for Bangladesh.

3.3.2 Raising revenues through removal of fossil fuel subsidies (for industries)

The measure

Gradual removal of fossil fuel subsidies for industries and the power sector, for natural gas (\$1.3 billion in 2025) and oil (\$7 million in 2025). The residential and transport sectors are not directly affected. Also, the subsidy for electricity remains unchanged. In total, this would raise \$1.3 billion in revenues in 2025 in each of the scenarios.³⁴⁴

Rationale

In 2017, Bangladesh imported \$4.8 billion worth of petroleum products.³⁴⁵ That same year, the country provided \$1.5 billion worth of fossil fuel subsidies;³⁴⁶ a quarter of the amount allocated towards social safety programs in the 2017 budget.³⁴⁷ When taking into account the external costs of burning fossil fuels (including global warming and air pollution), fossil fuel subsidies cost Bangladesh as much as \$8.8 billion in 2015.³⁴⁸ According to the Ministry of Finance of Bangladesh:

"Subsidized energy discourages initiatives that might develop better alternatives, for example green energy such as solar power. Renewable energies are relatively more expensive, however, and subsidies for fossil fuels put these alternative energy sources at a further disadvantage, discouraging their adoption."³⁴⁹

Natural gas main recipient of subsidies, supply risks loom

The bulk of fossil fuel subsidies support the use of natural gas. In Bangladesh, natural gas provides 58 percent of total primary energy supply and 80 percent of electricity generation.³⁵⁰ As of yet, natural gas is domestically sourced. BGR (2015) data for total natural gas resources and reserves show that Bangladesh could meet natural gas demand domestically until at least 2025 (assuming productive capacity can expand to exploit domestic resources). But dependence on imports is expected to rise given Bangladesh's current natural gas production levels, its natural gas reserves, and its fast-growing economy.³⁵¹ Supply risks are already emerging:

"Owing to substantial underpricing of natural gas compared to economic cost, natural gas consumption has been highly inefficient and constrained domestic investment in gas extraction. Rapid depletion of natural gas has now led to a severe gas rationing, thereby causing production loses in manufacturing and growing reliance on carbon emitting fuel oil for power production." (Ahmed 2018)³⁵²

Restructuring fuel subsidies would be rational

Even though natural gas subsidy reform is a highly sensitive issue, restructuring fuel subsidies would be rational. As the Ministry of Finance of Bangladesh stated in 2014:

"The adverse consequences of subsidy reduction on some segments of the population, (...) could be offset in the longer term by economy-wide benefits such as better fiscal sustainability, increased social spending targeting the poor, more efficient resource allocation, increased investment, and higher growth."³⁵³

Research by the Asian Development Bank found that:

"over time the new reality of higher-priced fossil fuels spurs users to change behavior and switch to cheaper forms of energy, which encourages investment in clean energy and drives down its cost. In time, the initial exaggerated effects of more expensive fossil fuels are softened as the economy returns to a path of cleaner energy and sustainable fiscal positions."³⁵⁴

And the World Bank supports gradual reform in combination with social assistance:

"hikes in gas and electricity prices can cause immediate economic distress, especially for the poor and vulnerable. Raising prices gradually while providing targeted social assistance can mitigate the impact. (...) price reform delivers large economic benefits in the long term".³⁵⁵

The measure supports multiple priorities

The measure ties in with the SDGs that mention domestic resource mobilization (1,10,12 and 17). It also supports SDG 3 (healthy lives), 7 (energy), 8 (sustainable economic growth), 12 (sustainable consumption and production patterns) and 13 (climate change). Finally, the measure ties in with the national priority: 'Ensure energy mix for energy security'.³⁵⁶

International examples

In 2009, G20 and Asia-Pacific Economic Cooperation leaders agreed to phase out inefficient fossil fuel subsidies in the medium term. Although implementation has been slow and patchy, several Asian countries have lowered subsidies, including India, Indonesia, Malaysia, the Philippines, Thailand and Vietnam.³⁵⁷ The IEA and IMF have documented fossil fuel reforms undertaken in almost 30 countries in 2013 and 2014, including:³⁵⁸

- **Egypt** raised fuel prices by 78% in 2014 and is doubling electricity prices over a period of five years.
- **Indonesia** raised petrol and diesel prices by an average of 33% in 2013 and by another 34% in 2014.
- India eliminated diesel subsidies in 2014 after incremental increases over the preceding two years.
- Iran raised petrol prices by 75% in 2015.
- Malaysia raised fuel prices by 10–20% in 2013 and again in 2014.
- Namibia removed subsidies steadily according to a three-year reform plan.
- A gradual approach was also adopted by **Kenya** (electricity), where the authorities were able to progressively gain support for broader reform by delivering improved services.

Earlier reforms in **Indonesia** and **Ghana** (in 2005) have been successful because of accompanying social policies:

"Indonesia's unconditional cash transfer program, which covered 35 percent of the population, was an important component of its successful strategy in overcoming social and political opposition to fuel subsidy reforms. (...) In Ghana, in 2005, the government commissioned an independent poverty and social impact analysis to assess the winners and losers from fuel subsidies and subsidy removal. This was an important foundation for persuasively communicating the necessity for reform and for designing policies to reduce the impact of higher fuel prices on the poor." (European Parliament 2017)³⁵⁹

Areas of concern and solutions to be studied more carefully

The revenue raising measures in sections 3.3.1 and 3.3.2 share a similar purpose and face similar challenges with regard to their impacts. Table 6 below provides some areas of concern and a few potential solutions. These issues are to be studied more carefully.

Table 6: Purpose, areas of concern and potential solution of the revenue-raising measures

Goals

- Creating a level playing field between energy sources (the 'polluter pays' principle).
- Aligning tax policy with climate and health goals.
- Creating fiscal space to invest in public services.
- Incentivizing a shift to clean technologies
- Limiting reliance on imported fossil fuels.

Area of concern

- Price increase for industries.
 Costs may be passed on to consumers.
- Production may shift to other countries (competitive disadvantage).

Potential solution

- Industries can shift to less polluting options.
- The measure is introduced gradually and accompanied by social policies.
- Additional measures to support investments in low-carbon technologies.

The next sections will take a closer look at the use of revenues in each of the scenarios under review.

3.3.3 Use of revenues: investing in clean technologies

The measure

In each of the scenarios under review, \$0.7 billion (20% of the newly raised revenues) in 2025 is allocated to support the Bangladeshi textiles sector in the transition to clean technologies and circular business models. With this measure, negative impacts of price increases in the scenarios are negated. The measures protect the long-term competitiveness of Bangladesh's most important export sector.

Rationale

Governments tend to provide tax exemptions to industries that are vulnerable to global competition, in order to reduce the risk of industries shifting their activities to less-regulated countries. In Bangladesh, for example, a reduced corporate income rate is applicable for the textiles industry (15% compared to 25% for listed and 35% for unlisted and private limited companies).³⁶⁰ In this study, the risk of outsourcing production is mitigated through innovation policy rather than exemptions. In the modelling, part of the revenues is allocated for cleantech funding targeted towards the textiles industry. Such investments may not neutralize expenses for the industry, but the investments do contribute to future-proofing production methods and mitigating the costs of climate disruption and water scarcity. It is estimated that:

"the adoption of resource-efficient and cleaner technologies in the Bangladeshi textile and leather industries could reduce long-term investments and operational expenditures needed to ensure a continued supply of water by up to US\$9 billion by 2030". (WRG 2015)³⁶¹

*"Gradual adoption of improved technology and cleaner production options could reduce wastewater volume by around 23% by 2021." (Sakamato 2019)*³⁶²

"36 per cent of the industry's climate impacts occur during dyeing and finishing (...). This is because the electricity and heat being used in this stage mostly comes from hard coal and natural gas. If we could substitute these fossil fuel-based energy sources with renewable energy sources we would make a lot of progress." (Schragger n.d.)³⁶³

Chinese mills taking effective measures

Investments in clean technologies can be highly effective. In China, for example, five pilot mills tested a series of "easy to implement, low cost, quick return" measures that are profitable and reduce environmental impact. Enabled by smart metering, the average reduction in water use was 9%, with the best mills achieving a 20% reduction. Average energy consumption fell by 6% and in some cases energy reduction was as high as 10%, with total savings equivalent to 61,000 tonnes of coal. Economic benefits totalled \$14.7 million.³⁶⁴

Types of investments

Cleantech funds could potentially be invested in R&D and technologies to deploy sustainable processes such as waterless dyeing and ultrasound laser cutting.³⁶⁵ Investments in energy saving solutions and renewable energy could also be included. In addition, funds could enable the development and use of sustainable dyes³⁶⁶ and new types of materials.³⁶⁷ Note that if such materials are locally sourced, they could contribute to Bangladesh becoming less dependent on cotton imports. The textiles sector is currently 90 percent dependent on imports.³⁶⁸

The measure supports multiple priorities

Depending on the specific use of the funding, the measure ties in with SDGs 6 (sustainable management of water), 7 (sustainable and modern energy), 8 (sustainable economic growth), 11 (safe and sustainable cities), 12 (sustainable consumption and production patterns), 13 (combat climate change), 15 (sustainable use of ecosystems). The measure also ties in with the national priority for 'Spending on Research and Development to constitute 1% of GDP'.³⁶⁹ Date on current levels of R&D spending have not yet been identified.

3.3.4 Use of revenues: investing in infrastructure

The measure

In the *Infrastructure Scenario*, \$3.6 billion (80% of the newly raised revenues) in 2025 is used to invest in infrastructure. The exact infrastructure projects are not defined; the supply side effects of the policy are not modelled.

Rationale

The *Global Competitiveness Report 2018* found that inadequate infrastructure is one of the most problematic factors for doing business in Bangladesh.³⁷⁰ Infrastructure is one of the core themes set under the 7th Five Year Plan of Bangladesh, with targets such as the construction of bridges, expressways and highways.³⁷¹ According to an IMF review:

"A planned investment of US\$409.0 billion is expected under the Seventh Plan, but the chronic under-implementation of the Annual Development Plan, despite its higher priority and continuing growth with each budget, needs to be addressed. (...) the government needs to raise tax revenues to create the fiscal space to upgrade infrastructure, such as electricity, roads, rails, and ports. This will in turn improve the business environment, attract FDI [Foreign Direct Investment], and diversify exports.³⁷²

The World Bank also identifies infrastructure bottlenecks as "a risk for future growth prospects" in Bangladesh.³⁷³

Foreign investments are not enough

Large infrastructural works are being financed through foreign investments. Under the *Belt and Road Initiative*, the Chinese authorities have planned to finance several infrastructure projects of Bangladesh in sectors like telecom, agriculture, power and energy. A line of credit with India totalling \$4.5 billion was signed in 2017.³⁷⁴ Despite these foreign investments, this measure is included in the scenarios based on the large ambitions of Bangladesh with regard to infrastructure, public utilities, R&D and sustainability.³⁷⁵ The modelling in this study (see Chapter 4) allows for a comparison of the impacts of investments in infrastructure versus increasing social spending, which will be discussed in section 3.3.5.

Potential role of local solutions

Bangladesh has a strongly developed culture of local, bottom-up initiatives. In practice, therefore, taking a bottom-up, participatory approach rather than taking a top-down planning approach, should be considered.³⁷⁶ Investments could also, for example, be targeted towards electric transit (including electric buses³⁷⁷ and electric rickshaws),³⁷⁸ which could contribute to cleaner air and less traffic congestion. Also, off-grid energy solutions (such as Solshare)³⁷⁹ and low-carbon agricultural infrastructure (such as digital demand and supply platforms³⁸⁰ and rooftop gardens³⁸¹) could be among potential investment opportunities.

The measure supports multiple priorities

Depending on the use of revenues, the measure potentially supports SDGs 6 (sustainable management of water and sanitation), 7 (energy), 9 (infrastructure and innovation), 10 (inequality), 11 (sustainable and safe cities), 12 (sustainable consumption and production patterns) and 13 (climate change). An IMF review identified access to clean water, electricity and reliable and affordable transportation also as key enablers to (female) labor force participation and inclusive growth (SDGs 5 (gender equality) and 8 (sustainable and inclusive growth and decent work)).³⁸²

3.3.5 Use of revenues: increasing social spending

The measure

In the *Social Spending Scenario*, \$3.6 billion (80% of the newly raised revenues) in 2025 is used to boost household income. The measure is targeted towards the lowest two income quintiles. The exact nature of the payment is not defined; in the modelling framework it is treated as a basic transfer from government to households. Implementation could take the form the government deems most suitable

(including child benefits, cash transfers, social security, in-kind transfers³⁸³ or perhaps employment guarantee schemes which provide a certain number of days of work to low-income groups).³⁸⁴

Rationale: shared prosperity for the bottom 40 percent

In the 2017-18 revised budget of Bangladesh, \$6.1 billion (13.1% of the budget, or 2.2% of GDP) was allocated for social safety net programs.³⁸⁵ The 7th Five Year Plan of Bangladesh includes the goals to 'Reduce or maintain the current income inequality' and 'Spending on Social Protection as a share of GDP to be increased to 2.3% of GDP' by 2021.³⁸⁶ In the modelling, the additional social spending is targeted towards the bottom 40 percent of the population. The World Bank promotes 'shared prosperity' defined as 'the income growth of the bottom 40 percent of the population'.³⁸⁷ The OECD *Inclusive Growth Initiative* also puts the emphasis on policies that can improve the perspectives of the bottom 40% of the income distribution.³⁸⁸ Such approach is in line with SDG 10 (Reduce inequality within and among countries), target 10.1, which reads:

"By 2030, progressively achieve and sustain income growth of the bottom 40 per cent of the population at a rate higher than the national average."

The measure supports multiple priorities

The measure potentially ties in with SDGs 1 (end poverty), 2 (food security), 3 (health), 7 (access to energy), 8 (inclusive economic growth), 10 (reduce inequality) and 11 (inclusive, safe human settlements). As mentioned before, the measure ties in with the national priorities to reduce or maintain the current income inequality, to increase spending on social protection, and to reduce the head-count poverty ratio and extreme poverty. Finally, it ties in with the National Social Security Strategy (NSSS) Priority Challenges to expand coverage of social protection for the extreme/hard-core poor and most vulnerable people of the society, ensuring that the most vulnerable women are provided with income security and expanding coverage to the residents of urban areas and to the socially excluded people.³⁸⁹

International example

In Egypt, in 2013, the phase-out of fossil fuel subsidies was inaugurated alongside two cash transfer programs to help offset the impact of fuel price increases on poor and vulnerable households. By 2017, these programs covered about 1.5 million families (6 million Egyptians) out of the 1.7 million families targeted.³⁹⁰ As mentioned in section 3.3.2, reforms in Indonesia and Ghana (in 2005) have been successful because of accompanying social policies.

Cambridge Econometrics modelled the impacts of the scenarios on tax revenues and macroeconomic and environmental indicators in Bangladesh. The next chapter will describe the results.

4. Modelling results

Cambridge Econometrics has modelled some of the impacts of two preliminary scenarios, which include putting a price on carbon emissions and abolishing fossil fuel subsidies, while using the revenues to invest in clean technologies, infrastructure and social spending. The modelling suggests that by 2025, such tax reforms could lead to higher GDP and employment levels, while reducing carbon emissions and energy imports. The transition can be highly progressive when revenues are mainly used to increase social spending.

4.1 Introducing the macro-econometric model

The modelling framework

Cambridge Econometrics is a UK based company founded in 1978 as a spin-off from the University of Cambridge, to take forward the work of Prof Sir Richard Stone, Nobel Laureate in Economics.³⁹¹ Cambridge Econometrics developed the 'E3ME model', a computer-based macro-econometric model of global economies, used for analysing the detailed linkages between the economy, materials, environment and energy.³⁹² The model was originally developed through the European Commission's research framework programmes³⁹³ and is now widely used in collaboration with a range of institutions for policy assessment, forecasting and research purposes.³⁹⁴ E3ME covers details of 61 countries and regions, including China, India, Korea, Taiwan and Indonesia. The other countries in Asia are grouped in the 'rest of ASEAN' category. Based on the parameters and expertise developed in E3ME, Cambridge Econometrics has created the Framework for Modelling Economies and Sustainability (FRAMES) model to estimate potential tax revenues and macro-economic impacts for this study on Bangladesh.

Assumptions and limitations

The E3ME and FRAMES: Bangladesh models are based on a post-Keynesian economic framework and its assumptions are consistent with this branch of economics.³⁹⁵ The approach is generally an empirical one, with behavioural parameters determined by relationships in the data. It is assumed that these relationships are maintained in the projection period, i.e. that behavioural responses remain consistent with those in the past. There are several important assumptions specific to this analysis:

Baseline projections. A 'baseline' scenario (assuming no policy intervention) was developed for the modelling, based on projections by the World Bank³⁹⁶ and HSBC.³⁹⁷ All results represent the difference on top of any changes that occur in the baseline.

National policy. The measures are introduced in the model on a national level in Bangladesh.

Phasing in. Policy measures are assumed to be introduced gradually from 2020, to reach the full measures by 2024 and remain the same beyond 2024. A linear path of introduction is applied over the five-year period 2020-2024, so the initial tax rates in 2020 are in general quite low.

Budget-neutrality. Each year, all revenues are used ('recycled') in accordance with the scenarios (this means there is no impact on the public deficit).

Price effects. The model captures price effects and does not include any awareness or signalling effects from the green taxes. This means that the responses to changes in tax rates should be attributed to the financial effects, rather than any publicity or virtue-signalling that accompanies the reforms.

Prices. All dollar-values in the results are in 2017 prices, unless specified differently.

Behavioural parameters. The behavioural parameters in the model are taken from E3ME. Bangladesh is not represented individually in E3ME; India was therefore chosen as the most appropriate region to be the proxy.

Impact beyond 2025. The tax reform is introduced gradually to 2024. The model assumes that by 2024, full behavioural responses are realised. Effects from 2025 onwards would be similar, unless further reform was introduced.

Appendix A provides a more detailed description of the model and the modelling results. Below is a summary of the key modelling results (section 4.2) and selected impacts by sector (section 4.3).

4.2 Key results

Addressing development and environment simultaneously

In the year 2025, both scenarios mentioned in section 3.3 are expected to raise \$4.3 billion in revenues. In the modelling, every year, the revenues are fully recycled. In the *Infrastructure Scenario*, all revenues are recycled through investments in clean technology and infrastructure. In the *Social Spending Scenario* (or '*Social Scenario*'), all revenues are recycled through investments in clean technology and social spending. The key message from the results is that it is possible to design policy measures that reduce harmful emissions and final energy consumption, while at the same time stimulating the economy of Bangladesh, creating jobs and (in the *Social Scenario*) increasing income for the bottom 40 percent. These results demonstrate that Bangladesh doesn't need to choose between development and environment.



Figure 11: Overall result: decoupling (2020-2025, % difference from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Decoupling of GDP and carbon emissions

Figure 11 provides some key results over the 2020-2025 period, demonstrating decoupling effects in the scenarios, as GDP is higher, and emissions are lower. An increase in employment is observed in each scenario. The scenarios add \$6.9 billion (in the *Infrastructure Scenario*) and \$7.8 billion (in the *Social Scenario*) to GDP over the 2020-2025 period (Table 7 provides cumulative impacts). Additional findings include, over a six-year period:

- **Resource mobilisation.** Phasing out fossil fuel subsidies could potentially raise \$4.7 billion in domestic resources, while a carbon tax could add another \$10.6 billion in domestic resources.
- Job creation. Both scenarios show significant increases in employment (540,000 and 670,000 years of employment respectively).
- **Carbon emission reductions.** Both scenarios demonstrate a significant reduction in carbon emissions (saving 19.9 and 18.5 megatonnes of carbon respectively).
- Savings on energy imports. In both scenarios, Bangladesh saves significant amounts on energy imports (\$429 million and \$405 million respectively).

• **Public investments.** In the *Infrastructure Scenario* \$12.8 billion is invested in infrastructure. In the *Social Scenario*, \$12.9 billion is invested in social protection. Cleantech investments in both scenarios are \$2.6 and \$2.5 billion respectively.

Distribution of benefits and cost

As with any reform, the benefits and costs will not be spread evenly (see section 4.3). In both scenarios, the textiles sector shows a slight negative result in terms of gross output (0.24% and 0.15% respectively) by 2025, but overall, the economy would be stronger and more competitive in terms of carbon intensity and energy import dependency. Also, it's important to note that the competitiveness impacts of the cleantech investments (totalling more than \$2.5 billion) are not yet captured in the model. The modelling results suggest that a progressive impact with higher benefits (in relative terms) for lower income households is possible, particularly in the *Social Scenario*.

Consumer expenditure key difference between scenarios

A key difference between the scenarios is in consumer expenditure. Whereas the *Infrastructure Scenario* does not compensate low-income groups for increased living expenses, the *Social Scenario* specifically boosts consumer spending (see Figure 12).

- In the *Infrastructure Scenario*, total consumer expenditure is 0.6% lower than it is in the baseline, in real terms. The infrastructure investment creates jobs and there are positive multiplier effects throughout the economy. Price rises outweigh this effect, however, meaning household income and consumption are lower in real terms.
- In the *Social Scenario*, total consumer expenditure is 0.4% higher than the baseline, in real terms. The revenue recycled in social spending increases consumer expenditure and creates a stimulus effect which creates jobs. These effects are substantial enough to outweigh the effect of price rises.



Figure 12: Key modelling results (2025, % difference from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Table 7 and Table 8 provide details for a number of line-items.

	Infrastructure Scenario	Social Scenario
GDP	\$6.9 billion	\$7.8 billion
Employment	543,000	670,000
CO ₂ emissions	- 19.9 million tCO ₂	- 18.5 million tCO ₂
Final energy consumption	- 5,346,000 toe	- 4,720,000 toe
Energy import savings	- \$429 million	- \$405 million

Table 7: Cumulative results (2020-2025, difference from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Table 8: Economic, social & environmental impacts (2025, difference from baseline,Bangladesh)

Difference from baseline	INFRASTRUCTURE		SOCIAL	
	(%)	(value)	(%)	(value)
Economic indicators				
GDP*	0.4%	\$1.9 bln	0.5%	\$2.2 bln
Output*	0.5%	\$5.6 bln	0.4%	\$4.2 bln
Consumer expenditure	-0.6%	-\$2.2 bln	0.4%	\$1.6 bln
Investment	3.3%	\$4.8 bln	0.8%	\$1.2 bln
Exports	-0.3%	-\$0.2 bln	-0.3%	-\$0.2 bln
Imports	0.3%	\$0.4 bln	0.3%	\$0.3 bln
Social indicators				
Employment	0.2%	139,000 persons	0.2%	172,000 persons
Change in household income poorest 1 st quintile	-0.4%	-\$4.7 bln	12.3%	\$135.0 bln
Change in household income 2 nd quintile	-0.4%	-\$11.3 bln	4.9%	\$129.1 bln
Environmental indicators				
Final energy consumption	-3.0%	-1,499 ktoe	-2.6%	-1,324 ktoe
CO ₂	-4.0%	-5.6Mt	-3.7%	-5.2Mt

* Non-discounted.

Source: Model projections, Cambridge Econometrics 2019.

Real incomes and inequality

Real incomes in the scenarios are affected by changes in wage income, income from government social spending, and prices. Rich and poor households spend their incomes in different ways and have different effective taxation rates; the impacts may vary across household groups. The important differences in the context of the scenarios are expenditure on energy and energy-intensive goods and services. In Bangladesh, higher income households spend a larger share of income in these areas, and therefore face higher price increases.

- In the *Infrastructure Scenario*, there are no explicit redistributive policies. The scenario is progressive, however, because the lower income quintile households experience smaller reductions in real incomes than those in the higher quintiles. Overall, the differences are small; the lowest quintile has a 0.4% real income reduction, compared to 0.6% for the highest quintile.
- When we account for the measure to boost incomes in the lowest two quintiles, the effect is substantially more pronounced. In the *Social Scenario*, real income increases by 12.3% for the lowest income quintile, and 4.9% for the second lowest. Real income in the highest three quintiles reduces by 0.6% (see Figure 13).



Figure 13: Social Scenario - real incomes per quintile (2025, % diff. from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

The next section will provide a brief explanation on the impacts by sector, in terms of output, employment and carbon emissions.

4.3 Impacts by sector

Output

Figure 14 provides the impacts of the scenarios on sectoral output. While GDP represents the value added in the economy, output is a gross measure that includes input material and energy (but not labour) costs.

- In the *Infrastructure Scenario*, the largest increases in output, both in relative and absolute terms, are in the construction, manufacturing (excluding textiles) and mining and quarrying sectors. These sectors benefit from the investments in infrastructure. Some of the manufacturing companies in the supply chain also benefit. Output falls marginally in sectors that are supplying consumer final demands, such as retail. In the textiles industry output falls because energy prices increase, leading to some loss of price competitiveness and lower exports.
- In the Social Scenario, most sectors demonstrate an increase in output as they benefit from higher local consumer spending. Manufacturing and construction also have relatively high increases, because of the investment in cleantech for textiles. Textiles lags behind due to an energy prices increase, leading to some loss of price competitiveness and lower exports. The overall implications and outlook for the textiles sector will be discussed in more detail below.



Figure 14: Output by sector (2025, % diff. from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Employment

The percentage changes in employment by sector (see Figure 15) largely follow the pattern of changes in output, whilst allowing for different returns to scale. Therefore, the largest changes are found in labour-intensive sectors, and those where output changes by the most.

- In the *Infrastructure Scenario*, the largest increases in employment, both in relative and absolute terms, are in construction, mining and quarrying and real estate (see Figure 15). These sectors benefit from the investments in infrastructure. Some of the manufacturing companies in its supply chain also benefit. In absolute terms, 63% of new jobs in this scenario are created in the construction sector. Agriculture creates the second greatest number of jobs, because of the size of the sector and its labour-intensity. Employment falls in the energy and utilities sectors because of reduced demand for energy products, and in the textiles industry because of some loss of price competitiveness and lower exports.
- In the Social Scenario, all sectors (except textiles) demonstrate an increase in employment as they benefit from higher local consumer spending. The textiles industry lags behind due to some loss of price competitiveness and lower exports. In absolute numbers, wholesale and retail trade create the most jobs (53% of the total), followed by agriculture and construction. Jobs are created in construction because of the investment in cleantech for textiles, and the induced positive effects in economy-wide investment.





Source: Model projections, Cambridge Econometrics 2019.

Carbon emissions

In 2025, the policy measures in the scenarios reduce CO_2 emissions by 4.0% and 3.7% compared to the baseline, in the *Infrastructure Scenario* and *Social Scenario* respectively. The impacts on emissions from the different sectors vary according to their fuel mix, existing fuel prices (including tax rates) and whether they are included in the new measures. The largest proportional reductions in CO_2 in both scenarios is across the industrial sectors (see Figure 16). This is because, in Bangladesh, industry is responsible for approximately 85% of coal use, the fuel which is most affected by these reforms.

The power generation sector has minimal changes in emissions, because of limitations in the modelling framework. The model assumes that the power generation mix is constant over time and across scenarios. These tax reforms provide a strong incentive for different power generation technologies. However, the time horizon of analysis is 2025 and given the lifetime of power generation technologies, there would be limited retirement of capacity to be replaced over these years.



Figure 16: CO₂ emissions per sector (2025, % difference from baseline, Bangladesh)

Source: Model projections, Cambridge Econometrics 2019.

Focus on the textiles sector

The textiles sector is unique in the Bangladeshi economy and in the dynamics it faces from tax reform. Textiles is the only sector that is reliant on exports, rather than domestic demand. It is therefore the most vulnerable to the loss of international competitiveness arising from increased costs of production resulting from tax reform. Exports reduce by 0.4 percent in both scenarios by 2025. A portion of the revenue accruing from the tax reforms is allocated to investing in the sustainability of production in the textiles industry. This investment would ameliorate the negative price competitiveness effects on the industry, with positive quality effects. These effects have not been quantified in the model.

Considering the importance of the textiles sector to the economy in Bangladesh, the next chapter will look closer at the perspectives and dynamics in the textiles sector.

5. Exploring the implications of tax reform for the textiles sector

1.10

The textiles sector in Bangladesh faces significant challenges to remain competitive in the face of global megatrends. This chapter reviews the dynamics of tax reform and its impact on the textiles sector in Bangladesh.

5.1 Setting the scene: global megatrends

The disposable clothes trend

Fast fashion constantly offers new styles to buy, as the average number of collections released by European apparel companies per year has gone from two in 2000 to five in 2011, with, for instance, Zara offering as many as 24 new clothing collections each year. This has led to consumers to see cheap clothing items increasingly as perishable goods that are nearly disposable, and that are thrown away after wearing them only seven or eight times.³⁹⁸ Between 1996 and 2012, the amount of clothes bought per person in the European Union increased by 40 percent,³⁹⁹ but about a third of those dresses, shirts and pants end up sitting in closets largely unused.⁴⁰⁰ The fashion industry produces 100 billion garments per year; nearly 14 items of clothing for every person on earth.⁴⁰¹

Massive environmental impact

Apparel and footwear account for 8% of global greenhouse gas emissions.⁴⁰² Globally, less than 1% of garments are recycled into new clothing. Every second, a garbage truck of textiles is landfilled or incinerated.⁴⁰³ In 2015, the global textiles and clothing industry was responsible for the consumption of 79 billion cubic metres of water, 1,715 million tons of CO₂ emissions and 92 million tons of waste. By 2030, under a business-as-usual scenario, these numbers would increase by at least 50%.⁴⁰⁴

Global fashion market committed to circularity

Leading industry groups, together with global fashion organizations, are pushing for the industry to change its ways. The 2020 *Circular Fashion System Commitment* was signed by 94 companies, representing 12.5 percent of the global fashion market. This commitment focuses on four immediate action points: 1) Implementing design strategies for cyclability; 2) Increasing the volume of used garments and footwear collected; 3) Increasing the volume of used garments and footwear resold; and 4) Increasing the share of garments and footwear made from recycled post-consumer textile fibres.⁴⁰⁵

Making the business case

The goals of the apparel industry to become sustainable and circular are ambitious especially since financial incentives embedded in current tax systems favour the business-as-usual linear economy. As indicated in Chapter 1, pollution and resource use are practically tax-free and even subsidized. In a high-volume, low-cost industry, such as the apparel industry, under these circumstances, it is particularly difficult to make a business case around saving water, using renewable energy and abolishing pollution. The development of sustainable supply chains requires more time, effort and R&D; all of which require more labour input than simply sustaining the linear model. In a system that puts a high or rising tax burden on labour, making such investments is even less financially attractive. Therefore, there is no level playing field for cleaner and innovative solutions.

The risk of corporations shifting their production to regions with ever-lower wages and lower environmental standards is eminent. Countries like Bangladesh therefore are required to perform a balancing act, combining a need to reduce environmental and health damage, while increasing the number of decent jobs *and* keeping costs as low as possible, in order not to deter producers. This struggle is likely to intensify if global trade is to adhere to the goal of keeping global warming within safe limits, as will be discussed next.

5.2 Staying competitive in a 1.5-degree warmer world?

A global competitive market

Bangladesh is battling to keep its position as the world's second-largest exporter of clothing after China, as it faces intensifying competition from Cambodia, Vietnam, Myanmar and now African countries like Ethiopia, as global brands search for cheap labour. H&M, for instance, has started sourcing from an Ethiopian clothing factory it set up with garment maker DBL.⁴⁰⁶ Ultimately, to break this cycle of shifting to countries with ever-lower production costs, consumers (in high-income countries) will probably need to pay more for products.⁴⁰⁷ Paying living wages to garment workers would add just one percent to the retail price of a piece of clothing.⁴⁰⁸ However, if all negative social and environmental impacts were taken into account, a pair of jeans from Bangladesh, for example, would cost €33 more.⁴⁰⁹

Textiles industry at crossroads

The global market is changing fast, with water supply risks and carbon pricing on the rise. In Bangladesh:

"If "business as usual" water demand continues for the textile sector, in particular, this will result in an additional water demand of over 6,750 megalitres per day by 2030. This is equivalent to the annual water needs of a population of approximately 60 million people in Bangladesh.

(...) It is estimated that the level of investment in new assets (water abstraction treatment and distribution plant as well as effluent treatment plant) to support growth in the textile sector will be in the order of \$19 to \$30 billion through to 2030 under a "business as usual" water demand scenario." (WRG 2015)⁴¹⁰

It is clear that in future, water usage will not be free of charge, as the costs of water scarcity to society can no longer be ignored. It's also clear that carbon emissions can't be free of charge - or even subsidized - in a 1.5°C warmer world. Nor can brands ignore the impacts of toxic chemicals and the ever louder call for more social practices, such as paying living wages and creating a safe working environment. Global demand for clothing is also changing, with resale growing 21 times faster than the retail apparel market over the past three years.⁴¹¹ The textiles industry is therefore at a crossroads; continuing the linear model (while imposing external costs to society and future generations), or shifting to circular models, and adapting to changing circumstances. According to the Ellen MacArthur Foundation, a leading think tank in sustainability, a 'New Textiles Economy' is 'distributive by design' and 'reflects the true cost (environmental and societal) of materials and production processes in the price of products'.⁴¹²

Textiles and tax reform

This study demonstrates how aligning tax policy with the SDGs can be relatively neutral to the sector while sustaining an overall better functioning economy, higher investments in future-proofing technologies and higher consumer spending in Bangladesh. Will this be reason enough for the fashion producers to support tax reform? Maybe not; as nobody likes to pay more for something that was previously for free. But looking at global trends, tax reform, including fossil fuel subsidy reform, would be a way to reduce risks and future-proof the sector. According to the World Bank, carbon taxes in Bangladesh can benefit firms, including exporters:

- There is well-established empirical evidence that environmental regulations can stimulate innovation.⁴¹³ Note that the scenarios under review directly contribute to innovation in the textiles sector.
- A carbon tax would allow firms producing in Bangladesh to market their products as more sustainable than those produced in other countries.⁴¹⁴
- A carbon tax would help Bangladeshi exporters get 'ahead of the curve' as consumers (and laws) in developed markets become more environmentally conscious in their sourcing of imports.⁴¹⁵

There is a risk of carbon leakage if firms leave Bangladesh for countries without carbon pricing mechanisms. The evidence to date suggests that such negative impacts are non-existent to minimal.

However, they may impact some sectors, and Bangladesh's garment sector may be one example that deserves consideration. The World Bank notes that the risks are reducing as more countries impose a carbon tax.⁴¹⁶ Also, research shows that investments in resource efficiency, secure work environments and sustainable materials would actually boost profitability by up to 1-2 percentage points in EBIT margin by 2030.⁴¹⁷

Opportunities for circular business models in Bangladesh?

Impacts on global supply chains of circular practices need to be researched, as many issues remain unclear. For example:

- How could Bangladesh shift from activities with relatively low value-added towards higher value-added activities, such as design, reuse and customization and towards production methods with a smaller footprint and more quality employment?
- How could smart policies support the transition to a circular supply chain? In Bangladesh, a reduced tax rate of ten percent has been introduced for garment factories that have an internationally recognized green building certification.⁴¹⁸ Is this measure effective? Should it be complemented with other policies?
- How could a country like Bangladesh be part of the 'closed loops' of fashion brands? Could products be returned to Bangladesh in a responsible way to be recycled or modified?
- What are the opportunities in local or regional circular business models? How to connect local and regional loops?
- If shipping were a country, it would be the sixth biggest in terms of emissions share.⁴¹⁹ How will global trade be sustained in a 1.5-degree warmer world? Will high-volume carbon-neutral shipping be possible?

Questions like these should be the subject of continued research with and in Bangladesh.

Adaptation in the DNA of business

The textiles sector is an important driver for growth in Bangladesh, but it still fails to pass on equally to the population. 'Business as usual' will be difficult to sustain considering global megatrends such as climate disruption and displacements, water supply risks, the global focus on the SDGs, social needs, and circular business models. Fortunately, adapting to changing circumstances is in the DNA of business and the global sense of responsibility in business is growing. As the CEO of Royal DSM (an \in 8.6 billion health, nutrition and materials company) stated:

"[As business leaders] We cannot be successful, nor call ourselves successful, in a society that fails. The increased impact and power of business need to lead to an increased responsibility to contribute to the real, higher goals of our economy, to serve society."⁴²⁰

Considering all of the above, business can and should play a prominent role in assisting governments in aligning tax systems with the SDGs.

6. Conclusion and recommendations

Charles and the second

State Construct Sold and

We have entered the SDG era; an era of unprecedented global social and environmental challenges. The most daunting task will be to adapt the metabolism of our economies to match the carrying capacity of the earth and stay well below 2°C global warming. We face equally important social challenges in our societies, including enabling a growing population to develop to their full potential and find decent work.

Aligning economic growth with the goals of the SDGs will be key. The linear (take-make-waste) economy is no longer maintainable. A shift is needed towards inclusive circular economies, which are regenerative, carbon neutral and distributive. Tax systems play a fundamental role in this transition and there is now widespread support for the principles of tax reform; putting a price on pollution and resources and using the revenues for social impact.

Updating the tax system is not a simple task, especially in countries with low- and middle-incomes, which face the multiple challenge of developing the economy and social systems, while at the same time preserving natural resources. This study confirms that countries may not need to choose between those goals. Smart policies could help countries to 'leapfrog' into the SDG era.

It may be clear that many details and complications still need to be researched. The question is whether to resolve these issues or allow them to immobilize our current systems, that were built for a different era.

We therefore call upon all stakeholders, businesses, governments and NGOs to do what is in their power to turn tax into a force for good. And to help build modern tax systems that enable prosperity based less on natural resource use and more on the abundance of human capacities and talents. For this is growth that can be sustained by generations to come.

Below are four recommendations for next steps in Bangladesh and other countries to be taken by business leaders and governments.

RECOMMENDATIONS FOR BUSINESSES:

1) Evaluate the risks and opportunities related to global environmental and socioeconomic megatrends.

Companies should focus on getting the insights necessary to evaluate the company's external costs and benefits and disclose consistent information on the risks and opportunities.

2) Gain and share insights on the impact of tax reform from a business perspective.

Tax reform changes the dynamics in business. There is a substantial lack of knowledge about the risks and opportunities for companies; how does a shift in taxation affect strategic choices concerning products, services and new technologies? Businesses should help governments gain more insight in the transformational power of businesses. This will enable a well-informed discussion between policy makers and businesses.

3) Lead by example.

Businesses can be a gamechanger for the SDGs. Business leaders could lead by example by:

- Joining the global community in support of climate action and carbon pricing.
- Applying internal carbon pricing and extending such mechanisms to other externalities.
- Shifting towards more inclusive and sustainable business models, thereby contributing to the goals of the SDGs.

4) Engage proactively with government to implement policies to support the SDGs.

Businesses should engage proactively with governments in the process of aligning public and private interests in support of the SDGs. Responsible business associations could play a significant role in this process.

RECOMMENDATIONS FOR GOVERNMENTS:

1) Extend data systems to have accurate information in place for effective policymaking.

A robust and future-proof tax system will require adequate risk assessments and increasing level of responsiveness to urgent matters. Detailed and up-to-date economic, environmental and social data are needed to assess the potential for policy measures. This starts with detailed measurement of the metabolism of our economies.

2) Create interdisciplinary research programs.

As tax policy is intertwined with economic, environmental and social issues, a systemic and longterm approach is needed. Fostering cooperation between government departments (Tax, Finance, Environment, Economic Affairs and Employment), as well as businesses and other stakeholders (such as NGOs and research agencies), will be key for the development of effective and fair policies.

3) Start the implementation process.

Practical steps towards tax reform include:

Step 1: Assessing the pathway necessary to achieve national, regional and global ambitions.

Step 2: Putting a price on pollution and natural resource use, starting with abolishing fossil fuel subsidies and pricing carbon emissions.

Step 3: Using the revenues to lower the tax burden on labour, improve social protection (in particular addressing the needs of lower-income households) and increase public investments.

Step 4: Monitoring and adjusting policy measures over time.

During the process, engaging with businesses and the public ahead of any change and communicating the impacts in a transparent manner will be key.

4) Seek international cooperation.

Work together with other countries and regions to achieve a coherent international tax strategy serving the SDGs. This lays the ground for global coordination.

The world has moved on; tax systems need to do the same.

Appendix A: FRAMES: Bangladesh

By Cambridge Econometrics

Description of the model

FRAMES: Bangladesh is an economic model of the Bangladeshi economy, developed specifically for this study, by Cambridge Econometrics. The model was built to examine the socioeconomic and environmental effects of energy-environment-economy (E3) policies in Bangladesh. The key features of FRAMES: Bangladesh are:

- An economic accounting framework based on the system of national accounts.
- Integrated treatment of the economy, energy, and the environment, with linkages between each component.
- Detailed sectoral disaggregation, and a national level input-output table, reflecting the specific structure of the economy.
- Calculations of income effects by quintile, providing distributional results.

Figure 1 details the basic economic structure in the model. At the core of the economic modelling in FRAMES: Bangladesh is a national input-output table for Bangladesh¹. Relationships for investment, prices, employment, and trade are modelled using elasticities which have been econometrically estimated. The FRAMES: Bangladesh model is based on a post-Keynesian economic framework, and its assumptions are consistent with this branch of economics. The application of this theory to economic modelling are informed by E3ME¹, a computer-based model of global economies, used for analysing the detailed linkages between the economy, materials, environment and energy.

Figure 1: Economic Structure in FRAMES: Bangladesh



Modelling ETR

The analysis in this study focuses on a scenario of Environmental Tax Reform (ETR) in Bangladesh; specifically, adjustments to fossil fuel consumption subsidies, and introduction of a carbon tax. The first order effect of both policies is to increase the price of fuel use; the two measures are additive in their effect on prices. The revenues generated by these reforms are recycled. The details of the polices are provided below.

Carbon Tax:

- The carbon tax rate in 2025 is \$30/tCO₂ (2017 prices).
- The carbon tax levied per unit of energy use is determined by calculating the carbon content of individual fuels for each unit of energy.
- The carbon tax rate is added to the price of fuel use for all fuel users affected by the tax.
- The carbon tax is only levied on power generation and industry. The residential and transport sectors are not covered by the tax.

Oil and Natural Gas Consumption Subsidies:

- The value of oil and natural gas consumption subsidies are calculated using data from IEA. In the absence of data detailing the relative subsidies received by different fuel users, the per unit subsidy is calculated by dividing total subsidy value by total fuel use. In the baseline, the per unit subsidy is assumed to be constant over time.
- The value of the withdrawn subsidy is added to the prices of oil and natural gas use, for all fuel users affected by the tax.
- The subsidies are withdrawn only from power generation and industry. The residential and transport sectors are not affected.

Revenue Recycling:

- By design, the government implements full revenue recycling so that the tax reforms are revenue neutral. The full value of the carbon tax receipts, and the savings from subsidy removal are offset by compensating measures.
- In both scenarios, a share of the revenue from ETR is allocated to compensate the textiles sector for the negative impacts it faces from ETR. The value of this compensation is equal to the reduction in gross value added (GVA) in the textiles sector in each scenario. The compensation is in the form of 'cleantech' investment, directed to improve the sustainability of production practices in the Bangladeshi textile industry.
- In the first scenario, remaining revenue is spent on a programme of public investment in infrastructure. The exact infrastructure projects are not defined; the supply side effects of the policy are not modelled.
- In the second scenario, remaining revenue is spent on benefit payments to the bottom two quintiles in the income distribution. Benefit payments are spent equally on the two quintiles. The exact nature of the payment is not defined; in the modelling framework it is treated as a basic transfer from government to households.

Note that if a carbon tax was levied on transport fuels, total revenues would be notably larger, given that transport is responsible for over 11 percent of fossil fuel consumption in Bangladesh. Subsidy removal would be less significant because the unit subsidy on oil is relatively small. The effects of tax reform would then be directly felt by households. Consumer price rises would be larger. Further, the energy import bill would reduce substantially more than in the current scenario. Oil is the main component of the energy import bill, and transport, currently exempt in the scenario, is the largest consumer of oil.

Limitations and Assumptions

- A 'baseline' scenario was developed for the modelling. The importance of the baseline is minimal, because the modelling framework solves scenarios as differences from the baseline. The assumed GDP growth in the baseline is 7.1% per annum to 2030, taken from 'The World in 2030' (HSBC, 2018). The report provides details of its assumptions to produce the growth forecasts. The working-age population growth rate is assumed to be 1.6% per annum (extrapolation of the 1.6% per annum from 2018-2023 detailed in HSBC (2018)).
- The baseline assumes that the individual components of GDP grow at the same rate as total GDP. Labour force is assumed to grow in proportion to working-age population (constant labour participation rate). Energy use per unit of gross output is assumed to be constant over time. The proportion of income accruing to each quintile is assumed to be constant over time.
- Total power generation equals demand in each given year. Composition of the power generation mix is assumed to be constant over time. This is equivalent to assuming that as any capacity retires, it is replaced with the same technology. And that investment in new capacity follows historical shares.
- Environmental tax reforms are modelled to be introduced linearly to 2024. By 2024, the carbon tax rate is \$30/tCO₂ (2017 prices), and oil and natural gas subsidies are fully removed for power generation and industry.
- As noted above, by design, each scenario is budget-neutral in terms of policy costs/revenues.
- The behavioural parameters in the model are taken from E3ME. Bangladesh is not represented individually in E3ME; India was therefore chosen as the most appropriate region to be the proxy.
- Switching between fuels is not estimated in the modelling. Coal, oil, and gas use long-run price elasticities from E3ME. The values of these estimates are taken from literature, rather than econometrically estimated. See the E3ME manual (https://www.e3me.com/wpcontent/uploads/2019/04/E3ME-Technical-Manual-v6.1.pdf).
- The cost pass-through rate in each sector (share of change in costs passed on to final prices) is determined by the econometric parameters; importantly, reflecting different levels of competitiveness in individual sectors. The exception is power generation, which is not subject to international competition: here, the full cost of taxes is passed through to prices.
- Change in demand for oil is directed to imports, given the extant minimal domestic production in Bangladesh. Change in demand for coal is met by a reduction in domestic production, and reduction in imports, proportional to the import share in 2016. Change in demand for gas is met entirely by adjustments to domestic production. Bangladesh's cumulated natural gas consumption to 2030, in the modelling baseline, is less than total natural gas reserves (data for natural gas reserves is taken from BGR (2015) (https://www.bgr.bund.de/EN/Themen/Energie/Downloads/energiestudie_2015 _en.pdf?__blob=publicationFile&v=2). This assumption would need to be reconsidered if the modelling horizon extended beyond 2030.
- **National policy.** The measures are introduced in the model on a national level in Bangladesh. Policy in all other global regions is constant across scenarios.
- **Price effects.** The model captures price effects and does not include any awareness or signalling effects from the green taxes. This means that the responses to changes in tax rates should be attributed to the financial effects, rather than any publicity or virtue-signalling that accompanies the reforms.
- Prices. The model is solved in constant prices, in Bangladeshi taka, 2016 prices.
- **Impact beyond 2025.** The tax reform is introduced gradually to 2024. The model assumes that by 2024, full behavioural responses are realised. Effects from 2025 onwards would be very similar, unless further reform was introduced. The caveat is that Bangladesh has limited natural gas reserves; economic benefits of reduced gas use would increase if natural gas was not sourced domestically.

Scenario results

Scenario results

The ETR policies are modelled across two scenarios of revenue recycling. For the 'infrastructure scenario', revenues are spent on a programme of investment in public infrastructure. For the 'social spending scenario', revenues are spent on increasing the real incomes of the lowest two income quintiles.

Variable	Infrastructure	Consumption
GDP	0.4	0.5
Consumer Expenditure	-0.6	0.4
Investment	3.3	0.8
Imports	0.3	0.3
Exports	-0.3	-0.3
Total CO2 Emissions	-4.0	-3.7
Employment	0.2	0.2
Employment ('000s)	139	172
Environmental Tax Revenue as a Share of Total Tax Revenue	11.8%	11.9%
Total Environmental Tax Receipt (Million 2017 USD)	4305	4314
Total Final Energy Consumption (% baseline)	-3.0	-2.6
Total Tax 2020-2025 Cumulative (% baseline)	8.3%	8.3%

National Level Results

GDP

GDP is higher than the baseline in both scenarios: 0.4% in the infrastructure scenario, and 0.5% in the social spending scenario. The main driver of positive macroeconomic impacts is the effect of redistributing profits to investment and consumption. Prices in the economy generally increase less than the cost of the environmental taxes, subject to the level of competition in each sector, leading to a reduction in profits. The full values of the taxes are recycled, however, leading to a net stimulus effect. Reductions in energy use also produce positive macroeconomic effects because of the reduction in the value of energy imports.

Consumption (Household Expenditure)

In the design of the tax reform, none of the reforms directly affect households. Consumer prices do increase across consumption categories, however, because power generation and industry face higher costs of production.

- In the infrastructure scenario, total consumer expenditure is 0.6% lower than it is in the baseline, in real terms. Nominal incomes are higher than in the baseline; the infrastructure investment creates jobs and there are positive multiplier effects throughout the economy. Price rises outweigh this effect, however, meaning household income and consumption are lower in real terms.
- In the social spending scenario, total consumer expenditure is 0.4% higher than the baseline, in real terms. The revenue recycled in social spending increases consumer expenditure and creates a stimulus effect which creates jobs. These effects are substantial enough to outweigh the effect of price rises.

Investment

Investment is higher in both scenarios than in the baseline. The positive macroeconomic results in both scenarios mean higher production levels in the economy, which stimulates investment in additional

productive capacity. The investment increase is much greater in the infrastructure scenario, because the revenue recycling mechanism directly increases investment in the economy. Investment is 0.8% higher in the social spending scenario, and 3.3% higher in the infrastructure scenario.

Imports and Exports

- Total imports are higher in both scenarios than in the baseline, by a similar amount. Higher levels of production in the economy increases the demand for imports of production inputs. Part of the higher consumer expenditure in the social spending scenario is spent on imported goods. Imports are a 'leakage' from the domestic economy, reducing the positive multiplier effects of the investment and consumption stimuli.
- Energy imports are lower in both scenarios, because higher prices of fuel use reduce demand. The value of coal imports is approximately 34% lower in both scenarios, and oil imports are 1% lower. The reduction in coal imports is much greater because: 1) the proportional price increase of coal is substantially more than for oil (coal is less expensive per unit of energy, and contains more carbon); 2) transport is protected from the tax reforms, and transport is responsible for 60% of the final energy consumption of oil products.
- Total exports are lower than in the baseline in both scenarios. Bangladeshi firms face higher costs of production because of the direct tax incidence, and the higher prices of domestic production inputs. Resulting price increases reduce the competitiveness of exports from Bangladesh. If other regions globally were to implement comparable environmental policy, this competitiveness effect would be mitigated. Investment is made in the sustainability of production in the textiles sector, to compensate for this effect, and protect the long-term competitiveness of Bangladesh's most important export sector.

Employment

Total employment increases in both scenarios, by 0.2%. Employment increases as a result of higher economic activity in each scenario. The revenue recycling mechanisms more than outweigh the negative effect on output of price increases due to higher energy costs.

Energy Consumption

Total final energy consumption falls in both scenarios, as expected. Carbon tax and subsidy removal directly increase the price of fuel use, reducing demand. In the absence of revenue recycling, energy consumption in the model is reduced by 3.9%. In the infrastructure and social spending scenarios, however, energy consumption decreases by 3.0% and 2.6%, respectively, given the rebound effects of higher production and consumer expenditure.

CO₂ emissions

Both scenarios have lower total CO_2 energy emissions than the baseline. CO_2 emissions fall by 4.0% and 3.7% in the infrastructure and social spending scenarios, respectively. The reduction in CO_2 is greater than energy consumption; the carbon tax increases the price of coal use substantially given its baseline low price relative to other fuels, and high carbon content. The tax levied on natural gas and oil is a much lower share of the baseline price.

Air Pollution

None of tax reforms in the scenarios directly affect air pollution, but given the reforms reduce use of polluting energy products, air pollutants such as SOx, NOx, PM_{10} and $PM_{2.5}$ would be expected to fall as well. These effects have not been quantified in the modelling.

Inequality/Real Incomes

Real incomes in the scenarios are affected by changes in wage income, income from government social spending, and prices. Households in different income groups spend their income in different ways. The important differences in the context of ETR are expenditure on energy and energy intensive

goods/services. In Bangladesh, higher income households spend a larger share of income in these areas, and therefore face higher price increases.

- In the infrastructure scenario, there are no explicit redistributive policies. The scenario is progressive, however, because the lower income quintile households experience smaller reductions in real incomes than those in the higher quintiles. Overall, the differences are small; the lowest quintile has a 0.4% real income reduction, compared to 0.6% for the highest quintile.
- In the social spending scenario, the inequality results are substantially more marked. The revenue recycling mechanism is designed to improve the incomes of the bottom two quintiles. Real income in the highest three quintiles reduces by 0.6%. But real income increases by 12.3% for the lowest quintile, and 4.9% for the second lowest.

Sector Level Results

Output

The main differences in the output structure of the economy are determined by changes in exports, and the revenue recycling mechanism. The modelling framework uses a fixed input-output structure and does not include a treatment of consumption substitution across categories.

- In the infrastructure scenario, the sectors with largest increases in output are construction, manufacturing (excluding textiles) and mining and quarrying. The public infrastructure investment increases demand substantially in each of these sectors, either directly, or indirectly through supply chains. Gross output is marginally lower in sectors which are supplying consumer final demands, such as retail.
- In the social spending scenario, gross output increases in all sectors except textiles. The sectors with the largest increases are those associated with consumer expenditure. Manufacturing and construction also have relatively high increases, because of the investment in 'cleantech' for textiles. Textiles loses out because of a loss of exports.

Employment

The percentage changes in employment by sector largely follow the pattern of changes in output, whilst allowing for different returns to scale. Therefore, the largest changes are found in labour-intensive sectors, and those where output changes by the most. In the infrastructure scenario, 63% of new jobs are created in the construction sector. Agriculture creates the second greatest number of jobs, because of the size of the sector and its labour-intensity. In the social spending scenario, employment changes are more even across the economy. Wholesale and retail trade create the most jobs (53% of the total), followed by agriculture and construction. Jobs are created in construction because of the investment in 'cleantech' for textiles, and the induced positive effects in economy-wide investment.

\mathbf{CO}_2

The largest proportional reductions in CO₂ in both scenarios is across the industrial sectors. This is because, in Bangladesh, industry is responsible for approximately 85% of coal use, the fuel which is most affected by these reforms. The fuel user contributing least, proportionally, to emissions reductions is the residential sector; the residential sector was protected in scenario design from the tax reforms. In the social spending scenario, emissions from the residential sector increase, as total consumer expenditure increases. The power generation sector has minimal changes in emissions, because of limitations in the modelling framework. The model assumes that the power generation mix is constant over time and across scenarios. These tax reforms provide a strong incentive to different power generation technologies. However, the time horizon of analysis is 2025 and given the lifetime of power generation technologies, there would be limited retirement of capacity to be replaced over these years.

Energy Consumption

Changes in energy consumption by fuel user follow a very similar pattern to the CO_2 results, though the differences across sectors are less marked. Where industrial sectors reduce coal use, the relative CO_2 reduction is larger than that of energy use.

Textiles Sector

The textiles sector is unique in the Bangladeshi economy, and in the dynamics it faces from ETR. Textiles is the only sector that is reliant on exports, rather than domestic demand. It is therefore the most vulnerable to the loss of international competitiveness arising from increased costs of production resulting from tax reform. Exports reduce by 0.4% in both scenarios by 2025. A portion of the revenue accruing from the tax reforms is allocated to investing in the sustainability of production in the textiles industry. This investment would ameliorate the negative price competitiveness effects on the industry, with positive quality effects. These effects have not been quantified in the model.

Model Classifications

Sectors	
1. Agriculture, forestry & fishing	
2. Mining & quarrying	
3. Manufacturing (excl. textiles)	
4. Textiles	
5. Electricity gas, & water supply	
6. Construction	
7. Wholesale & retail trade, & repair	
8. Transport, storage, & communication	
9. Financial intermediations	
10. Real estate renting, & business activities	
11. Other service activities	
12. Public administration & defence	
Fuel Users	
1. Agriculture, forestry & fishing	
2. Mining & quarrying	
3. Manufacturing (excl. textiles)	
4. Textiles	
5. Electricity gas, & water supply	
6. Construction	
7. Wholesale & retail trade, & repair	
8. Transport, storage, & communication	
9. Financial intermediations	
10. Real estate renting, & business activities	
11. Other service activities	
12. Public administration & defence	
13. Residential	

Consumption Categories

- 1. Food & non-alcoholic beverages
- 2. Alcoholic beverage, tobacco & narcotics
- 3. Clothing & footwear
- 4. Housing, water, electricity, gas & other fuels

5. Furnishings, household equipment & household maintenance

- 6. Health
- 7. Transport
- 8. Communication
- 9. Restaurants & hotels
- 10. Miscellaneous goods & services

Fuels

- 1. Coal
- 2. Oil
- 3. Natural Gas
- 4. Electricity
- 5. Biofuel

Appendix B: Sources

³ This report includes statements by the author that were previously published by ACCA (2018), Tax as a force for good; Rebalancing our tax systems to support a global economy fit for the future. Femke Groothuis. https://www.accaglobal.com/content/dam/ACCA_Global/professional-insights/Tax-as-force-for-good/pi-

environmental-tax.pdf

⁴ Landrigan, P.J., Fuller, R. (et al.) (2018), The Lancet Commission on pollution and health. The Lancet, Vol. 391, No. 10119, pp. 462–512. <u>https://www.thelancet.com/pdfs/journals/lancet/PIIS0140-6736(17)32345-0.pdf</u>
 ⁵ World Bank (2015), Climate Change Complicates Efforts to End Poverty.

http://www.worldbank.org/en/news/feature/2015/02/06/climate-change-complicates-efforts-end-poverty ⁶ Burek, P. (et al.) (2016), Water futures and solution-fast track initiative.

http://pure.iiasa.ac.at/id/eprint/13008/1/WP-16-006.pdf

⁷ Abeygunawardena, P., (et al.) (2003), Poverty and Climate Change. Part 1. Reducing the Vulnerability of the Poor through Adaptation.

http://documents.worldbank.org/curated/en/534871468155709473/pdf/521760WP0pover1e0Box35554B01PUBLI C1.pdf

⁸ IPBES (2019), Summary for policymakers of the global assessment report on biodiversity and ecosystem services of the Intergovernmental Science-Policy Platform on Biodiversity and Ecosystem Services. <u>https://www.ipbes.net/sites/default/files/downloads/spm_unedited_advance_for_posting_htn.pdf</u>

⁹ FAO (2018), Food Loss and Food Waste. http://www.fao. org/food-loss-and-food-waste/en/

¹⁰ WEF (World Economic Forum), Ellen MacArthur Foundation, McKinsey & Company (2016), The New Plastics Economy: Rethinking the Future of Plastics. <u>http://www.weforum.org/reports/the-new-plastics-economy-rethinking-the-future-of-plastics</u>

¹¹ ILO (2019), World Employment Social Outlook: Trends 2019. <u>https://www.ilo.org/wcmsp5/groups/public/---</u> <u>dgreports/---dcomm/---publ/documents/publication/wcms_670542.pdf</u>

¹² In 2018. ILO (2019), World Employment Social Outlook: Trends 2019.

https://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/---publ/documents/publication/wcms_670542.pdf ¹³ United Nations (2015), Transforming our world: the 2030 Agenda for Sustainable Development.

https://sustainabledevelopment.un.org/content/documents/21252030 Agenda for Sustainable Development web.pdf

¹⁴ United Nations (2014), World Investment Report 2014: Investing in the SDGs: An Action Plan. <u>https://unctad.org/en/PublicationsLibrary/wir2014_en.pdf</u>

¹⁵ OECD (2019), Development aid drops in 2018, especially to neediest countries.

http://www.oecd.org/dac/financing-sustainable-development/development-finance-data/ODA-2018-detailed-summary.pdf

¹⁶ OECD (Accessed May 2019), Official Development Assistance (ODA). <u>https://www.oecd.org/dac/financing-</u> sustainable-development/development-finance-standards/official-development-assistance.htm

¹⁷ Denmark, Luxembourg, Norway, Sweden and the United Kingdom. United Nations (May 10, 2018), Progress towards the Sustainable Development Goals Report of the Secretary-General.

https://unstats.un.org/sdgs/files/report/2018/secretary-general-sdg-report-2018--EN.pdf ¹⁸ UNEP FI (2018), Rethinking impact to finance the SDGs.

https://www.unepfi.org/wordpress/wp-content/uploads/2018/11/Rethinking-Impact-to-Finance-the-SDGs.pdf¹⁹ Niculescu, M. (2017), Impact Investment to Close the SDG Funding Gap.

https://www.undp.org/content/undp/en/home/blog/2017/7/13/What-kind-of-blender-do-we-need-to-finance-the-SDGs-.html

²⁰ United Nations (May 10, 2018), Progress towards the Sustainable Development Goals Report of the Secretary-General. <u>https://unstats.un.org/sdgs/files/report/2018/secretary-general-sdg-report-2018--EN.pdf</u>

²¹ Gaspar, V., L. Jaramillo, P. Wingender (2016), Tax Capacity and Growth: Is there a Tipping Point? IMF WP No.
 16. <u>https://www.imf.org/external/pubs/ft/wp/2016/wp16234.pdf</u>

²² Top-20 countries (excluding the largest oil exporters are excluded, because they have alternative revenue sources): Nigeria, Indonesia, Bangladesh, Mexico, Sudan, Pakistan, Egypt, Myanmar, Philippines, Malaysia, Guatemala, Panama, Peru, Sri Lanka, Ethiopia, Democratic Republic of Congo, Tanzania, Afghanistan, Algeria and Cameroon. Junquera-Varela, Felix, Haven, Bernard (December 18, 2018), Getting to 15 percent: addressing the largest tax gaps. <u>https://blogs.worldbank.org/governance/getting-15-percent-addressing-largest-tax-gaps</u>

²³ United Nations (2015), Transforming our world: the 2030 Agenda for Sustainable Development. <u>https://sustainabledevelopment.un.org/content/documents/21252030 Agenda for Sustainable Development web.pdf</u>

¹ IPCC (2018), Global warming of 1.5°C. An IPCC Special Report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. https://www.ipcc.ch/site/assets/uploads/sites/2/2019/06/SR15_Full_Report_Low_Res.pdf

² UN DESA (Accessed November 2018), World Population Projected to Reach 9.8 billion in 2050, and 11.2 billion in 2100. <u>https://www.un.org/development/desa/en/news/population/world-population-prospects-2017.html</u>

²⁴ World Bank (2018), Taxation and the Sustainable Development Goals. Conference Report (English). Platform for Collaboration on Tax. <u>http://documents.worldbank.org/curated/en/860581538762337418/Taxation-and-the-Sustainable-Development-Goals-Conference-Report</u>

²⁵ World Bank (2018), Taxation and the Sustainable Development Goals. Conference Report (English). Platform for Collaboration on Tax. <u>http://documents.worldbank.org/curated/en/860581538762337418/Taxation-and-the-Sustainable-Development-Goals-Conference-Report</u>

²⁶ Of the total, Latin America and the Caribbean received \$5.1 billion, sub-Saharan Africa received \$4.6 billion and Southern Asia received \$3.8 billion. The three main sectors assisted were public administration, environment and energy, which received a total of \$10.2 billion. United Nations (May 10, 2018), Progress towards the Sustainable Development Goals Report of the Secretary-General.

https://unstats.un.org/sdgs/files/report/2018/secretary-general-sdg-report-2018--EN.pdf

²⁷ United Nations (2014), World Investment Report 2014: Investing in the SDGs: An Action Plan. https://unctad.org/en/PublicationsLibrary/wir2014 en.pdf

²⁸ BSDC (Business and Sustainable Development Commission) (2017), Better Business, Better World: Executive Summary <u>http://report.businesscommission.org/uploads/Executive-Summary.pdf</u>.

²⁹ UNEP FI (2018), Rethinking impact to finance the SDGs.

https://www.unepfi.org/wordpress/wp-content/uploads/2018/11/Rethinking-Impact-to-Finance-the-SDGs.pdf ³⁰ Ellen MacArthur Foundation (2017), A new textiles economy: Redesigning fashion's future. http://www.ellenmacarthurfoundation.org/publications

³¹ Tearfund, Fauna & Flora International (FFI), WasteAid and The Institute of Development Studies (IDS) (2019), No time to waste; Tackling the plastic pollution crisis before it's too late. <u>http://live-fauna-flora-international.pantheonsite.io/wp-content/uploads/2019/05/2019</u> No-time-to-waste-report.pdf

³² Landrigan, P.J., Fuller, R. (et al.) (2018), The Lancet Commission on pollution and health. The Lancet, Vol.
 391, No. 10119, pp. 462–512. <u>https://www.thelancet.com/pdfs/journals/lancet/PIIS0140-6736(17)32345-0.pdf</u>
 ³³ Ricke, K., Drouet, L., (et al.) (2018), Country-level Social Cost of Carbon', Nature Climate Change, 8: 895–
 900. <u>https://www.nature.com/articles/s41558-018-0282-y</u>, as mentioned in Nature (September 25, 2018), The costs of climate inaction. <u>https://www.nature.com/articles/d41586-018-06827-x</u>

³⁴ UNEP (2014), Valuing plastics: the business case for measuring, managing and disclosing plastic use in the consumer goods industry. <u>http://wedocs.unep.org/handle/20.500.11822/9238</u>

³⁵ World Bank and Institute for Health Metrics and Evaluation (2016) cited in ILO (2018), World Employment and Social Outlook 2018 - Greening with Jobs. <u>https://www.ilo.org/weso-</u>

greening/documents/WESO Greening EN web2.pdf

³⁶ ILO (2018), World Employment and Social Outlook 2018 - Greening with Jobs. <u>https://www.ilo.org/weso-greening/documents/WESO_Greening_EN_web2.pdf</u>

³⁷ UNEP Finance Initiative (2019), Changing Course; A comprehensive investor guide to scenario-based methods for climate risk assessment, in response to the TCFD. <u>https://www.unepfi.org/wordpress/wp-</u>content/uploads/2019/05/TCFD-Changing-Course.pdf

³⁸ The notion of putting a price on externalities has been around since Arthur Pigou introduced the concept in 1920. Pigou, A.C. (1920), The Economics of Welfare.

³⁹ OECD (Accessed May 2019), Glossary of Statistical Terms. <u>https://stats.oecd.org/glossary/</u>

⁴⁰ Barrios, S., Pycroft, J., Saveyn, B. (2013), The Marginal Cost of Public Funds in the EU: The Case of Labour versus Green Taxes, EC Working Paper No.35.

https://ec.europa.eu/taxation_customs/sites/taxation/files/resources/

<u>documents/taxation/gen_info/economic_analysis/tax_papers/taxation_paper_35_en.pdf;</u> European Commission (2013), The 2013 Annual Growth Survey: Towards Fair and Competitive Tax Systems.

<u>http://europa.eu/rapid/press-release MEMO-12-915 en.htm;</u> European Commission (2014), Annual Growth Survey 2014. <u>https://ec.europa.eu/info/system/files/2014-european-semester-annual-growth-survey-en.pdf;</u> European Commission (2015), Tax Reforms in EU Member States 2015. Tax Policy Challenges for Economic Growth and Fiscal Sustainability. <u>https://ec.europa.eu/info/sites/info/files/file_import/ip008_en_2.pdf</u>

⁴¹ Pavel, J., Vitek, L. (2012), Transaction Costs of Environmental Taxation: The Administrative Burden. http://dx.doi.org/10.4337/978178 1952146.00025

⁴² Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. <u>https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe</u> <u>d=y</u>

⁴³ Fankhausera, S., Hepburn, C., Park, J. (2011), Combining Multiple Climate Policy Instruments: How Not to Do It, Centre for Climate Change Economics Policy Working Paper No. 48; Grantham Research Institute on Climate Change and the Environment Policy Working Paper No. 38. <u>http://www.lse.ac.uk/GranthamInstitute/wpcontent/uploads/2014/02/WP38_UK-carbon-tax.pdf</u>

⁴⁴ Karplus, V. J., Paltsev, S. (et al.) (2013), 'Should a Vehicle Fuel Economy Standard be Combined with an Economy-Wide Greenhouse Gas Emissions Constraint? Implications for Energy and Climate Policy in the United States', Energy Economics.

https://pdfs.semanticscholar.org/2b4d/dc66a60d1ce3b0fb4ef44b554e90d879b2e9.pdf

⁴⁵ Environmental tax as a share of GDP declined in 47 out of 76 countries for which data are available in the OECD database. OECD.stat (Accessed May 2019), Environmentally related tax revenue. <u>https://stats.oecd.org/#</u>
⁴⁶ The latest period for which OECD has environmental tax data is used for each country; (2011: India); (2014: Brazil, Korea, Malaysia, Canada); 2015: China, the Philippines, Kazakhstan, South Africa, Rwanda, Cameroon and Singapore); (2016: Japan, Costa Rica, Mexico, US). OECD.stat (Accessed June 1, 2019), Environmentally related tax revenue. <u>https://stats.oecd.org/#</u>

⁴⁷ Shirai, T. and Adam, Z. (2017), Commentary: Fossil-fuel Consumption Subsidies are Down, but not Out.
 <u>http://www.iea.org/newsroom/news/2017/december/commentary-fossil-fuel-consumption-subsidies-are-down-but-not-out.html?utm_content=buffera5c72&utm_medium=social&utm_source=twitter.com&utm_campaign=buffer
 ⁴⁸ WTO (2006), World Trade Report 2006: Exploring the Links between Subsidies, Trade and WTO.
 <u>https://www.wto.org/english/res_e/booksp_e/anrep_e/world_trade_report06_e.pdf</u>
</u>

⁴⁹ Carrington, D. and Davies, H. (2015), US taxpayers subsidising world's biggest fossil fuel companies. <u>http://www.theguardian.com/environment/2015/may/12/us-taxpayers-subsidising-worlds-biggest-fossil-fuel-companies</u>

⁵⁰ OECD (2015), OECD Companion to the Inventory of Support Measures for Fossil Fuels 2015. <u>https://read.oecd-ilibrary.org/energy/oecd-companion-to-the-inventory-of-support-measures-for-fossil-fuels-2015_9789264239616-en#page1</u>

⁵¹ OECD (2018), OECD Companion to the Inventory of Support Measures for Fossil Fuels 2018. <u>https://read.oecd-ilibrary.org/energy/oecd-companion-to-the-inventory-of-support-measures-for-fossil-fuels-</u> 2018 9789264286061-en#page11

 ⁵² Asian Development Bank (ADB) (2016) Fossil Fuel Subsidies in Asia Trends, Impacts, and Reforms. Integrative Report. <u>https://www.adb.org/sites/default/files/publication/182255/fossil-fuel-subsidies-asia.pdf</u>
 ⁵³ OECD (2018), OECD-IEA Fossil Fuel Support and Other Analysis: Methodology. http://www.oecd.org/site/tadffss/methodology/

⁵⁴ 'Pre-tax subsidies' (estimated at \$296 billion in 2017) reflect the difference between what people pay for a fuel and what it cost to produce. 'Post-tax subsidies' reflect the difference between consumer fuel prices and the full societal and environmental costs of a fuel (estimated at \$5.2 trillion in 2017). The largest subsidizers in 2015 were China (\$1.4 trillion), United States (\$649 billion), Russia (\$551 billion), European Union (\$289 billion), and India (\$209 billion). By region, Emerging/Developing Asia accounts for nearly 40 percent of global energy subsidies, followed by Advanced Economies (27 percent), Commonwealth of Independent States (15 percent), Middle East, North Africa, Afghanistan, and Pakistan (9 percent), Latin America/Caribbean (5 percent), Emerging/Developing Europe (3 percent), and Sub-Sahara Africa (2 percent). By fuel, coal remains the largest source of subsidies (44 percent), followed by petroleum (41 percent), natural gas (10 percent), and electricity output (4 percent). IMF (2019), Global Fossil Fuel Subsidies Remain Large: An Update Based on Country-Level Estimates. IMF working paper, prepared by David Coady, Ian Parry, Nghia-Piotr Le, and Baoping Shang. https://t.co/2n6lgtrD6l?amp=1 ⁵⁵ Asian Development Bank (ADB) (2016), Fossil Fuel Subsidies in Asia Trends, Impacts, and Reforms. Integrative Report. https://www.oecd.org/site/default/files/publication/182255/fossil-fuel-subsidies-asia.pdf ⁵⁶ OECD (2019), OECD-IEA analysis of fossil fuels and other support. https://www.oecd.org/site/tadffss/

⁵⁷ Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. <u>https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe</u> d=y

⁵⁸ António Guterres (May 28, 2019), Message on Twitter, accompanying the announcement of 'The Secretary-General's remarks at the R20 Austrian World Summit'.

https://twitter.com/antonioguterres/status/1133313859094237184?s=11 ⁵⁹ World Bank Group (2019), State and Trends of Carbon Pricing 2019.

https://openknowledge.worldbank.org/handle/10986/31755

⁶⁰ Oceana (2017), Save the oceans feed the world. <u>https://oceana.org/reports/save-oceans-feed-world</u>

⁶¹ Tax revenues would have been higher by 3.8% of global GDP. IMF (2019), Global Fossil Fuel Subsidies Remain Large: An Update Based on Country-Level Estimates. IMF working paper, prepared by David Coady, Ian Parry, Nghia-Piotr Le, and Baoping Shang. <u>https://t.co/2n6lgtrD6l?amp=1</u>

⁶² Global Commission on the Economy and Climate (2018), Unlocking the inclusive growth story of the 21st century: Accelerating climate action in urgent times. New Climate Economy.

http://www.newclimateeconomy.report

⁶³ European Commission (2018), Taxation Trends in the European Union. p264.

https://ec.europa.eu/taxation_customs/sites/taxation/files/taxation_trends_report_2018.pdf

⁶⁴ Labour taxes as a share of GDP increased in 53 out of 73 countries for which data are available in the OECD database over the 2000-2016 period. Labour taxes are assumed to consist of the sum of the following line items in the OECD database: 1) (1100) 'taxes on income of individuals'; 2) (2000) 'social security contributions' and 3) (3000) 'payroll taxes'. Costa Rica's earliest available data for PIT: 2003. OECD (Accessed May 2019), Dataset: Revenue Statistics - OECD countries: Comparative tables. <u>https://stats.oecd.org/Index.aspx?DataSetCode=REV</u>

⁶⁵ For a single individual without children at the income level of the average worker. The tax wedge varies between different types of household and income intervals. OECD (2018), Taxing Wages 2018. <u>https://www.oecd.org/tax/tax-policy/taxing-wages-brochure.pdf</u>

⁶⁶ The latest period for which OECD has environmental tax data is used for each country; (2014: Brazil, Korea, Malaysia); (2015: the Philippines, Kazakhstan, South Africa, Rwanda, Cameroon and Singapore); (2016: Japan, Costa Rica, Mexico). OECD.stat (Accessed June 2019), Environmentally related tax revenue.

https://stats.oecd.org/#; OECD.stat (Accessed June 2019), Global Revenue Statistics Database. https://stats.oecd.org/#;

⁶⁷ OECD (2011), Taxation and Employment. Tax Policy Studies No21. <u>http://www.oecd.org/ctp/48859441.pdf</u>
 ⁶⁸ Perret, Sarah (2018), Tax design for inclusive growth in OECD countries.

https://www.un.org/development/desa/dspd/wp-content/uploads/sites/22/2018/06/8-1.pdf

⁶⁹ Labour demand and supply are affected by different aspects of labour taxation. See for example: Brys, B. (2011), 'Making Fundamental Tax Reform Happen'. <u>https://doi.org/10.1787/5kg3h0v54g34-en</u>; Dolenc P. and Laporšek, S. (2010), 'Tax Wedge on Labour and its Effect on Employment Growth in the European Union'. <u>https://www.vse.cz/polek/download.php?jnl=pep&pdf=381.pdf;</u> Pierluigi, B. and Roma, M. (2008), Labour Cost and Employment Across Euro Area Countries and Sectors.

<u>https://www.ecb.europa.eu/pub/pdf/scpwps/ecbwp912.pdf;</u> Vermeend, W., van der Ploeg, R. and Timmer, J. (2008), Taxes and the Economy; European Commission (1993), Growth, competitiveness and employment. Challenges and the ways forward into the 21th century.

http://aei.pitt.edu/1139/1/growth_wp_COM_93_700_Parts_A_B.pdf

⁷⁰ Dolenc P., Laporšek, S. (2010), Tax Wedge on Labour and its Effect on Employment Growth in the European Union. <u>https://www.vse.cz/polek/download.php?jnl=pep&pdf=381.pdf</u>

⁷¹ OECD (2011), Taxation and Employment. Tax Policy Studies No21. <u>http://www.oecd.org/ctp/48859441.pdf</u> ⁷² Bocconi University (2011), The role and impact of labour taxation policies.

http://ec.europa.eu/social/BlobServlet?docId=7404&langId=enQ

⁷³ OECD (2018), Opportunities for All: A Framework for Policy Action on Inclusive Growth. <u>https://read.oecdilibrary.org/economics/opportunities-for-all_9789264301665-en#page1</u>; Perret, Sarah (2018), Tax design for inclusive growth in OECD countries. [Note for the Expert Group Meeting on "Addressing inequalities and challenges to social inclusion through fiscal, wage and social protection policies", held at UN Headquarters in New York on 25–27 June 2018],

https://www.un.org/development/desa/dspd/wp-content/uploads/sites/22/2018/06/8-1.pdf; OECD (2011), Tax reform can create jobs. http://www.oecd.org/newsroom/taxreformcancreatejobs.htm

⁷⁴ Bogetic, Z., Onder, H., Onal, A., Skrok, E., Schwartz, A., Winkler, H. (2015), Fiscal Policy Issues in the Aging Societies. <u>http://documents.worldbank.org/curated/en/846591468327934918/pdf/95266-REVISED-NWP-PUBLIC-Box391483B-MFM-Discussion-Paper-1-ZB-May-2015.pdf</u>; World Bank (2018), World Development Report 2019. http://documents.worldbank.org/curated/en/816281518818814423/pdf/2019-WDR-Report.pdf

⁷⁵ IMF (September 30, 2014), The IMF's Advice on Labor Market Issues. Fact Sheet.

<u>https://www.imf.org/external/np/exr/facts/labor.htm;</u> IMF (2015), Fiscal Policy and Long-Term Growth. IMF Policy Paper. <u>https://www.imf.org/external/np/pp/eng/2015/042015.pdf</u>

⁷⁶ European Commission (2017), Joint Employment Report 2017.

http://ec.europa.eu/social/BlobServlet?docld=17224&langId=en; European Commission (2013), Tax Reforms in EU Member States 2013. Tax Policy Challenges for Economic Growth and Fiscal Sustainability.

https://ec.europa.eu/economy finance/publications/european economy/2013/ee5 en.htm.

⁷⁷ The Eurogroup comprises the finance ministers of countries whose currency is the euro. Eurogroup (2014), 'Structural Reform Agenda - Thematic Discussions on Growth and Jobs - Reduction of the Tax Wedge'. <u>http://italia2014.eu/media/2066/eurogroup-statement- 12-september-2014.pdf</u>

⁷⁸ Singh, V, (2014), Impact of Labour Cost in Overall Performance of Hotels and Restaurants.

http://www.erpublications.com/sitepanel/download.php?filename=download/download 28 10 2014 12 26 27.pd f

⁷⁹ See for example: "Pushing more shoppers to scan their own items and make payments without the help of a cashier, has the potential to save Wal-Mart millions of dollars. For every one second in average transaction time at the Walmart U.S. chain, the company said it spends about \$12 million in cashier wages". Wohl, J. (2012), Wal-Mart to Add more Self-checkout Lanes. <u>https://www.reuters.com/article/walmart/update-2-wal-mart-to-add-more-self-checkout-lanes-idUSL2E8E72FF20120307</u>

⁸⁰ Hoskins, Tansy (July 16, 2016), Robot factories could threaten jobs of millions of garment workers. The Guardian. <u>https://www.theguardian.com/sustainable-business/2016/jul/16/robot-factories-threaten-jobs-millions-garment-workers-south-east-asia-women</u>

⁸¹ Segran, E. (2017), These Fashion Startups Offer the Prestige of "Made In Italy" without Inflated Prices. <u>https://www.fastcompany.com/3067010/these-footwear-startups-offer-the-prestige-of-made-in-italy-without-inflat</u>

⁸² Croxford, Rianna (2018), UN Agency's U-turn after Unpaid Internships Row. https://www.bbc.com/news/world-45605768 ; He, H. (2018), Small Chinese Firms Seek "Lessons in Survival" as they Brace for Impact of Social Welfare Taxes. <u>https://www.scmp.com/economy/china-economy/article/2166266/small-chinese-firms-seek-lessons-survival-they-brace-impact</u>

⁸³ Milanez, A. and B. Bratta (2019), Taxation and the future of work: How tax systems influence choice of employment form, OECD Taxation Working Papers, No. 41. <u>https://doi.org/10.1787/20f7164a-en</u>.
 ⁸⁴ Maxim, R. and Muro, M. (2018), Rethinking Worker Benefits for an Economy in Flux.

https://www.brookings.edu/blog/the-avenue/2018/03/29/rethinking-worker-benefits-for-an-economy-in-flux/ ⁸⁵ Ghosh, S. (April 2018), Undercover Author Finds Amazon Warehouse Workers in UK "Peed in Bottles" over Fears of Being Punished for Taking a Break. <u>https://www.businessinsider.nl/amazon-warehouse-workers-have-to-pee-into-bottles-2018-4/?international=true&r=US</u>

⁸⁶ See for example: Barrett, Paul, Baumann-Pauly, Dorothée (2019), Made in Ethiopia: Challenges in the Garment Industry's New Frontier. New York University Stern for Business and Human Rights.

https://issuu.com/nyusterncenterforbusinessandhumanri/docs/nyu_ethiopia_final_online?e=31640827/69644612; Heath, M. (2015), China is Set to Lose Manufacturing Crown: Manufacturers will be Drawn to Southeast Asia's Strengths, including the Strategic Location and Cheap Labor of Myanmar, Cambodia and Laos. https://www.bloomberg.com/news/articles/2015-04-29/china-is-set-to-lose-manufacturing-crown

⁸⁷ Reuters (July 7, 2016), Millions of SE Asian jobs may be lost to automation in next two decades: ILO. <u>https://www.reuters.com/article/us-southeast-asia-jobs/millions-of-se-asian-jobs-may-be-lost-to-automation-in-next-two-decades-ilo-idUSKCN0ZN0HP</u>

⁸⁸ Chang, J. H., Huynh, P. (2016), ASEAN in transformation the future of jobs at risk of automation. International Labour Organization. <u>https://www.ilo.org/wcmsp5/groups/public/---ed_dialogue/---</u>

act emp/documents/publication/wcms 579554.pd

⁸⁹ UN DESA (2017), World Population Projected to Reach 9.8 billion in 2050, and 11.2 billion in 2100. <u>https://www.un.org/development/desa/en/news/ population/world-population-prospects-2017.html</u>

⁹⁰ The Ex'tax Project is inspired by the concept of Value Extracted Tax (VET). See: Wintzen, Eckart (1994) Reengineering the Planet. Three Steps to a Sustainable Free Market Economy. <u>http://www.ex-</u> tax.com/files/8013/9264/4105/Reengineering the Planet Eckart Wintzen 1993 from Environmental Report B

SO Origin.pdf

⁹¹ Farid, Mai, Keen, Michael, Papaioannou, Michael, Parry, Ian, Pattillo, Catherine, Ter-Martirosyan, Anna and other IMF Staff (2016), After Paris: Fiscal, Macroeconomic, and Financial Implications of Climate Change. IMF Staff Discussion Note. <u>https://www.imf.org/external/pubs/ft/sdn/2016/sdn1601.pdf</u>; IMF (2015), Kingdom of the Netherlands: Staff Concluding Statement of the 2015 Article IV Mission.

https://www.imf.org/en/News/Articles/2015/09/28/04/52/mcs111015a; Parry, Ian, de Mooij, Ruud, Keen, Michael (Eds.) (2012), Fiscal Policy to Mitigate Climate Change. A Guide for Policymakers.

http://www.greenfiscalpolicy.org/wp-

<u>content/uploads/2013/08/Fiscal Policy to Mitigate Climate Change IMF2012 Summary.pdf</u>; Heine, D., Norregaard, J. and Parry, I.W.H. (2012), Environmental Tax Reform: Principles from Theory and Practice to Date. IMF Working Paper. <u>https://www.imf.org/external/pubs/ft/wp/2012/wp12180.pdf</u>.

⁹² Montt, Guillermo, Fraga, Federico, Harsdorff, Marek (2018), The future of work in a changing natural environment: Climate change, degradation and sustainability. ILO Future of Work Research Paper Series. <u>https://www.ilo.org/wcmsp5/groups/public/---dgreports/---cabinet/documents/publication/wcms_644145.pdf;</u> Poschen (et al.) (2012), Working towards sustainable development: Opportunities for decent work and social

inclusion in a green economy. The Green Jobs Initiative (UNEP, ILO, IOE, ITUC). <u>https://www.ilo.org/global/publications/books/WCMS_181836/lang--en/index.htm;</u> Torres, Raymond (Ed.) (2009) World of Work Report 2009: The Global Jobs Crisis and Beyond / International Institute for Labour Studies.

https://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/---publ/documents/publication/wcms_120079.pdf ⁹³ Bogetic, Z., Onder, H., Onal, A., Skrok, E., Schwartz, A., Winkler, H. (2015), Fiscal Policy Issues in the Aging Societies. <u>http://documents.worldbank.org/curated/en/846591468327934918/pdf/95266-REVISED-NWP-PUBLIC-</u> Box391483B-MFM-Discussion-Paper-1-ZB-May-2015.pdf; Fay, M., Hallegatte, S., Vogt-Schilb, A., Roszenberg,

J., Narloch, U. and Kerr, T. (2015), Decarbonising Development. Three Steps to a Zero-Carbon Future. The World Bank. <u>https://www.worldbank.org/content/dam/Worldbank/document/Climate/dd/decarbonizing-</u> development-report.pdf

⁹⁴ IPCC (2018), Summary for Policymakers of IPCC Special Report on Global Warming of 1.5°C Approved by Governments, Chapter 2. <u>http://report.ipcc.ch/sr15/pdf/sr15_chapter2.pdf</u>

⁹⁵ OECD (2018), Economic Policy Reforms 2018. Going for Growth Interim Report. <u>https://read.oecd-</u> <u>ilibrary.org/economics/economic-policy-reforms-2018_growth-2018-en#page1</u>; OECD (2015), Towards Green Growth? Tracking Progress. <u>https://read.oecd-ilibrary.org/environment/towards-green-growth_9789264234437-</u> <u>en#page1</u>; OECD (2015), Environment at a Glance 2015: OECD Indicators. <u>https://read.oecd-</u> <u>ilibrary.org/environment/environment-at-a-glance-2015_9789264235199-en#page1</u>; OECD (2013), Water

Security for Better Lives. OECD Studies on Water.

https://www.oecd.org/env/resources/Water%20Security%20for%20Better%20Lives-%20brochure.pdf ⁹⁶ European Parliament (April 15, 2015), Green growth opportunities for SMEs.

http://www.europarl.europa.eu/doceo/document/A-8-2015-0135 EN.pdf; European Parliament (2013), Living

Well, within the Limits of our Planet, 7th EAP – The New General Union Environment Action Programme to 2020. https://ec.europa.eu/environment/pubs/pdf/factsheets/7eap/en.pdf; European Parliament (2012), European

Parliament resolution of 24 May 2012 on a Resource-efficient Europe (2011/2068(INI)). http://www.europarl.europa.eu/sides/getDoc.do?pubRef=-//EP//TEXT+TA+P7-TA-2012-0223+0+DOC+XML+V0//EN;

⁹⁷ European Commission (2018), Recommendation for a Council Recommendation on the economic policy of the euro area. SWD(2018) 467 final. https://eur-lex.europa.eu/legal-

<u>content/EN/TXT/?qid=1547734022733&uri=CELEX%3A52018DC0759</u>; European Commission (2018), A Clean Planet for all A European strategic long-term vision for a prosperous, modern, competitive and climate neutral economy. <u>https://ec.europa.eu/clima/sites/clima/files/docs/pages/com_2018_733_en.pdf</u>; European Commission (2017), European Semester Thematic Factsheet Taxation.

<u>https://ec.europa.eu/info/sites/info/files/file_import/european-semester_thematic-factsheet_taxation_en_1.pdf;</u> European Commission (2016), European Semester Thematic Factsheet

Taxation. <u>https://ec.europa.eu/info/sites/info/files/european-semester_thematic-factsheet_taxation_en.pdf;</u> European Commission (2015), Smart Taxation: A Winning Strategy. Video. https://www.youtube.com/watch?v=8kGgpaC6ID8

⁹⁸ WBCSD (World Business Council for Sustainable Development) (2010), Vision 2050: The New Agenda for Business. https://www.wbcsd.org/contentwbc/download/1746/21728

⁹⁹ Ellen MacArthur Foundation, McKinsey and SUN (2015), Growth Within: A Circular Economy Vision for a Competitive Europe.

https://www.ellenmacarthurfoundation.org/assets/downloads/publications/EllenMacArthurFoundation Growth-Within_July15.pdf

¹⁰⁰ BSDC (Business and Sustainable Development Commission (2017), Better Business, Better World: Executive Summary. <u>http://report.businesscommission.org/uploads/Executive-Summary.pdf</u>

¹⁰¹ European Commission (2015), Smart Taxation: A Winning Strategy. Video. https://www.youtube.com/watch?v=8kGgpaC6ID8

¹⁰² United Nations Secretary-General António Guterres (June 30, 2019), Secretary-General's remarks to Climate Summit Preparatory Meeting. Abu Dhabi. <u>https://www.un.org/sg/en/content/sg/statement/2019-06-30/secretary-generals-remarks-climate-summit-preparatory-meeting</u>. See also: Reuters (2019), Tax pollution, not people, says UN chief on Pacific visit. Video. <u>https://uk.reuters.com/video/2019/05/13/tax-pollution-not-people-says-un-chiefo?rpc=401&videoId=549359664&feedType=VideoRSS&feedName=LatestVideosUK&rpc=401&videoChannel=11 7763</u>

¹⁰³ IPCC (2018), Summary for Policymakers of IPCC Special Report on Global Warming of 1.5°C Approved by Governments, Chapter 2. <u>http://report.ipcc.ch/sr15/pdf/sr15_chapter2.pdf</u>

¹⁰⁴ OECD (2015), Towards Green Growth? Tracking Progress. <u>https://read.oecd-ilibrary.org/environment/towards-green-growth_9789264234437-en#page1</u>

¹⁰⁵ CPLC (2017), Report of the High-level Commission on Carbon Prices.

https://www.carbonpricingleadership.org/report-of-the-highlevel-commission-on-carbon-prices/

¹⁰⁶ European Commission (2015), Closing the loop - An EU action plan for the Circular Economy. <u>https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52015DC0614</u>

¹⁰⁷ Government of China (2013), Circular Economy Development Strategies and Action Plan. (Available in Chinese) <u>http://www.gov.cn/zwgk/2013-02/05/content_2327562.htm</u>

¹⁰⁸ Federal Ministry for the Environment, Nature Conservation, Building and Nuclear Safety (2016), German Resource Efficiency Programme II: Programme for the sustainable use and conservation of natural resources.
 <u>https://circulareconomy.europa.eu/platform/sites/default/files/german_resource_efficiency_programme_ii_bf.pdf</u>
 ¹⁰⁹ Republic of France (2018), Circular Economy roadmap of France: 50 measures for a100% circular economy.
 <u>https://circulareconomy.europa.eu/platform/sites/default/files/frec_anglais.pdf</u>

¹¹⁰ The Scottish Government (2016), Making Things Last: a circular economy strategy for Scotland. https://circulareconomy.europa.eu/platform/sites/default/files/making_things_last.pdf

¹¹¹ Republic of Slovenia (2018), Roadmap towards the Circular Economy in Slovenia.

https://circulareconomy.europa.eu/platform/sites/default/files/roadmap_towards_the_circular_economy_in_sloveni a.pdf

¹¹² Ministry of Environment, Portugal (2017), Leading the transition [Action plan for circular economy in Portugal: 2017-2020]. <u>https://circulareconomy.europa.eu/platform/sites/default/files/strategy</u> -

portuguese action plan paec en version 3.pdf

¹¹³ Ministry for the Environment, Land and Sea Ministry of Economic Development (2017), Towards a Model of Circular Economy for Italy Overview and Strategic Framework.

<u>https://circulareconomy.europa.eu/platform/sites/default/files/strategy - towards a model eng completo.pdf</u> ¹¹⁴ Sitra (2016), Leading the cycle – Finnish road map to a circular economy 2016-2025.

https://circulareconomy.europa.eu/platform/sites/default/files/selvityksia121.pdf

¹¹⁵ Ministry of Environment & Energy of Greece (2018), National Circular Economy Strategy. https://circulareconomy.europa.eu/platform/sites/default/files/national_circular_economy_strategy.pdf ¹¹⁶ Government of The Netherlands (2016), A Circular Economy in the Netherlands by 2050.

https://www.government.nl/binaries/government/documents/policy-notes/2016/09/14/a-circular-economy-in-thenetherlands-by-2050/17037+Circulaire+Economie_EN.PDF

¹¹⁷ Including the circular strategies of the 28 countries of the European Union, the Netherlands, Portugal, Italy, Finland and Greece.

¹¹⁸ World Bank (2011), Issues in estimating the employment generated by energy sector activities.
 <u>http://siteresources.worldbank.org/EXTENERGY2/Resources/MeasuringEmploymentImpactofEnergySector.pdf</u>
 ¹¹⁹ ILO (2018), World Employment and Social Outlook 2018 - Greening with Jobs. <u>https://www.ilo.org/weso-greening/documents/WESO_Greening_EN_web2.pdf</u>

¹²⁰ CDP (June 4, 2019), World's biggest companies face \$1 trillion in climate change risks. https://www.cdp.net/en/articles/media/worlds-biggest-companies-face-1-trillion-in-climate-change-risks

¹²¹ TCFD (2018), 2018 Status Report; Task Force on Climate-related Financial Disclosures: Status Report. https://www.fsb-tcfd.org/wp-content/uploads/2018/08/FINAL-2018-TCFD-Status-Report-092518.pdf

¹²² Prince of Wales's Corporate Leaders' Group (2015), L'Appel de Paris: The Paris Pledge for Action. <u>https://www.corporateleadersgroup.com/about/international-leadership/paris-pledge-for-action</u>; https://www. wearestillin.com/.

¹²³ The Investor Agenda (September 19, 2019) Record 515 institutional investors managing \$35 trillion in assets urge governments to step up ambition to tackle climate change Global Investor Statement to Governments on Climate Change. Press release. <u>https://theinvestoragenda.org/wp-</u> content/uploads/2019/09/Press-release-19.09.pdf

¹²⁴ Hafner, A., Janoska, P., Lee, C. (2018), The Importance of Real World Policy Packages to Drive Energy Transitions. <u>https://www.iea.org/newsroom/news/2018/july/commentary-the-importance-of-real-world-policy-packages-to-drive-energy-</u>

transiti.html?utm_content=buffer380c9&utm_medium=social&utm_source=twitter.com&utm_campaign=buffer ¹²⁵ CDP (2018), Commit to Putting a Price on Carbon. <u>https://www.cdp.net/en/campaigns/commit-to-action/price-</u>

<u>on-carbon</u>

¹²⁶ Unilever (2018), Explainer: What is Carbon Pricing and Why is it Important?

https://www.unilever.com/news/news-and-features/Feature-article/2018/explainer-what-is-carbon-pricing-and-why-is-it-important.html

¹²⁷ Ahluwalia, M.B. (2017), Carbon Pricing: A Case for Transformative Climate Action.

https://www.carbonpricingleadership.org/blogs/2017/11/7/carbon-pricing-a-case-for-transformative-climate-action ¹²⁸ CDP (November 6, 2017), Global Water Report: Three-fold Rise in Companies Excelling in Water

Management. <u>https://www.cdp.net/ en/articles/media/global-water-report-three-fold-rise-in-companies-excelling-in-water-management</u>

¹²⁹ H&M Group (April 2017), Sustainability Report 2016.

https://sustainability.hm.com/content/dam/hm/about/documents/en/CSR/2016%20Sustainability%20report/HM_gr oup_SustainabilityReport_2016_FullReport_en.pdf

¹³⁰ Kering (2019), Sustainability; Care. <u>https://www.kering.com/en/sustainability/our-strategy/care/</u>

¹³¹ IKEA (2018), IKEA takes sustainable living to a new level, with new commitments to become people and planet positive by 2030. <u>https://www.ikea.com/us/en/about_ikea/newsitem/060718-IKEA-commits-to-become-people-planet-positive-2030</u>

¹³² Nestlé (2018), Coffee. <u>https://www.nestle.com/csv/raw-materials/coffee</u>

¹³³ "Waste heat from a neighbouring glass factory helps run the plant's boilers. A water treatment plant (...) will purify water used in production and reuse 30% of it in other processes. As the water treatment plant runs, it will produce biogas, which can also be used for heat in the brewing process. The sludge from the plant will go to nearby farms to improve the soil, while spent grains will become cattle feed. Broken glass is recycled at the next-door glass factory." Peters, A. (2018), This Brewery is Designed as a Model for the Circular Economy. https://www.fastcompany.com/40536868/this-brewery-is-designed-as-a-model-for-the-circular-economy

¹³⁴ Chasan, E. (March 1, 2019), Tyson bets on omnivores with new alternative protein business.
 https://www.bloomberg.com/news/articles/2019-03-01/tyson-bets-on-omnivores-with-new-alternative-protein-

https://www.bloomberg.com/news/articles/2019-03-01/tyson-bets-on-omnivores-with-new-alternative-protection-business?utm_campaign=socialflow-

organic&utm_content=business&utm_medium=social&utm_source=twitter&cmpid=socialflow-twitter-business ¹³⁵ Total (2016), Total Launches Vast Program to install Solar Power in 5,000 Service Stations Worldwide. <u>https://www.total.com/en/media/news/press-releases/total-launches-vast-program-install-solar-power-5000-service-stations-worldwide</u>

¹³⁶ For more information see <u>http://www.butamax.com/</u>

¹³⁷ WindEurope (2018), World's First Offshore Wind Farm Without Subsidies to be Built in the Netherlands. <u>https://windeurope.org/newsroom/press-releases/worlds-first-offshore-wind-farm-without-subsidies-to-be-built-in-the-netherlands/</u>

¹³⁸ Clover, I. (December 2015), Tata Power Solar commissions world's largest rooftop solar array. <u>http://reneweconomy.com.au/2015/tata-power-solar-commissions-worlds-largest-rooftop-solar-array-27272</u> ¹³⁹ Teck (2011), Recycling E-waste Creates Value through Product Life Cycle Business Opportunities. <u>https://www.teck.com/news/stories/2011/recycling-e-waste-creates-value-through-product-life-cycle-business-opportunities</u>

¹⁴⁰ Tata Steel (2018), Innovative HIsarna technology exceeds expectations in sustainable steel production. <u>https://www.tatasteeleurope.com/sv/nyheter/nyheter/hisarna%E2%80%93tech%E2%80%93exceeds%E2%80%9</u> <u>3expectations</u>

¹⁴¹ Duurzaam Bedrijfsleven (2018), Vattenfall bouwt proeffabriek voor staal op basis van waterstof. <u>https://www.duurzaambedrijfsleven.nl/industrie/29111/vattenfall-bouwt-proeffabriek-voor-staal-op-basis-van-waterstof?utm_source=nieuwsbrief&utm_medium=e-</u>

mail&utm_campaign=Daily%20Focus%2026%20Juni&usertoken=1431506968J9HFg9JmiDxU0bxr5amOcub2pZunGNkkhcl9uz5fpwdcfpUddb6aXP4GRyrnfuJA

¹⁴² Ellen MacArthur Foundation (2017), Conserving Materials for the Next Mobility Revolution. https://www.ellenmacarthurfoundation.org/case-studies/avoiding-3-million-tonnes-of-waste

¹⁴³ Springwise (August 2015), Steel plant will convert harmful byproducts into renewable energy.

http://www.springwise.com/steel-plant-wil-convert-harmful-gases-renewable-energy/

¹⁴⁴ Lance Anderson, J.R. (August 2018), Canadian gold mine cleans up its act with electricity. <u>https://www.greencarreports.com/news/1118187_canadian-gold-mine-cleans-up-its-act-with-electricity</u>

¹⁴⁵ CNN Business (February 20, 2019), Glencore will cap coal production, but some climate groups say that isn't good enough. <u>https://edition.cnn.com/2019/02/20/business/glencore-coal-cap-climate/index.html</u>

¹⁴⁶ Hemp can produce two to three times more fiber than an acre of cotton; can detoxify soil by removing harmful chemicals and pollutants while enriching the soil with nitrogen and oxygen; and is a more durable material, making products last longer. Hodge, K. (April 2019), Levi's "Cottonized" Hemp Is Ready to End Cotton's Reign. <u>https://www.highsnobiety.com/p/levis-hemp-sustainable-fabric/;</u> Levi (March 2019), Levi's Wellthread X Outerknown Present_Cottonized Hemp. <u>https://www.levi.com/US/en_US/blog/article/levis-wellthread-x-outerknown-present-cottonized-hemp/</u>

¹⁴⁷ Kharpal, A. (2018), Adidas Sold 1m Shoes Made out of Ocean Plastic in 2017. <u>https://www-cnbc-com.cdn.ampproject.org/c/s/www.cnbc.com/amp/2018/03/14/adidas-sold-1-million-shoes-made-out-of-ocean-plastic-in-2017.html</u>

¹⁴⁸ Wilson, M. (April 17, 2019), <u>Exclusive: Adidas's radical new shoe could change how the world buys sneakers</u>. <u>https://www.fastcompany.com/90335038/exclusive-adidass-radical-new-shoe-could-change-how-the-world-buys-sneakers</u>

¹⁴⁹ Peters, A. (June 6, 2018), North Face is cutting waste by selling refurbished old coats.

https://www.fastcompany.com/40581355/north-face-is-cutting-waste-by-selling-refurbished-old-coats

¹⁵⁰ Kapfunde, M. (August 15, 2018), Reebok's (Nearly) Biodegradable "Cotton + Corn" Trainers Available This Autumn 2018. <u>https://fashnerd.com/2018/08/reebok-biodegradable-trainers-sneaker-tech/</u>

¹⁵¹ Italie, P. (April 18, 2019), Earth, meet Polo: Ralph Lauren unveils plastic bottle shirt.

https://www.apnews.com/4f10afd9933c43059fd7cabc23705ef6

¹⁵² Reuters (April 5, 2019), Fashion backwards? H&M to trial sales of vintage garments.

https://www.reuters.com/article/us-hennes-mauritz-environment/fashion-backwards-hm-to-trial-sales-of-vintage-garments-idUSKCN1RH1PN

¹⁵³ Ellen MacArthur Foundation (2017), A Wardrobe in the Cloud. <u>https://www.ellenmacarthurfoundation.org/case-</u> studies/a-closet-in-the-cloud

¹⁵⁴ Pownall, A. (February 20, 2019), IKEA to begin renting furniture as part of wider sustainable push. <u>https://www.dezeen.com/2019/02/20/ikea-rental-furniture-circular-economy-design/</u>

¹⁵⁵ IKEA Group (2014), Sustainability Report FY14.

https://www.ikea.com/ms/en_US/pdf/sustainability_report/sustainability_report_2014.pdf ¹⁵⁶ Net-Works (2019), <u>http://net-works.com/</u>

¹⁵⁷ Telegraph.co.uk (September 7, 2018), Carlsberg glues beer cans together becoming one of the first breweries to abandon plastic rings. <u>https://www.telegraph.co.uk/news/2018/09/07/carlsberg-glues-beer-cans-together-becoming-first-brewery-abandon/amp/</u>

¹⁵⁸ Unilever (2015), Annual Report and Accounts 2015.

https://www.unilever.com/Images/annual_report_and_accounts_ar15_tcm244-478426_en.pdf

¹⁵⁹ New Plastics Economy and Ellen MacArthur Foundation (2018), Leading Brands, Retailers, and Packaging Companies Work towards 100% Reusable, Recyclable or Compostable Packaging by 2025 or Earlier. <u>https://newplasticseconomy.org/news/11-companies-commit-to-100-reusable-recyclable-or-compostable-</u>

packaging-by-

<u>2025?utm_content=buffera1fbe&utm_medium=social&utm_source=twitter.com&utm_campaign=buffer</u> ¹⁶⁰ Apple (2018), Apple Now Globally Powered by 100 percent Renewable Energy.

https://www.apple.com/uk/newsroom/2018/04/apple-now-globally-powered-by-100-percent-renewable-energy/ ¹⁶¹ Deutsche Post DHL Group (2017), Deutsche Post Expands StreetScooter Production.

https://www.dpdhl.com/en/media-relations/press-releases/2017/deutsche-post-expands-streetscooterproduction.html ¹⁶² Central Japan Railway Company (2017), Annual Report 2017. <u>https://global.jr-</u>central.co.jp/en/company/ir/annualreport/_pdf/annualreport2017.pdf

¹⁶³ Ellen MacArthur Foundation (2017d), Using Product Passports to Improve the Recovery and Reuse of Shipping Steel. <u>https://www.ellenmacarthurfoundation.org/case-studies/using-product-passports-to-improve-the-recovery-and-reuse-of-shipping-steel</u>

¹⁶⁴ Springwise (August 2015), Sustainable jet fuel is made from household garbage.

https://www.springwise.com/sustainable-jet-fuel-household-garbage/

¹⁶⁵ Spector, J. (December 11, 2015), The Dirty Industries COP21 Ignores.

https://newrepublic.com/article/125664/dirty-industries-cop21-

ignores?utm_content=bufferd69bd&utm_medium=social&utm_source=twitter.com&utm_campaign=buffer ¹⁶⁶Weir, B. (May 2019), A significant day for aviation': Qantas launches zero garbage flight.

https://www.smh.com.au/business/companies/a-significant-day-for-aviation-qantas-launches-zero-garbage-flight-20190508-p51l6f.html

¹⁶⁷ Geuss, M. (April 25, 2019), Daimler North America CEO says future "does not include plug-in hybrids". https://arstechnica.com/cars/2019/04/daimler-north-america-ceo-says-future-does-not-include-plug-in-hybrids/

¹⁶⁸ BBC.com (December 11, 2015), Ford to invest \$4.5bn in electric cars. <u>https://www.bbc.com/news/business-</u> 35069582

¹⁶⁹ BMW Group (2016), Battery Storage System Electrified by BMW Announced at EVS29 in Montreal. <u>https://www.press.bmwgroup.com/canada/article/attachment/T0261315EN/361853</u>

¹⁷⁰ Rosengren, C. (2018) Volvo unveils another electric refuse collection vehicle.

https://www.wastedive.com/news/volvo-launch-fully-electric-collection-vehicle-europe/521611/

¹⁷¹ Matousek, Mark (September 5, 2019), Porsche just revealed the version of its stunning \$153,510 electric Tesla rival that you can actually buy — here's what it looks like. Business Insider.

https://www.businessinsider.nl/porsche-reveals-taycan-turbo-and-taycan-turbo-s-production-model-2019-9?international=true&r=US

¹⁷² Italcementi Group (2014), Annual Report 2014.

http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.728.9559&rep=rep1&type=pdf ¹⁷³ Aether Cement (2018), http://www.aether-cement.eu/

¹⁷⁴ AkzoNobel (2018), Getting Ready for Innovation's Third Era. <u>https://www.akzonobel.com/en/for-media/media-</u>releases-and-features/getting-ready-innovations-third-era

¹⁷⁵ ABN AMRO (n.d.), ABN AMRO Boosts Circular Economy. <u>https://www.abnamro.com/en/about-abnamro/in-</u>society/sustainability/our-focal-points/circular-economy/index.html

¹⁷⁶ SFC (Sustainable Finance Collective) Asia (2018), About SFC Asia. <u>http://www.sfc-asia.com/funding-panel</u> ¹⁷⁷ Phillips (January 24, 2008), Philips spearheads the Circular Economy with firm 2020 pledge.

https://www.philips.com/a-w/about/news/archive/standard/news/articles/2018/20180124-philips-spearheads-thecircular-economy-with-firm-2020-pledge.html

¹⁷⁸ Cuff, M. (September 2015), Starbucks, Nike and Walmart commit to sourcing 100% renewable electricity. <u>https://www.theguardian.com/environment/2015/sep/23/starbucks-nike-walmart-commit-sourcing-100-renewable-electricity</u>

¹⁷⁹ Value added tax (VAT) is also included in the tool. Primarily, VAT is a consumption tax, not an environmental tax. Rate differentiation does affect supply and demand, which in turn influences consumer decisions on specific products and services.

¹⁸⁰ OECD (2018), Global Material Resources Outlook to 2060; Economic drivers and environmental consequences; Highlights. <u>http://www.oecd.org/environment/waste/highlights-global-material-resources-outlook-to-2060.pdf</u>

¹⁸¹ Diffenbaugh, N. S., Burke, M. (2019), Global warming has increased global economic inequality. <u>https://www.pnas.org/content/pnas/early/2019/04/16/1816020116.full.pdf</u>

¹⁸² As suggested by Huq (2019), such climate damages payments could, for example, take the form of a climate damages tax on fossil fuel companies or an international air passenger levy to raise funding for adaptation measures. A levy of \$10 on each international passenger in economy class (and \$50 on business class), for example, could raise over \$10 billion a year to the Adaptation Fund under the UNFCCC. Huq, Saleemul (May 8, 2019), Making polluters pay for loss and damage. The Daily Star. https://www.thedailystar.net/opinion/politics-climate-change/news/making-polluters-pay-loss-and-damage-1740100

¹⁸³ For an overview of the discussion, see for example: González, J.F. (2018), Environmental taxation and the double dividend hypothesis in CGE modelling literature: A critical review. Journal of Policy Modeling. Volume 40, Issue 1, January–February 2018, Pages 194-223.

https://www.sciencedirect.com/science/article/pii/S0161893817301205?via=ihub

¹⁸⁴ A few examples:

- OECD research found that if revenues from green taxes are used to lower labour taxes, employment gains of nearly 2 per cent were possible by 2030, as compared to a business-as-usual scenario. Château, Saint-Martin and Manfredi (2011) in Montt, G., Fraga, F., Harsdorff, M. (2018), The future of work in a changing natural environment: Climate change, degradation and sustainability.

ILO. <u>https://www.ilo.org/wcmsp5/groups/public/---dgreports/---</u> cabinet/documents/publication/wcms 644145.pdf;

- A 2016 study found that shifting taxes from labour and towards natural resources and consumption in 27 Member States of the European Union over the 2016-2020 period would add €842 billion in GDP, enable 6.6 million more people to be in employment, cut carbon emissions by 8.2%, all by 2020, and save €27.7 billion on the energy import bill over the five-year period. The Ex'tax Project, Deloitte, EY, KPMG, Meijburg and PwC (2016), New Era. New plan. Europe. A Fiscal Strategy for an Inclusive, Circular Economy. <u>http://www.neweranewplan.com/</u>

Several meta-studies looked at series of simulations:

- A meta-study of 69 different simulations from 40 studies using General Equilibrium models showed 55% of simulations achieved both environmental and economic dividends. 90% of the studies that reduced social security contributions achieved such double dividend. The majority of simulations that recycled via labor incomes, capital taxes and other taxes achieved a double dividend. Recycled revenues towards lump-sum transfers were least successful in achieving a double dividend across these studies. González, J.F. (2018), Environmental taxation and the double dividend hypothesis in CGE modelling literature: A critical review. Journal of Policy Modeling. Volume 40, Issue 1, January–February 2018, Pages 194-223. https://www.sciencedirect.com/science/article/pii/S0161893817301205?via=ihub
- Gago, Labandeira, and López-Otero (2014) reviewed 699 simulations, using different models and methodologies, from 100 papers on the effects of different taxes that have in common that they are applied to one or more energy products, in most cases with an environmental rationale and part of wider tax reform schemes. 95% of the simulations reported decreases in emissions with respect to the business-as-usual scenario. The effects, on average, were positive on GDP, employment and consumer prices, and negative on energy demand, energy prices and welfare. Gago, Alberto, Labandeira, Xavier, Lopez-Otero, Xiral (2014), A Panorama on Energy Taxes and Green Tax Reforms. Revista Hacienda Pública Española. 208. 145-190. 10.7866/HPE-RPE.14.1.5.

https://www.researchgate.net/publication/284350683 A Panorama on Energy Taxes amd Green Tax Re forms

Patuelli, Nijkamp and Pels (2005) looked at a total of 186 model simulations from 61 separate studies. On average, all of the different groupings of studies predicted net job creation with significant reductions in CO₂ emissions. Patuelli, Roberto, Nijkamp, Peter, Pels, Eric (2005), Environmental tax reform and the double dividend: A meta-analytical performance assessment. Ecological Economics, Elsevier, vol. 55(4), pages 564-583, December, cited in Aarhus University, Eunomia (2015), Study on Environmental Fiscal Reform Potential in 14 EU Member States: Appendices. Final Report to DG Environment of the European Commission. https://ieep.eu/uploads/articles/attachments/51df7de3-0b82-4771-83d7-

85ee0a348a17/Eunomia_EFR_Final_Report_APPENDICES_V0.1.pdf?v=63664509889.

¹⁸⁵ Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. <u>https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe</u> <u>d=y</u>

¹⁸⁶ Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. <u>https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe</u> <u>d=v</u>

¹⁸⁷ Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. <u>https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowed=y</u>

¹⁸⁸ Dechezleprêtre, A., D. Nachtigall and F. Venmans (2018), The joint impact of the European Union emissions trading system on carbon emissions and economic performance, OECD Economics Department Working Papers, No. 1515. <u>http://dx.doi.org/10.1787/4819b016-en</u>

¹⁸⁹ Sweden (initial year of the reforms: 1991), Denmark (1993), the Netherlands (1996), Finland (1997), Slovenia (1997), Germany (1999) and the UK (2001). Andersen, M. S., Speck, S.; Ekins, P.; Scott, S.; Fitz Gerald, J. (2007), Competitiveness Effects of Environmental Tax Reforms (COMETR). Final Report to the European Commission. <u>http://pure.au.dk/portal/files/128999763/COMETR_Summaryreport.pdf</u>

¹⁹⁰ OECD, World Bank, UN Environment (2018), Financing Climate Futures: Rethinking Infrastructure. <u>https://doi.org/10.1787/9789264308114-en</u>

¹⁹¹ Fay, M., Hallegatte, S., Vogt-Schilb, A., Roszenberg, J., Narloch, U. and Kerr, T. (2015), Decarbonising Development. Three Steps to a Zero-Carbon Future. The World Bank. https://www.worldbank.org/content/dam/Worldbank/document/Climate/dd/decarbonizing-development-report.pdf ¹⁹² Scientific American (December 1, 2015), A Tax on Carbon Pollution Can Benefit Business. Low Oil and Gas Prices Make this the Right Time to Tax Fossil Fuels. <u>https://www.scientificamerican.com/article/a-tax-on-carbon-pollution-can-benefit-business/?redirect=1</u>

¹⁹³ Beck et al. (2015) in Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank.

https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe d=y

¹⁹⁴ World Bank Group (2019), State and Trends of Carbon Pricing 2019. <u>https://openknowledge.worldbank.org/handle/10986/31755</u>

¹⁹⁵ Rafaty, R. (2018), Perceptions of Corruption, Political Distrust, and the Weakening of Climate Policy. <u>https://www.mitpressjournals.org/doi/abs/10.1162/glep_a_00471</u>

¹⁹⁶ Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe

¹⁹⁷ See for example: Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank.

https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe d=y; Funke, Franziska and Mattauch, Linus (August 10, 2018), Why is carbon pricing in some countries more successful than in others? Our World in Data. Guest blog. <u>https://ourworldindata.org/carbon-pricing-popular</u>; Ahmed, Sadiq (2018), Environmental Fiscal Reforms in Bangladesh.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPape%2B7%2BEnviron mental%2BFiscal%2BReforms%2Bin%2BBangladesh.pdf; World Bank (2018), Policy Note on: Options for a carbon tax in Bangladesh. http://documents.worldbank.org/curated/en/721381535709573696/pdf/Carbon-Taxation-in-Bangladesh.pdf

¹⁹⁸ Partnership for Market Readiness (PMR), Carbon Pricing Leadership Coalition (CPLC), 2018. Guide to Communicating Carbon Pricing. <u>http://documents.worldbank.org/curated/en/668481543351717355/pdf/132534-WP-WBFINALonline.pdf</u>

¹⁹⁹ Inchauste, Gabriela, Lustig, Nora (Eds.) (2017), The Distributional Impact of Taxes and Transfers: Evidence from Eight Developing Countries. World Bank. <u>https://www.worldbank.org/en/topic/poverty/publication/the-distributional-impact-of-taxes-and-transfers-evidence-from-eight-developing-countries?CID=POV_TT_Poverty_EN_EXT?cid=EXT_WBSocialShare_EXT</u>

²⁰⁰ Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank. https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe d=v

²⁰¹ Mirrlees, J. (et al.) (2011), Tax by Design. <u>https://www.ifs.org.uk/docs/taxbydesign.pdf</u>

²⁰² Dartanto (2013); Cooke et al. (2016). In Pigato, Miria A. (Ed.) (2019), Fiscal Policies for Development and Climate Action. The World Bank.

https://openknowledge.worldbank.org/bitstream/handle/10986/31051/9781464813580.pdf?sequence=4&isAllowe d=y

²⁰³ Mani, Muthukumara, Bandyopadhyay, Sushenjit, Chonabayashi, Shun, Markandya, Anil, Mosier, Thomas (2018), South Asia's Hotspots: Impacts of Temperature and Precipitation Changes on Living Standards. South Asia Development Matters. World Bank. <u>https://openknowledge.worldbank.org/handle/10986/28723</u>

²⁰⁴ UNCTAD (December 2018), Leapfrogging: Look before you leap. Policy Brief based on UNCTAD, 2018, Technology and Innovation Report 2018: Harnessing Frontier Technologies for Sustainable Development. <u>https://unctad.org/en/PublicationsLibrary/presspb2018d8_en.pdf</u>

²⁰⁵ Pilling, David (August 13, 2018), African economy: the limits of 'leapfrogging. The Financial Times. <u>https://www.ft.com/content/052b0a34-9b1b-11e8-9702-5946bae86e6d</u>

²⁰⁶ Demirgüç-Kunt, Asli, Leora Klapper, Dorothe Singer, Saniya Ansar, and Jake Hess (2018), The Global Findex Database 2017: Measuring Financial Inclusion and the Fin- tech Revolution. World Bank. https://globalfindex.worldbank.org/

²⁰⁷ Lu, Lu (April 4, 2018), How Did Shenzhen, China Build World's Largest Electric Bus Fleet? World Resources Institute website. <u>https://www.wri.org/blog/2018/04/how-did-shenzhen-china-build-world-s-largest-electric-bus-fleet</u>

²⁰⁸ Pilling, David (August 13, 2018), African economy: the limits of 'leapfrogging. The Financial Times. <u>https://www.ft.com/content/052b0a34-9b1b-11e8-9702-5946bae86e6d</u>

²⁰⁹ Hallegatte, Stephane, Mook Bangalore, Laura Bonzanigo (et al.) (2016), Shock Waves: Managing the Impacts of Climate Change on Poverty. World Bank.

https://openknowledge.worldbank.org/bitstream/handle/10986/22787/9781464806735.pdf

²¹⁰ Nicholls, R.J. (et al.) (Eds.) (2018), Ecosystem Services for Well-Being in Deltas. <u>https://doi.org/10.1007/978-</u> <u>3-319-71093-8_4</u>

²¹¹ United Nations (2017), World Population Prospects: The 2017 Revision. Accessed May 2019. https://population.un.org/wpp/Download/Standard/Population/ ²¹² The World Bank Group (Accessed May 2019), Population, total.

https://data.worldbank.org/indicator/SP.POP.TOTL?end=2017&locations=BD&start=2017&view=bar

²¹³ United Nations (2017), World Population Prospects: The 2017 Revision. Accessed May 2019. <u>https://population.un.org/wpp/Download/Standard/Population/</u>

²¹⁴ FAO (2011), Ganges-Brahmaputra-Meghna Basin. Accessed May 2019.

http://www.fao.org/nr/water/aguastat/basins/gbm/index.stm

²¹⁵ FAO (2011), Ganges-Brahmaputra-Meghna Basin, Accessed May 2019.

http://www.fao.org/nr/water/aquastat/basins/gbm/index.stm

²¹⁶ World Bank (July 25, 2018), Bangladesh Disaster Risk and Climate Resilience Program. Brief. Accessed May 2019.

https://www.worldbank.org/en/country/bangladesh/brief/bangladesh-disaster-risk-climate-change-program

²¹⁷ Internal Displacement Monitoring Centre (Accessed March 13, 2019), Bangladesh.

http://www.internal-displacement.org/countries/bangladesh

²¹⁸ Dasgupta (et al.) (2011), Cyclones in a Changing Climate: The Case of Bangladesh.

https://www2.gwu.edu/~iiep/signatureinitiatives/adaptation/docs/Dasgupta,%20Cyclones%20in%20a%20Changin g%20Climate-The%20Case%20of%20Bangladesh%20(updated).pdf

²¹⁹ See: Ban Ki-moon and Sheikh Hasina (July 13, 2019) Bangladesh leaps ahead in preparing for the worst of climate change. Here is what we can learn. <u>https://www.newsweek.com/bangladesh-leads-way-prepare-climate-change-1447410</u>

²²⁰ World Bank (July 25, 2018), Bangladesh Disaster Risk and Climate Resilience Program.

https://www.worldbank.org/en/country/bangladesh/brief/bangladesh-disaster-risk-climate-change-program ²²¹ Center for Environmental and Geographic Information Services (CEGIS) (2013), Vulnerability to Climate Induced Drought. Scenario & Impacts. Study Report. Saha, Sanjib Kumar, Ali, Mirza Swawkat, Shahid, Shaila (Eds.). <u>https://www.undp.org/content/dam/bangladesh/docs/Publications/Pub2016/Study%20Report%20-Vulnerability%20to%20Climate%20Induced%20Drought%20Scenario%20and%20Impacts%20-%202013.pdf</u> ²²² World Bank (July 25, 2018), Bangladesh Disaster Risk and Climate Resilience Program.

https://www.worldbank.org/en/country/bangladesh/brief/bangladesh-disaster-risk-climate-change-program²²³ FAO (2011), Ganges-Brahmaputra-Meghna Basin, Accessed May 2019.

http://www.fao.org/nr/water/aquastat/basins/gbm/index.stm

²²⁴ Ministry of Environment and Forests (MOEF) Government of the People's Republic of Bangladesh (September 2015), Intended Nationally Determined Contributions (INDC).

https://www4.unfccc.int/sites/ndcstaging/PublishedDocuments/Bangladesh%20First/INDC 2015 of Bangladesh. pdf

²²⁵ World Economic Forum (April 16, 2019), Online Tweet.

https://twitter.com/wef/status/1117970886513106944/photo/1

²²⁶ IOM (December 2016), Assessing the Climate Change Environmental Degradation and Migration Nexus in South Asia.

http://publications.iom.int/system/files/pdf/environmental_degradation_nexus_in_south_asia.pdf ²²⁷ OECD (2017), OECD Development Co-operation Peer Reviews: The Netherlands 2017. http://dx.doi.org/10.1787/9789264278363-en

²²⁸ Ahsan, Ahmad, Sattar, Zaidi (2019), State of the economy 2018: Managing vulnerabilities, creating opportunities to sustain growth. Policy Insights January 2019. 'Bangladesh Reform Priorities'. Policy Research Institute of Bangladesh. P.14-21. <u>https://www.pri-bd.org/wp-content/uploads/2019/05/Policy-Inshight-Issue-</u>
 <u>4 January-2019.pdf</u>

²²⁹ Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018. <u>https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4</u> <u>7/Ch-01%20%28English-2018%29.pdf</u>

²³⁰ World Bank, World Development Indicators. (2019). GDP (current US\$) [Data file]. Retrieved from https://data.worldbank.org/indicator/NY.GDP.MKTP.CD?locations=BD

²³¹ CIA (2019), The World Factbook; Bangladesh. <u>https://www.cia.gov/library/Publications/the-world-factbook/geos/print_bg.html</u>

²³² Obe, M. (November 2018), Bangladesh fights for future of its garment industry.

<u>https://asia.nikkei.com/Business/Business-trends/Bangladesh-fights-for-future-of-its-garment-industry</u> ²³³ Central Intelligence Agency (2019), The World Factbook; Bangladesh.

https://www.cia.gov/library/Publications/the-world-factbook/geos/print_bg.html

²³⁴ OECD (2017), Peer review; The Netherlands 2017. <u>http://dx.doi.org/10.1787/9789264278363-en</u>

²³⁵ Ahmed, Sadiq (2018), Environmental Fiscal Reforms in Bangladesh. Policy Research Institute. Economic Dialogue on Green Growth, UK Aid from the British People.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPape%2B7%2BEnviron mental%2BFiscal%2BReforms%2Bin%2BBangladesh.pdf

²³⁶ The 'upper poverty line' of 2,268 BDT per month in 2016 (or \$28.8, exchange rate Jan 1, 2016). Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018.

https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4 7/Ch-01%20%28English-2018%29.pdf

²³⁷ World Bank (October 11, 2018), Bangladesh: Access to Clean Water Will Reduce Poverty Faster. Press release.

https://www.worldbank.org/en/news/press-release/2018/10/11/bangladesh-access-to-clean-water-will-reduce-poverty-faster

²³⁸ ILO (Accessed April 2019), ILOSTAT database. https://www.ilo.org/ilostat/

²³⁹ ILO (2017), The ILO in Bangladesh. <u>https://www.ilo.org/wcmsp5/groups/public/@asia/@ro-bangkok/@ilo-dhaka/documents/publication/wcms_226527.pdf</u>

²⁴⁰ CIA (2019), The World Factbook; Bangladesh. <u>https://www.cia.gov/library/Publications/the-world-</u>factbook/geos/print_bg.html

²⁴¹ Year 2017/18. Bangladesh Garment Manufacturers and Exporters Association (BGMEA) (2019), Trade Information. <u>http://www.bgmea.com.bd/home/pages/tradeinformation</u>

²⁴² ILO (Accessed April 2019), ILOSTAT database. https://www.ilo.org/ilostat/

²⁴³ "Own-account workers and contributing family workers have a lower likelihood of having formal work arrangements, and are therefore more likely to lack elements associated with decent employment, such as adequate social security and a voice at work. Therefore, the two statuses are summed to create a classification of 'vulnerable employment'." International Labour Organization, ILOSTAT database. Data retrieved in April 2019. https://www.ilo.org/ilostat/.

²⁴⁴ CIA (2019), The World Factbook; Bangladesh. <u>https://www.cia.gov/library/Publications/the-world-factbook/geos/print_bg.html</u>

²⁴⁵ ILO, ILOSTAT database. Data retrieved in April 2019. <u>https://www.ilo.org/ilostat/</u>

²⁴⁶ CIA (2019), The World Factbook; Bangladesh. <u>https://www.cia.gov/library/Publications/the-world-</u>factbook/geos/print_bg.html

²⁴⁷ ILO.stat (Accessed 29/05/2019), Hours of Work.

²⁴⁸ BBS, ILO (2015), Child Labour Survey

2013. https://www.ilo.org/ipec/Informationresources/WCMS_IPEC_PUB_28175/lang--en/index.htm

²⁴⁹ World Bank (November 15, 2018), Bangladesh: Reducing Poverty and Sharing Prosperity. Result Briefs. <u>https://www.worldbank.org/en/results/2018/11/15/bangladesh-reducing-poverty-and-sharing-prosperity</u>

²⁵⁰ Urban access to electricity: 93%; Rural 73%. IEA/OECD (2018), World Energy Outlook 2018. <u>https://www.iea.org/sdg/electricity/</u>

²⁵¹ Planning Commission (2012), Perspective Plan of Bangladesh 2010-2021: Making Vision 2021 a Reality. Government of Bangladesh.

https://bangladesh.gov.bd/sites/default/files/files/bangladesh.gov.bd/page/6dca6a2a_9857_4656_bce6_139584b 7f160/Perspective-Plan-of-Bangladesh.pdf

²⁵² Ministry of Environment and Forests (MOEF) Government of the People's Republic of Bangladesh (September 2015), Intended Nationally Determined Contributions (INDC).

https://www4.unfccc.int/sites/ndcstaging/PublishedDocuments/Bangladesh%20First/INDC 2015 of Bangladesh. pdf

²⁵³ Infrastructure Development Company Limited (IDCOL) (Accessed April 2019), website.

http://idcol.org/home/solar

²⁵⁴ The ranking is based on 10 areas in the life cycle of a business: starting a business, dealing with construction permits, getting electricity, registering property, getting credit, protecting minority investors, paying taxes, trading across borders, enforcing contracts and resolving insolvency. World Bank (2018), Doing business 2019. Training for reform. Economy profile Bangladesh.

http://www.doingbusiness.org/content/dam/doingBusiness/country/b/bangladesh/BGD.pdf

²⁵⁵ "The *Executive Opinion Survey* was conducted in 2018 between January and June, and the survey's risk-related question received 12,548 responses. The respondents were presented with our core list of 30 global risks (...) and asked to select "the five global risks that you believe to be of most concern for doing business in your country within the next ten years"." World Economic Forum (WEF) (2018), Regional Risks for Doing Business 2018. Insight Report. <u>http://www3.weforum.org/docs/WEF_Regional_Risks_Doing_Business_report_2018.pdf</u>
²⁵⁶ OECD (2017), Peer review; The Netherlands 2017. <u>http://dx.doi.org/10.1787/9789264278363-en</u>

²⁵⁷ Butler, Sarah (Jan 21, 2019), Why are wages so low for garment workers in Bangladesh? The Guardian.
 <u>https://www.theguardian.com/business/2019/jan/21/low-wages-garment-workers-bangladesh-analysis</u>
 ²⁵⁸ Van Elven, M. (September 14, 2018), Bangladesh raises minimum wage for garment workers.

https://fashionunited.uk/news/business/bangladesh-raises-minimum-wage-for-garment-workers/2018091438912 ²⁵⁹ Barrett, Paul, Baumann-Pauly, Dorothée (2019), Made in Ethiopia: Challenges in the Garment Industry's New Frontier. New York University Stern for Business and Human Rights.

https://issuu.com/nyusterncenterforbusinessandhumanri/docs/nyu_ethiopia_final_online?e=31640827/69644612 ²⁶⁰ Muntean, M., Guizzardi, D., Schaaf, (et al.) (2018), Fossil CO2 emissions of all world countries - 2018 Report. http://edgar.jrc.ec.europa.eu/overview.php?v=booklet2018 ²⁶¹ Ahmed, Sadiq, Khondker, Bazlul H. (April 2018), Towards a Carbon Tax in Bangladesh. Economic Dialogue on Green Growth, UK Aid from the British People.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPaper%2B9%2BToward s%2Ba%2BCarbon%2BTax%2Bin%2BBangladesh.pdf

²⁶² Up to 2016. Our World in Data (2017), Cumulative CO2 Emissions. Accessed May 7, 2019. https://ourworldindata.org/grapher/cumulative-co-emissions

²⁶³ Estimate by World Bank cited in Ministry of Environment and Forests (MOEF) Government of the People's Republic of Bangladesh (September 2015), Intended Nationally Determined Contributions (INDC).

https://www4.unfccc.int/sites/ndcstaging/PublishedDocuments/Bangladesh%20First/INDC_2015_of_Bangladesh.pdf

²⁶⁴ Ministry of Environment and Forests (MOEF) Government of the People's Republic of Bangladesh (September 2015), Intended Nationally Determined Contributions (INDC).

https://www4.unfccc.int/sites/ndcstaging/PublishedDocuments/Bangladesh%20First/INDC_2015_of_Bangladesh.pdf

²⁶⁵ The government of Bangladesh has allocated approximately \$390 million to the Bangladesh Climate Change Trust Fund (BCCTF) during the last seven fiscal years. The other trust fund is the Bangladesh Climate Change Resilience Fund (BCCRF), a multi donor trust fund, established in 2010. Up to September 2016, BCCRF provided \$146 million in funding. Ministry of Foreign Affairs of The Netherlands (April 2018), Climate Change

Profile Bangladesh. <u>https://www.government.nl/binaries/government/documents/publications/2019/02/05/climate-change-profiles/Bangladesh.pdf</u>

²⁶⁶ UNDP (Accessed April 2019), Bangladesh Climate Change Trust Fund.

http://www.bd.undp.org/content/bangladesh/en/home/operations/projects/environment_and_energy/inclusivebudgeting-and-financing-for-climate-resilience1/national-policies-and-strategies/bangladesh-climate-change-trustfund-.html

²⁶⁷ IEA/OECD (2018), World Energy Outlook 2018.

https://www.iea.org/statistics/?country=BANGLADESH&year=2016&category=Energy%20supply&indicator=TPE SbySource&mode=table&dataTable=BALANCES

²⁶⁸ Ahmed, Sadiq, Khondker, Bazlul H. (April 2018), Towards a Carbon Tax in Bangladesh. Economic Dialogue on Green Growth, UK Aid from the British People.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPaper%2B9%2BToward s%2Ba%2BCarbon%2BTax%2Bin%2BBangladesh.pdf

²⁶⁹ In the most recent WHO guidelines, 'polluting fuels' include biomass (wood, dung, crop residues and charcoal), coal (including coal dust and lignite) and kerosene. 'Clean fuels' include electricity, LPG, piped natural gas (PNG), biogas, solar and alcohol fuels.' Data for the years 2016 and 2011 respectively. World Health Organization (2016), Burning opportunity: clean household energy for health, sustainable development, and wellbeing of women and children.

https://apps.who.int/iris/bitstream/handle/10665/204717/9789241565233_eng.pdf;jsessionid=BFCCDDF59B09A8 150293FCC1D9117C0E?sequence=1

²⁷⁰ Health Effects Institute (2019), State of Global Air 2019.

https://www.stateofglobalair.org/sites/default/files/soga 2019 report.pdf

²⁷¹ The guideline for annual average PM2.5 concentration is set at 10 μg/m3. Health Effects Institute (2019), State of Global Air 2019. <u>https://www.stateofglobalair.org/sites/default/files/soga_2019_report.pdf</u>

²⁷² Health Effects Institute (2019), State of Global Air 2019. Special

Report. https://www.stateofglobalair.org/sites/default/files/soga_2019_report.pdf

²⁷³ Joseph, G., Haque, S.S. (et al.) (2018), Promising progress: a diagnostic of water supply, sanitation, hygiene, and poverty in Bangladesh: Main report (English). WASH Poverty Diagnostic. World Bank Group.

http://documents.worldbank.org/curated/en/152341520607754248/Main-report

²⁷⁴ 82 percent of households are exposed to concentrations of iron in drinking water above WHO standards. Mohiuddin, Ak. (2019), Chemical Contaminants and Pollutants in the Measurable Life of Dhaka City.

https://www.researchgate.net/publication/331482819_Chemical_Contaminants_and_Pollutants_in_the_Measurab le_Life_of_Dhaka_City; 41 percent of all improved water sources are contaminated with E. Coli bacteria, which suggests a high prevalence of faecal contamination. World Bank (October 11, 2018), Bangladesh: Access to Clean Water Will Reduce Poverty Faster. Press release.

https://www.worldbank.org/en/news/press-release/2018/10/11/bangladesh-access-to-clean-water-will-reduce-poverty-faster

²⁷⁵ BBS 2017 mentioned in: Ahmed, Sadiq (2018), Environmental Fiscal Reforms in Bangladesh. <u>http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPape%2B7%2BEnvironmental%2BFiscal%2BReforms%2Bin%2BBangladesh.pdf</u>

²⁷⁶ Joseph, G., Haque, S.S. (et al.) (2018), Promising progress: a diagnostic of water supply, sanitation, hygiene, and poverty in Bangladesh: Main report (English). WASH Poverty Diagnostic. World Bank Group. http://documents.worldbank.org/curated/en/152341520607754248/Main-report ²⁷⁷ World Bank (October 11, 2018), Bangladesh: Access to Clean Water Will Reduce Poverty Faster. Press release.

https://www.worldbank.org/en/news/press-release/2018/10/11/bangladesh-access-to-clean-water-will-reduce-poverty-faster

²⁷⁸ World Bank (October 11, 2018), Bangladesh: Access to Clean Water Will Reduce Poverty Faster. Press release.

https://www.worldbank.org/en/news/press-release/2018/10/11/bangladesh-access-to-clean-water-will-reduce-poverty-faster

²⁷⁹ UN Environment (April 25, 2018), From birth to ban: A history of the plastic shopping bag.

https://www.unenvironment.org/news-and-stories/story/birth-ban-history-plastic-shopping-bag

²⁸⁰ Mohiuddin, Ak. (2019), Chemical Contaminants and Pollutants in the Measurable Life of Dhaka City.

https://www.researchgate.net/publication/331482819_Chemical_Contaminants_and_Pollutants_in_the_Measurab le_Life_of_Dhaka_City

²⁸¹ Sakamoto, M., Ahmed, T., Begum, S., Huq, H. (2019), Water Pollution and the Textile Industry in Bangladesh: Flawed Corporate Practices or Restrictive Opportunities?

https://www.sciencedirect.com/science/article/pii/S2211464517301926?via%3Dihub

²⁸² Mohiuddin, Ak. (2019), Chemical Contaminants and Pollutants in the Measurable Life of Dhaka City.

https://www.researchgate.net/publication/331482819 Chemical Contaminants and Pollutants in the Measurab le Life of Dhaka City

²⁸³ UNHCR (Accessed May 2019), Rohingya emergency. Website.

²⁸⁴ Mukul, Sharif A., Huq, Saleemul (et al.) (April 12, 2019), Rohingya refugees and the environment. Science, Vol 364, Issue 6436. <u>https://science.sciencemag.org/content/364/6436/138</u>

12 April 2019: 138. <u>https://science.sciencemag.org/content/364/6436/138</u>; Irfanullah, Haseeb (May 22, 2019), Environmental conservation in a refugee crisis: when calling it a cross-cutting issue is not enough. Humanitarian Practice Network website. <u>https://odihpn.org/blog/environmental-conservation-refugee-crisis-calling-cross-cutting-issue-not-enough/</u>

²⁸⁵ World Bank (Accessed September 2019), Tax revenue (% of GDP). International Monetary Fund, Government Finance Statistics Yearbook and data files, and World Bank and OECD GDP estimates. <u>https://data.worldbank.org/indicator/GC.TAX.TOTL.GD.ZS</u>

²⁸⁶ Junquera-Varela, Felix, Haven, Bernard (December 18, 2018), Getting to 15 percent: addressing the largest tax gaps. <u>https://blogs.worldbank.org/governance/getting-15-percent-addressing-largest-tax-gaps</u>

²⁸⁷ People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018.

https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4 7/Ch-01%20%28English-2018%29.pdf 201,596 crore, exchange rate May 29, 2019. XE.com. https://www.xe.com/currencytables/?from=BDT&date=2017-01-01

²⁸⁸ Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018. https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4 7/Ch-01%20%28English-2018%29.pdf

²⁸⁹ IMF (2018), Bangladesh: 2018 Article IV Consultation-Press Release; Staff Report; and Statement by the Executive Director for Bangladesh. <u>https://www.imf.org/en/Publications/CR/Issues/2018/06/08/Bangladesh-2018-Article-IV-Consultation-Press-Release-Staff-Report-and-Statement-by-the-45959</u>

²⁹⁰ Ahmed, Sadiq (June 12, 2018), Economic and social impact of the FY2019 budget.

https://www.thedailystar.net/business/economic-and-social-impact-the-fy2019-budget-1589989

²⁹¹Bdnews24.com (June 2018), Bangladesh targets 10 million registered taxpayers by the end of FY2023. <u>https://bdnews24.com/economy/2018/06/07/bangladesh-targets-10-million-registered-taxpayers-by-the-end-of-</u> fy2023

²⁹² Islam, Rafikul (November 13, 2018), Week-long Income Tax Fair 2018 begins across country. Dhaka Tribune.
 <u>https://www.dhakatribune.com/business/2018/11/13/finance-minister-inaugurates-week-long-income-tax-fair-2018</u>
 ²⁹³ Bdnews24.com (November 30, 2018), We must make a 'habit' of paying income tax for development: acting

NBR chief. <u>https://bdnews24.com/economy/2018/11/30/we-must-make-a-habit-of-paying-income-tax-for-</u> development-acting-nbr-chief

²⁹⁴ Newagebusiness (November 4, 2018), 656 highest taxpayers to get NBR award.

http://www.newagebd.net/article/54970/656-highest-taxpayers-to-get-nbr-award

²⁹⁵ KPMG (Accessed 02/02/2019), Tax rates online; KMPG interactive tax rates tool.

https://home.kpmg/ky/en/home/services/tax/tax-tools-and-resources/tax-rates-online.html

²⁹⁶ Limited data are available; years 2013, 2014, 2015 and 2016, respectively for India, Pakistan, Afghanistan and Bhutan. There are no excise taxes in Afghanistan. IMF (2018), Government Finance Statistics Yearbook 2017. <u>https://books.google.nl/books?id=OtVVDwAAQBAJ&printsec=frontcover&hl=nl&source=gbs_ge_summary_r&cad</u> <u>=0#v=onepage&q=bangladesh&f=false</u>

²⁹⁷ KPMG Bangladesh (September 2018), Taxation Handbook.

https://assets.kpmg/content/dam/kpmg/bd/pdf/TaxationHandbookFinanceAct2018.pdf

²⁹⁸ OECD (2018), Development Co-operation Report 2018: Joining Forces to Leave No One Behind. <u>https://doi.org/10.1787/dcr-2018-en</u>.

²⁹⁹ OECD (2019), Society at a Glance: Asia/Pacific 2019. <u>https://doi.org/10.1787/soc_aag-2019-en</u> ³⁰⁰ IMF (2018), Government Finance Statistics Yearbook 2017.

https://books.google.nl/books?id=OtVVDwAAQBAJ&printsec=frontcover&hl=nl&source=gbs_ge_summary_r&cad =0#v=onepage&g=bangladesh&f=false

³⁰¹ Limited data are available; years 2014, 2016 and 2017, respectively for Pakistan, Bhutan and Nepal. IMF (2018), Government Finance Statistics Yearbook 2017.

https://books.google.nl/books?id=OtVVDwAAQBAJ&printsec=frontcover&hl=nl&source=gbs_ge_summary_r&cad =0#v=onepage&q=bangladesh&f=false

³⁰² Bdnews24.com (June 2018), Bangladesh targets 10 million registered taxpayers by the end of FY2023. <u>https://bdnews24.com/economy/2018/06/07/bangladesh-targets-10-million-registered-taxpayers-by-the-end-of-fy2023</u>

³⁰³ Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03_3c11_4e89_9f30_d7 9639595c67/7th FYP 18 02 2016.pdf

³⁰⁴ Not a single country in the database is on track to achieve all the goals by 2030. Sachs, J., Schmidt-Traub, G., Kroll, C., Lafortune, G., Fuller, G. (2018), SDG Index and Dashboards Report 2018. Bertelsmann Stiftung and Sustainable Development Solutions Network (SDSN). <u>https://dashboards.sdgindex.org/#/BGD</u>

³⁰⁵ Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03_3c11_4e89_9f30_d7 9639595c67/7th_FYP_18_02_2016.pdf

³⁰⁶ Centre for Policy Dialogue (CPD) (2007), Bangladesh Vision 2021. <u>http://cpd.org.bd/wp-</u> <u>content/uploads/2007/08/Bangladesh-Vision-2021-English.pdf</u> By the end of 2019, the next Vision 2041 is expected to be released.

³⁰⁷ Government of the People's Republic of Bangladesh (2017), Bangladesh Delta Plan 2100.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/effd0c7c_cb60_4120_914b_e4 9d752ee093/Bangladesh%20Delta%20Plan%202100%20English%20Version%20(2).pdf

³⁰⁸ Ministry of Food, Government of the People's Republic of Bangladesh (2017) Bangladesh Second country Investment Plan Nutrition-sensitive food systems (2016-2020).

http://fpmu.gov.bd/agridrupal/sites/default/files/file/CIP2FinalPDFPrintedCopy.pdf

³⁰⁹ Government of People's Republic of Bangladesh (2009), Bangladesh Climate Change Strategy and Action Plan.

http://www.bd.undp.org/content/dam/bangladesh/img/Projects/IBFCR/pub/Bangladesh%20Climate%20Change% 20Strategy%20and%20Action%20Plan%20(BCCSAP).pdf

³¹⁰ Ministry of Environment and Forests (MOEF) Government of the People's Republic of Bangladesh (2015), Intended Nationally Determined Contributions (INDC).

https://www4.unfccc.int/sites/ndcstaging/PublishedDocuments/Bangladesh%20First/INDC_2015_of_Bangladesh.pdf

³¹¹ Government of the People's Republic of Bangladesh (July 2015), National Social Security Strategy (NSSS) of Bangladesh.

http://www.lged.gov.bd/UploadedDocument/UnitPublication/1/323/20.%20National%20Social%20Security%20Str ategy%20(NSSS).pdf

³¹² Khatun, F., Saadat, S.Y. (2018), Towards a Social Protection Strategy for Bangladesh. <u>https://cpd.org.bd/wp-content/uploads/2018/08/Working-Paper-117-Towards-a-Social-Protection-Strategy-for-Bangladesh.pdf</u>

³¹³ Ahmed, Sadiq, Chattopadhyay, Suvojit (September 20, 2018), The case for environmental fiscal reforms in Bangladesh. Policy Research Institute. Webpage. <u>http://www.pri-bd.org/main/view_publication/the-case-for-environmental-fiscal-reforms-in-bangladesh_418</u>

³¹⁴ The Ex'tax Project, Deloitte, EY, KPMG, Meijburg and PwC (2016), New Era. New plan. Europe. A Fiscal Strategy for an Inclusive, Circular Economy. <u>http://www.neweranewplan.com/</u>

³¹⁵ Such measures have been suggested by Ahmed, Sadiq (2018), Environmental Fiscal Reforms in Bangladesh. Policy Research Institute. Economic Dialogue on Green Growth, UK Aid from the British People. <u>http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPape%2B7%2BEnviron</u> <u>mental%2BFiscal%2BReforms%2Bin%2BBangladesh.pdf</u>)</u>

³¹⁶ In general, revenue-neutral means that the revenues from new tax rates introduced are used to offset something else. Budget neutral means that the overall government budget position is unchanged, which in a modelling exercise would only be possible if the model covered every source of tax and expenditure.

³¹⁷ See for example: Taylor, M. (February 19, 2019), Norway is now paying Indonesia for not cutting down trees. https://www.weforum.org/agenda/2019/02/norway-starts-payments-to-indonesia-for-cutting-forest-emissions

```
<sup>318</sup> Diffenbaugh, N.S., Burke, M. (2019), Global warming has increased global economic inequality. 
<u>https://www.pnas.org/content/pnas/early/2019/04/16/1816020116.full.pdf</u>
```

³¹⁹ Ministry of Finance, Bangladesh (June 2014), Climate Fiscal Framework. <u>http://www.bd.undp.org/content/dam/bangladesh/docs/Projects/IBFCR/national/Bangladesh%20Climate%20Fisca</u> 1%20Framework.pdf

³²⁰ Davies, Simon, Fan, Qumiao, Hussain, Zahid (November 22, 2016), Carbon taxes: an opportunity for Bangladesh? <u>https://www.brookings.edu/blog/future-development/2016/11/22/carbon-taxes-an-opportunity-for-bangladesh/</u>

³²¹ Davies, Simon, Fan, Qumiao, Hussain, Zahid (November 22, 2016), Carbon taxes: an opportunity for Bangladesh? <u>https://www.brookings.edu/blog/future-development/2016/11/22/carbon-taxes-an-opportunity-for-bangladesh/</u>

³²² "(...) revenues from carbon taxes are less susceptible to economic cycles than most other taxes. While these do vary with the economic cycle, they tend to do so by less than other factors that impact tax receipts since, while energy consumption tends to decline (or growth slow) during periods of slow growth, it does so by less than firm, labor market and import performance, which generate the bulk of fiscal revenue. This helps to make them less pro-cyclical than other taxes, helping to shore up fiscal revenue during periods of economic slow-down." World Bank (2018), Policy Note on: Options for a carbon tax in Bangladesh.

http://documents.worldbank.org/curated/en/721381535709573696/pdf/Carbon-Taxation-in-Bangladesh.pdf ³²³ Muntean, M., Guizzardi, D. (et al.) (2018), Fossil CO2 emissions of all world countries - 2018 Report. http://edgar.jrc.ec.europa.eu/overview.php?v=booklet2018

³²⁴ Ministry of Finance, Bangladesh (June 2014), Climate Fiscal Framework.

http://www.bd.undp.org/content/dam/bangladesh/docs/Projects/IBFCR/national/Bangladesh%20Climate%20Fisca 1%20Framework.pdf

³²⁵ Climate Home News (June 2, 2017), Bangladesh backtracks on carbon tax proposal ahead of election year.
 <u>https://www.climatechangenews.com/2017/06/02/bangladesh-backtracks-carbon-tax-proposal-ahead-election/</u>
 ³²⁶ World Bank (2018), Policy Note on: Options for a carbon tax in Bangladesh.

http://documents.worldbank.org/curated/en/721381535709573696/pdf/Carbon-Taxation-in-Bangladesh.pdf ³²⁷ Ahmed, Sadiq, Khondker, Bazlul H. (2018), Towards a Carbon Tax in Bangladesh. Economic Dialogue on Green Growth, UK Aid from the British People.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPaper%2B9%2BToward s%2Ba%2BCarbon%2BTax%2Bin%2BBangladesh.pdf

³²⁸ Ahmed, Sadiq, Khondker, Bazlul H. (April 2018), Towards a Carbon Tax in Bangladesh. Economic Dialogue on Green Growth, UK Aid from the British People.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPaper%2B9%2BToward s%2Ba%2BCarbon%2BTax%2Bin%2BBangladesh.pdf

³²⁹ World Bank (October 2018), Bangladesh Development Update.

http://documents.worldbank.org/curated/en/556621538579959997/pdf/130459-WP-P168106-BDUOctober.pdf

³³⁰ UNCTAD (December 2018), Leapfrogging: Look before you leap. Policy Brief based on UNCTAD, 2018, Technology and Innovation Report 2018: Harnessing Frontier Technologies for Sustainable Development. <u>https://unctad.org/en/PublicationsLibrary/presspb2018d8_en.pdf</u>

³³¹ Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03 3c11 4e89 9f30 d7 9639595c67/7th FYP 18 02 2016.pdf

³³² OECD (2018), Taxing Energy Use 2018. <u>https://read.oecd-ilibrary.org/taxation/taxing-energy-use-</u>2018 9789264289635-en#page43

³³³ CPLC (2017), Report of the High-level Commission on Carbon Prices.

https://www.carbonpricingleadership.org/report-of-the-highlevel-commission-on-carbon-prices/

³³⁴ World Bank Group (2019), State and Trends of Carbon Pricing 2019.

https://openknowledge.worldbank.org/handle/10986/31755

³³⁵ Ahmed, Sadiq and Khondker, Bazlul H. (April 2018), Towards a Carbon Tax in Bangladesh. Economic Dialogue on Green Growth, UK Aid from the British People.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPaper%2B9%2BToward s%2Ba%2BCarbon%2BTax%2Bin%2BBangladesh.pdf

³³⁶Ministry of Finance, Bangladesh (June 2014), Climate Fiscal Framework.

http://www.bd.undp.org/content/dam/bangladesh/docs/Projects/IBFCR/national/Bangladesh%20Climate%20Fisca 1%20Framework.pdf

³³⁷ A carbon tax sets a price and the volume of pollution emerges as an outcome. An ETS sets the total quantity of carbon emission and lets the market set the price of carbon through an auction and internal trading.

IBRD/World Bank and ECOFYS (2018), State and Trends of Carbon Pricing 2018. https://www.ecofys.com/files/files/worldbank-state-trends-carbon- pricing-2018.pdf

³³⁸ "While a lower-end estimate of the climate cost of carbon alone is EUR 30 per tCO2 (see OECD, 2016 and references provided therein), the Mexican carbon tax effectively prices carbon emissions from energy used for heating and process purposes at only MXN 22.24 per tCO2 (EUR 1.08 per tCO2 using the 2016 average EUR-MXN exchange rate) on a weighted-average basis across all fuels. The effective tax rate on carbon emissions is even lower in the electricity sector, where the tax prices carbon emissions from electricity generation at just MXN 15.43 per tCO2 (EUR 0.75 per tCO2)." Arlinghaus, J., van Dender, K. (2017), The environmental tax and subsidy reform in Mexico. OECD Taxation Working Papers, No. 31. https://doi.org/10.1787/a9204f40-en

³³⁹ Cottrell, Jacqueline, Falcão, Tatiana (2018), A Climate of fairness; Environmental Taxation and Tax Justice in Developing Countries. <u>http://www.vidc.org/fileadmin/Bibliothek/DP/A Climate of Fairness.pdf</u>

³⁴⁰ France24 (May 27, 2019), Coal-hungry S.Africa introduces carbon tax.

https://www.france24.com/en/20190527-coal-hungry-safrica-introduces-carbon-tax

³⁴¹ CBS (2015), De invloed van de aardgaswinning op de Nederlandse economie. <u>https:// www.cbs.nl/nl-</u>

 $\underline{nl/achtergrond/2015/49/de-invloed-van-de-aardgas winning-\ op-de-nederlandse-economie}$

³⁴² CBS (May 28, 2019), Aardgasbaten uit gaswinning bijna 417 miljard euro. https://www.cbs.nl/nl-

nl/nieuws/2019/22/aardgasbaten-uit-gaswinning-bijna-417-miljard-euro

³⁴³ Mulder, M., Perey, P. (2018), Gas production and earthquakes in Groningen.

https://www.rug.nl/feb/news/2018/180605-gas-production-and-earthquakes.pdf

³⁴⁴ Projections based on 2017 IEA data. Coal is not subsidized in Bangladesh. The subsidy for electricity was
 \$529.1 million in 2017. IEA (2019), Fossil-fuel subsidies. https://www.iea.org/weo/energysubsidies/
 ³⁴⁵ Year: 2017. UN Comtrade (Accessed March 18, 2019), Bangladesh; import; petroleum products.

https://comtrade.un.org/labs/data-explorer/

³⁴⁶ IEA (2019), Fossil-fuel subsidies. <u>https://www.iea.org/weo/energysubsidies/</u>

³⁴⁷ Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018. <u>https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4</u> <u>7/Ch-01%20%28English-2018%29.pdf</u>

³⁴⁸ Such 'Post-tax' subsidies include: 'Pre-tax' subsidies (\$2.48 bln), Global warming (\$2.73 bln), Local air pollution (\$1.1 bln), Congestion (\$0.02 bln), Accidents (\$0.31 bln), Road damage (\$0.03 bln) and Foregone consumption tax revenue (\$2.17 bln). IMF (July 17, 2015), IMF Survey: Counting the Cost of Energy Subsidies. https://www.imf.org/en/News/Articles/2015/09/28/04/53/sonew070215a

³⁴⁹ Ministry of Finance, Bangladesh (June 2014), Climate Fiscal Framework.

http://www.bd.undp.org/content/dam/bangladesh/docs/Projects/IBFCR/national/Bangladesh%20Climate%20Fisca 1%20Framework.pdf

³⁵⁰ Year: 2016. IEA (Accessed May 22, 2019), World Energy Balances 2018.

https://www.iea.org/statistics/?country=BANGLADESH&year=2016&category=Energy%20supply&indicator=TPE SbySource&mode=chart&dataTable=BALANCES

³⁵¹ Cambridge Econometrics projections based on BGR (2015), Energy Study 2015. Reserves, resources and availability of energy resources.

https://www.bgr.bund.de/EN/Themen/Energie/Downloads/energiestudie 2015 en.pdf? blob=publicationFile&v= 2

³⁵² Ahmed, Sadiq (July 2018), Environmental Fiscal Reforms in Bangladesh.

http://www.greengrowthknowledge.org/sites/default/files/downloads/resource/EDGG%2BPape%2B7%2BEnviron mental%2BFiscal%2BReforms%2Bin%2BBangladesh.pdf

³⁵³ Ministry of Finance, Bangladesh (June 2014), Climate Fiscal Framework,

http://www.bd.undp.org/content/dam/bangladesh/docs/Projects/IBFCR/national/Bangladesh%20Climate%20Fisca 1%20Framework.pdf

³⁵⁴ Asian Development Bank (ADB) (2016), Policy Report: Fossil fuel subsidies in Asia: Trends, Impacts and reforms. <u>https://www.adb.org/sites/default/files/publication/182255/fossil-fuel-subsidies-asia.pdf</u>
 ³⁵⁵ World Bank (October 2018), Bangladesh Development Update.

http://documents.worldbank.org/curated/en/556621538579959997/pdf/130459-WP-P168106-BDUOctober.pdf ³⁵⁶ Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03_3c11_4e89_9f30_d7 9639595c67/7th_FYP_18_02_2016.pdf

³⁵⁷ Asian Development Bank (ADB) (2016) Fossil Fuel Subsidies in Asia Trends, Impacts, and Reforms.
 Integrative Report. <u>https://www.adb.org/sites/default/files/publication/182255/fossil-fuel-subsidies-asia.pdf</u>
 ³⁵⁸ European Parliament (2017), Fossil Fuel Subsidies: In-depth Analysis.

```
http://www.europarl.europa.eu/RegData/etudes/IDAN/2017/595372/IPOL IDA(2017)595372 EN.pdf
<sup>359</sup> European Parliament (2017), Fossil Fuel Subsidies: In-depth Analysis.
```

http://www.europarl.europa.eu/RegData/etudes/IDAN/2017/595372/IPOL IDA(2017)595372 EN.pdf

³⁶⁰ KPMG (Accessed February 2019), Tax rates online; KMPG interactive tax rates tool.

https://home.kpmg/ky/en/home/services/tax/tax-tools-and-resources/tax-rates-online.html

³⁶¹ WRG (Water Resources Group) (2015), An Analysis of Industrial Water Use in Bangladesh with a Focus on the Textile and Leather Industries. <u>https://www.2030wrg.org/wp-content/uploads/2016/02/WRG-Bangladesh-Report.pdf</u>

³⁶² Sakamoto, M., Ahmed, T., Begum, S., Huq, H. (2019), Water Pollution and the Textile Industry in Bangladesh: Flawed Corporate Practices or Restrictive Opportunities?

https://www.sciencedirect.com/science/article/pii/S2211464517301926?via%3Dihub

³⁶³ Mike Schragger, Swedish Textile Initiative for Climate Action (STICA), in Ecotextile news, (date unknown) Global warning. Hard Talk. Page 13-14.

³⁶⁴ Ellen McArthur Foundation (2018), The circular economy opportunity for urban and industrial innovation in China.

https://www.ellenmacarthurfoundation.org/assets/downloads/The-circular-economy-opportunity-for-urbanindustrial-innovation-in-China 19-9-18 1.pdf

³⁶⁵ See for example: Ralph Lauren (waterless dying:

<u>https://www.apnews.com/4f10afd9933c43059fd7cabc23705ef6</u>); Mistra Future Fashion (ultrasound laser cutting: <u>http://mistrafuturefashion.com</u>).

³⁶⁶ See for example: Earthcolours (dyes made from agricultural waste:

https://www.archroma.com/innovations/earth-colors-by-archroma), Seacolours (algae-based dyes:

http://www.seacolors.eu); Natsai Audrey Chieza (dyes that use bacteria: https://www.natsaiaudrey.co.uk/).

³⁶⁷ See for example: MuSkin (plant-based leather: <u>https://lifematerials.eu/en/shop/muskin/</u>); Fruitleather (leather-like material from fruit waste: <u>https://fruitleather.nl/</u>); JW Couture (fish skin leather:

https://www.bbc.com/news/business-47806892); Tencel (eucalyptus-based material: https://www.tencel.com); Ananas Anam (pineapple leaf fibre: https://www.ananas-anam.com/); Levi's (hemp:

<u>https://www.highsnobiety.com/p/levis-hemp-sustainable-fabric/</u>); Green Nettle Textile (stinging nettle) <u>http://greennettletextiles.com/</u>.

³⁶⁸ Kasabe, Nanda (April 24, 2018), Bangladesh emerges largest importer of Indian cotton.

https://www.financialexpress.com/market/commodities/bangladesh-emerges-largest-importer-of-indiancotton/1143136/

³⁶⁹ Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03_3c11_4e89_9f30_d7 9639595c67/7th_FYP_18_02_2016.pdf

³⁷⁰ CPD (2018), Press Briefing on Release of The Global Competitiveness Report 2018 and The Bangladesh Business Environment Study 2018. <u>https://cpd.org.bd/wp-content/uploads/2018/10/Presentation-on-Release-of-The-GCR-2018-and-The-Bangladesh-Business-Environment-Study-2018.pdf</u>

³⁷¹ Government of People's Republic of Bangladesh (2015), 7th Five Year Plan of Bangladesh (2016-2021). General Economics Division (GED) Bangladesh Planning Commission.

http://www.lged.gov.bd/UploadedDocument/UnitPublication/1/361/7th FYP 18 02 2016.pdf

³⁷² IMF (2018), Bangladesh: 2018 Article IV Consultation-Press Release; Staff Report; and Statement by the Executive Director for Bangladesh. <u>https://www.imf.org/en/Publications/CR/Issues/2018/06/08/Bangladesh-2018-Article-IV-Consultation-Press-Release-Staff-Report-and-Statement-by-the-45959</u>

³⁷³ World Bank (2015), Bangladesh - More and better jobs to accelerate shared growth and end extreme poverty: a systematic country diagnostic. <u>http://documents.worldbank.org/curated/en/950731468000249915/Bangladesh-Moreand-better-jobs-to-accelerate-shared-growth-and-end-extreme-poverty-a-systematic-country-diagnostic</u>

³⁷⁴ IMF (2018), Bangladesh: 2018 Article IV Consultation-Press Release; Staff Report; and Statement by the Executive Director for Bangladesh. <u>https://www.imf.org/en/Publications/CR/Issues/2018/06/08/Bangladesh-2018-Article-IV-Consultation-Press-Release-Staff-Report-and-Statement-by-the-45959</u>

³⁷⁵ See: Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03_3c11_4e89_9f30_d7 9639595c67/7th FYP 18 02 2016.pdf

³⁷⁶ As proposed by Dr. Huq (November 29, 2017), Easing the pressure on Dhaka.

https://www.thedailystar.net/opinion/politics-climate-change/easing-the-pressure-dhaka-1497694

³⁷⁷ Eckhouse, Brian (May 15, 2019), The U.S. Has a Fleet of 300 Electric Buses. China Has 421,000.

https://www.bloomberg.com/news/articles/2019-05-15/in-shift-to-electric-bus-it-s-china-ahead-of-u-s-421-000-to-300

³⁷⁸ India approved a \$1.4 billion, three-year scheme to subsidize electrics. Bahree, Megha (March 29, 2019), India goes electric with battery-swapping rickshaws. <u>https://www.wired.com/story/india-sun-mobility-electric-rickshaw-bus-battery-swap/</u>

³⁷⁹ Peters, Adele (Sept 28, 2018), This startup lets villagers create mini power grids for their neighbors. Fastcompany. <u>https://www.fastcompany.com/90241777/this-startup-lets-villagers-create-mini-power-grids-for-their-neighbors</u>

³⁸⁰ Paris Innovation Review (February 2016), Agriculture and food: the rise of digital platforms. http://parisinnovationreview.com/articles-en/agriculture-and-food-the-rise-of-digital-platforms

 ³⁸¹ Van Tuijl, E., Hospers, G. J., Van Den Berg, L. (2018), Opportunities and Challenges of Urban Agriculture for Sustainable City Development. <u>https://content.sciendo.com/downloadpdf/journals/esrp/25/2/article-p5.pdf</u>
 ³⁸² IMF (2018), Bangladesh: 2018 Article IV Consultation-Press Release; Staff Report; and Statement by the

Executive Director for Bangladesh. <u>https://www.imf.org/en/Publications/CR/Issues/2018/06/08/Bangladesh-2018-</u> <u>Article-IV-Consultation-Press-Release-Staff-Report-and-Statement-by-the-45959</u>

³⁸³ Cash transfers are found to be more cost-effective than food transfers in some selected areas of Bangladesh. Ahmed, A. U., Hoddinott, F. J., Roy, S., Sraboni, E., Quabili, R. W. and Margolies, A. (2016). Which Kinds of Social Safety Net Transfers Work Best for the Ultra Poor in Bangladesh?: Operation and Impacts of the Transfer Modality Research Initiative.

³⁸⁴ Bangladesh and India are familiar with such schemes. India has the Mahatma Gandhi National Rural Employment Guarantee Act (MGNREGA). In addition to providing a contribution to incomes, such schemes can contribute to climate change adaptation and mitigation through the work performed (for example in reforestation, water and soil conservation and flood control). ILO (2018), World Employment and Social Outlook 2018 - Greening with Jobs. <u>https://www.ilo.org/weso-greening/documents/WESO_Greening_EN_web2.pdf</u>

³⁸⁵ 485 billion BDT. (1 BDT = 78.910 USD). XE.com (Accessed 29/05/2019), Current and Historical Rate Tables. <u>https://www.xe.com/currencytables/?from=BDT&date=2017-01-01</u> Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018.

https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4 7/Ch-01%20%28English-2018%29.pdf

³⁸⁶ Government of the People's Republic of Bangladesh (December 2015), 7th Five Year Plan. FY2016-FY2020. Accelerating Growth, Empowering Citizens. General Economics Division (GED) Bangladesh Planning Commission.

http://plancomm.portal.gov.bd/sites/default/files/files/plancomm.portal.gov.bd/files/aee61c03_3c11_4e89_9f30_d7 9639595c67/7th_FYP_18_02_2016.pdf

³⁸⁷ World Bank (2018), Poverty and Shared Prosperity 2018: Piecing Together the Poverty Puzzle. <u>https://openknowledge.worldbank.org/bitstream/handle/10986/30418/9781464813306.pdf</u>

³⁸⁸ OECD, World Bank, UN Environment (2018), Financing Climate Futures: Rethinking Infrastructure. <u>https://doi.org/10.1787/9789264308114-en</u>

³⁸⁹ Government of the People's Republic of Bangladesh (July 2015), National Social Security Strategy (NSSS) of Bangladesh.

http://www.lged.gov.bd/UploadedDocument/UnitPublication/1/323/20.%20National%20Social%20Security%20Str ategy%20(NSSS).pdf

³⁹⁰ ILO (2018), World Employment and Social Outlook 2018 - Greening with Jobs. <u>https://www.ilo.org/weso-greening/documents/WESO_Greening_EN_web2.pdf</u>

³⁹¹ Cambridge Econometrics is a trading subsidiary of a charity, the Cambridge Trust for New Thinking in Economics.

³⁹² See for full description and manual: <u>www.e3me.com</u>.

³⁹³ Pollitt, Hector (2008), Combining Economic and Material Flows Analysis at the Sectoral level: Development of the E3ME Model and Application in the MATISSE Case Studies. Deliverable 8.6.1, Work Package 8, MATISSE, European Commission project No 004059 (GOCE).

³⁹⁴ See: Ekins, P., Pollitt, H., Summerton, P., Chewpreecha, U. (2012), Increasing Carbon and Material Productivity through Environmental Tax Reform, Energy Policy, 42 (3): 365-376; Pollitt, H., Chewpreecha, U. (2011), Macroeconomic modelling of sustainable development and the links between the economy and the environment. DG Environment, European Commission; European Commission (2014), Study on modelling of the economic and environmental impacts of raw material consumption; European Commission (Accessed July 2016), 2030 Energy Strategy; Cambridge Econometrics (2013), Modelling Milestones for Achieving Resource Efficiency: Economic Analysis of Waste Taxes Draft Report for the European Commission (DG Environment); Lee, S., Pollitt, H., Park, S-J. (Eds.) (2016), Low-carbon, Sustainable Future in East Asia: improving energy systems, taxation and policy cooperation, Routledge Studies in the Modern World Economy; European Commission (2018), Impacts of the Circular Economy on the labour market, see <u>https://publications.europa.eu/en/publication-detail/-/publication/fc373862-704d-11e8-9483-01aa75ed71a1/language-en;</u> New Climate Economy (2018), Unlocking the inclusive growth story of the 21st century, see <u>https://newclimateeconomy.report/2018/wp-</u>

content/uploads/sites/6/2018/09/NCE_2018_FULL-REPORT.pdf; Mercure, J-F., Pollitt, H., Viñuales, J.E. (et al.) (2018) Macroeconomic impact of stranded fossil fuel assets. Nature Climate Change, Volume 8, pp 588–593. ³⁹⁵ See e.g. Lavoie, Marc (2015) Post-Keynesian Economics: New Foundations. <u>https://www.e-</u>elgar.com/shop/post-keynesian-economics-15184

³⁹⁶ World Bank (October 2018), Bangladesh Development Update.

http://documents.worldbank.org/curated/en/556621538579959997/pdf/130459-WP-P168106-BDUOctober.pdf ³⁹⁷ HSBC (2018), The World in 2030. Our long-term projections for 75 countries. <u>https://enterprise.press/wp-</u>

content/uploads/2018/10/HSBC-The-World-in-2030-Report.pdf ³⁹⁸ Remy, Nathalie, Speelman, Eveline, Swartz, Steven (October 2016), Style that's sustainable: A new fastfashion formula.

https://www.mckinsey.com/business-functions/sustainability/our-insights/style-thats-sustainable-a-new-fast-fashion-formula

³⁹⁹ European Union (2019), Environmental impact of the textile and clothing industry. What consumers need to know. Briefing. European Parliamentary Research Service.

http://www.europarl.europa.eu/RegData/etudes/BRIE/2019/633143/EPRS_BRI(2019)633143_EN.pdf 400 WRAP (2012), Valuing our clothes.

http://www.wrap.org.uk/sites/files/wrap/VoC%20FINAL%20online%202012%2007%2011.pdf

⁴⁰¹ <u>https://www.mckinsey.com/business-functions/sustainability/our-insights/style-thats-sustainable-a-new-fast-</u>fashion-formula

⁴⁰² Quantis (2018), Measuring Fashion 2018. Environmental Impact of the Global Apparel and Footwear Industries Study Full report and methodological considerations.

https://quantis-intl.com/wp-content/uploads/2018/03/measuringfashion_globalimpactstudy_fullreport_quantis_cwf_2018a.pdf

⁴⁰³ Ellen MacArthur Foundation (2017), A new textiles economy: Redesigning fashion's future. <u>http://www.ellenmacarthurfoundation.org/publications</u>

⁴⁰⁴ Kerr, J., Landry, J. (2017), Pulse of the fashion industry. Global Fashion Agenda.

https://globalfashionagenda.com/wp-content/uploads/2017/05/Pulse-of-the-Fashion-Industry_2017.pdf ⁴⁰⁵ Signatories by June 2018. Global Fashion Agenda (Accessed May 22, 2019), 2020 Commitment. https://www.globalfashionagenda.com/commitment/#

⁴⁰⁶ Obe, M. (November 2018), Bangladesh fights for future of its garment industry.

https://asia.nikkei.com/Business/Business-trends/Bangladesh-fights-for-future-of-its-garment-industry

⁴⁰⁷ Nagaraj, Anuradha (Sept 17, 2018), Pay more for your clothes, Bangladesh workers tell global fashion brands. Reuters. <u>https://www.reuters.com/article/us-bangladesh-labor-fashion/pay-more-for-your-clothes-bangladesh-workers-tell-global-fashion-brands-idUSKCN1LX23X</u>

⁴⁰⁸ The Fashion Law (June 18, 2018), It Would Cost You 20 Cents More Per T-shirt to Pay an Indian Worker a Living Wage. <u>http://www.thefashionlaw.com/home/it-would-cost-you-20-cents-more-per-t-shirt-to-pay-an-indian-worker-a-living-wage</u>

⁴⁰⁹ In addition to a retail price of €14-€200. Report available in Dutch. ABN AMRO, Impact Institute (2019), De verborgen kosten van een spijkerbroek. True pricing in de jeansketen.

https://www.abnamro.nl/nl/images/Content/OneShop/Insights/Sectoren_en_trends/Retail/1795676/PdfOS_Spijker broek_bijna_33_euro_te_goedkoop.pdf

⁴¹⁰ WRG (The 2030 Water Resources Group) (2015), An Analysis of Industrial Water Use in Bangladesh with a Focus on the Textile and Leather Industries. <u>https://www.2030wrg.org/wp-content/uploads/2016/02/WRG-Bangladesh-Report.pdf</u>

⁴¹¹ Thredup (2019), 2019 Resale Report. <u>https://www.thredup.com/resale</u>

⁴¹² Ellen MacArthur Foundation (2017), A new textiles economy: Redesigning fashion's future. http://www.ellenmacarthurfoundation.org/publications

⁴¹³ World Bank (2018), Policy Note on: Options for a carbon tax in Bangladesh.

http://documents.worldbank.org/curated/en/721381535709573696/pdf/Carbon-Taxation-in-Bangladesh.pdf ⁴¹⁴ World Bank (2018), Policy Note on: Options for a carbon tax in Bangladesh.

http://documents.worldbank.org/curated/en/721381535709573696/pdf/Carbon-Taxation-in-Bangladesh.pdf ⁴¹⁵ Davies, Simon, Fan, Qumiao, Hussain, Zahid (November 22, 2016), Carbon taxes: an opportunity for Bangladesh? <u>https://www.brookings.edu/blog/future-development/2016/11/22/carbon-taxes-an-opportunity-forbangladesh/</u>

⁴¹⁶ World Bank (2018), Policy Note on: Options for a carbon tax in Bangladesh.

http://documents.worldbank.org/curated/en/721381535709573696/pdf/Carbon-Taxation-in-Bangladesh.pdf ⁴¹⁷ Compared to 2015. Global Fashion Agenda & Boston Consulting Group (2018). 2018 Pulse of the Fashion Industry. <u>https://www.globalfashionagenda.com/initiatives/pulse/</u>

⁴¹⁸ Government of the People's Republic of Bangladesh (December 2018), Bangladesh Economic Review 2018. <u>https://mof.portal.gov.bd/sites/default/files/files/mof.portal.gov.bd/page/e8bc0eaa_463d_4cf9_b3be_26ab70a32a4</u> <u>7/Ch-01%20%28English-2018%29.pdf</u>

⁴¹⁹ Schlanger, Zoe (April 18, 2018), If shipping were a country, it would be the world's sixth-biggest greenhouse gas emitter. World Economic Forum website. In collaboration with Quartz.

https://www.weforum.org/agenda/2018/04/if-shipping-were-a-country-it-would-be-the-world-s-sixth-biggest-greenhouse-gas-emitter

⁴²⁰ Feike Sijbesma (June 2013), We Need to Redesign Our Economy. <u>https://www.huffpost.com/entry/we-need-to-redesign-our-e_b_2597564</u>